UNITED STATES SECURITIES AND EXCHANGE COMMISSION

Washington, D.C. 20549

FORM 8-K

CURRENT REPORT Pursuant to Section 13 or 15(d) of the Securities Exchange Act of 1934

Date of Report (Date of earliest event reported): February 2, 2014

MYRIAD GENETICS, INC.

(Exact name of registrant as specified in its charter)

Delaware (State or other jurisdiction of incorporation) 0-26642 (Commission File Number) 87-0494517 (IRS Employer Identification No.)

320 Wakara Way Salt Lake City, Utah 84108 (Address of principal executive offices) (Zip Code)

Registrant's telephone number, including area code: (801) 584-3600

Not Applicable

(Former name or former address, if changed since last report)

Check the appropriate box below if the Form 8-K filing is intended to simultaneously satisfy the filing obligation of the registrant under any of the following provisions (*see* General Instruction A.2. below):

□ Written communications pursuant to Rule 425 under the Securities Act (17 CFR 230.425)

□ Soliciting material pursuant to Rule 14a-12 under the Exchange Act (17 CFR 240.14a-12)

Dere-commencement communications pursuant to Rule 14d-2(b) under the Exchange Act (17 CFR 240.14d-2(b))

Dere-commencement communications pursuant to Rule 13e-4(c) under the Exchange Act (17 CFR 240.13e-4(c))

ITEM 1.01 Entry into a Material Definitive Agreement.

On February 2, 2014, Myriad Genetics, Inc. ("Myriad" or the "Company") entered into an Amended and Restated Agreement and Plan of Merger (the "Merger Agreement") with privately-held Crescendo Biosciences, Inc. ("Crescendo"); Myriad Crescendo, Inc., a wholly-owned subsidiary of Myriad (the "Merger Subsidiary"); and a representative of the security holders of Crescendo (the "Representative"). The Merger Agreement supersedes the prior merger agreement, dated September 8, 2011.

The Merger Agreement provides that Myriad will pay an aggregate cash purchase price to the security holders of Crescendo of \$270 million minus \$25 million for the repayment of the loan made to Crescendo by Myriad in 2011 and subject to further adjustment for Crescendo's cash, other indebtedness, working capital and other amounts to be determined in accordance with the Merger Agreement (as adjusted, the "Merger Consideration"). Myriad is funding the purchase price out of cash and marketable securities on hand. The acquisition is structured as a reverse triangular merger transaction in which the Merger Subsidiary shall merge with and into Crescendo, with Crescendo surviving the merger as the surviving corporation and a wholly-owned subsidiary of Myriad and the separate corporate existence of the Merger Subsidiary shall cease (the "Merger"). On February 4, 2014, Myriad issued a press release announcing the execution of the Merger Agreement. A copy of the press release is attached hereto as Exhibit 99.1 and incorporated herein by reference.

The Merger, the Merger Agreement and the transactions contemplated thereby were unanimously approved by the boards of directors of Myriad, Crescendo and the Merger Subsidiary. After the execution and delivery of the Merger Agreement by the parties thereto, Crescendo's stockholders executed and delivered to Myriad a unanimous written consent approving the Merger, the Merger Agreement and the transactions contemplated thereby. Approval by Myriad's stockholders is not required. The transaction is expected to close before the end of Myriad's fiscal year 2014, subject to customary closing conditions and regulatory approval.

All shares of capital stock of Crescendo outstanding at the closing of the Merger (the "Closing"), including all stock options and warrants, will be converted into a right to receive a cash portion of the Merger Consideration. Also, upon the Closing, \$20.0 million will be deposited into an escrow account to fund (i) any post-closing adjustments to the Merger Consideration payable to Myriad based upon differences between the estimated working capital and the actual working capital of Crescendo at Closing, and (ii) any indemnification claims made by Myriad pursuant to the Merger Agreement for a period, based on the completion of an audit of Crescendo's financial statements, of no fewer than twelve nor more than fifteen months following Closing. Under the terms of the Merger Agreement, the stockholders of Crescendo are obligated to indemnify Myriad for all losses resulting from breaches of Crescendo's representations and warranties, Crescendo's breach or failure to perform its covenants under the Merger Agreement, and certain other specified matters. Myriad's indemnification claims are limited to and must be satisfied out of the escrow amount, except in certain circumstances.

The Merger Agreement may be terminated at any time prior to closing by either party through mutual consent, in the event all of the closing conditions are not satisfied by a specified date, or in the event of certain material breaches or misrepresentations by the other party. The foregoing description of the Merger Agreement is a summary and is qualified in its entirety by the terms of the Merger Agreement, a copy of which is filed as Exhibit 2.1 to this Current Report on Form 8-K.

The Merger Agreement has been attached as an exhibit to this Current Report on Form 8-K to provide investors and security holders with information regarding its terms. It is not intended to provide any other factual information about Myriad or Crescendo. The representations, warranties and covenants contained in the Merger Agreement were made only for the purposes of such agreement and as of

specified dates, were solely for the benefit of the parties to such agreement, and may be subject to limitations agreed upon by the contracting parties. The representations and warranties may have been made for the purposes of allocating contractual risk between the parties to the agreement instead of establishing these matters as facts, and may be subject to standards of materiality applicable to the contracting parties that differ from those applicable to investors. Investors are not third-party beneficiaries under the Merger Agreement and should not rely on the representations, warranties and covenants or any descriptions thereof as characterizations of the actual state of facts or condition of Myriad or Crescendo or any of their respective subsidiaries or affiliates. In addition, the assertions embodied in the representations and warranties contained in the Merger Agreement are qualified by information in a confidential disclosure schedule provided by Crescendo, which is not being filed with this Current Report on Form 8-K as permitted by the rules and regulations of the Securities and Exchange Commission. Accordingly, investors should not rely on the representations and warranties as characterizations of the actual state of facts, since (i) they were made only as of the date of such agreement or a prior, specified date, (ii) in some cases they are subject to qualifications with respect to materiality, knowledge and/or other matters, and (iii) they may be modified in important part by the underlying disclosure schedule. Moreover, information concerning the subject matter of the representations and warranties may change after the date of the Merger Agreement, which subsequent information may or may not be fully reflected in Myriad's public disclosures.

ITEM 2.02 Results of Operations and Financial Condition.

On February 4, 2014, Myriad announced its financial results for the three and six months ended December 31, 2013. The earnings release is attached hereto as an exhibit to this Current Report on Form 8-K and incorporated herein by reference, and is being furnished pursuant to this Item 2.02 as Exhibit 99.2 to this Current Report on Form 8-K.

ITEM 9.01 Financial Statements and Exhibits.

Exhibit <u>Number</u>	Description
2.1	Amended and Restated Agreement and Plan of Merger, dated as of February 2, 2014, by and among Myriad Genetics, Inc., Myriad Crescendo, Inc., Crescendo Bioscience, Inc. and the Representative.
99.1	Press release announcing entry into the Amended and Restated Agreement and Plan of Merger dated February 4, 2014.
99.2	Earnings release dated February 4, 2014 for the three and six months ended December 31, 2013.

The press releases contain hypertext links to information on our website or other parties' websites. The information on our website and other parties' websites is not incorporated by reference into this Current Report on Form 8-K and does not constitute a part of this Form 8-K.

In accordance with General Instruction B-2 of Form 8-K, the information set forth in Item 2.02 and in Exhibit 99.2 shall not be deemed to be "filed" for purposes of Section 18 of the Securities Exchange Act of 1934, as amended (the "Exchange Act"), or otherwise subject to the liability of that section, and shall not be incorporated by reference into any registration statement or other document filed under the Securities Act of 1933, as amended or the Exchange Act, except as shall be expressly set forth by specific reference in such filing.

SIGNATURES

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned hereunto duly authorized.

MYRIAD GENETICS, INC.

By: /s/ James S. Evans

James S. Evans Chief Financial Officer

Page 5

Date: February 4, 2014

EXHIBIT INDEX

- Exhibit Number
 Description

 2.1
 Amended and Restated Agreement and Plan of Merger dated as of February 2, 2014, by and among the Myriad Genetics, Inc., Myriad Crescendo, Inc., Crescendo Bioscience, Inc. and the Representative.
- 99.1 Press release announcing entry into the Amended and Restated Agreement and Plan of Merger dated February 4, 2014.
- 99.2 Earnings release dated February 4, 2014 for the three and six months ended December 31, 2013.

Exhibit 2.1

Execution Version

AMENDED AND RESTATED

AGREEMENT AND PLAN OF MERGER

BY AND AMONG

MYRIAD GENETICS, INC.,

MYRIAD CRESCENDO, INC.,

CRESCENDO BIOSCIENCE, INC.

AND

MDV IX, L.P.,

AS REPRESENTATIVE

Dated as of February 2, 2014

ARTICLE I	DEFINITIONS	PAGE 2
1.1	Certain Defined Terms	2
1.2	Additional Defined Terms	14
ARTICLE II	MERGER	18
2.1	[Reserved]	18
2.2	[Reserved]	18
2.3	The Merger	18
2.4	Effective Time	18
2.5	Effect of the Merger	18
2.6	Certificate of Incorporation and Bylaws of the Surviving Corporation	18
2.7	Directors and Officers	19
2.8	Taking of Necessary Action; Further Action	19
2.9	[Reserved]	19
2.10	Effect of Merger on Capital Stock	19
2.11	Actions at Closing on the Closing Date	21
2.12	Exchange of Certificates and Company Warrants	22
2.13	Options	23
2.14	Warrants	25
2.15	Appraisal Rights	27
2.16	Escrow	27
2.17	Working Capital Adjustment	28
ARTICLE III	I REPRESENTATIONS AND WARRANTIES OF THE COMPANY	31
3.1	Organization, Good Standing, Qualification and Power	31
3.2	Subsidiaries	32
3.3	Organizational Documents	33
3.4	Capitalization	33
3.5	Authorization; Binding Obligation	35
3.6	Consents	35
3.7	No Violation	36
3.8	Required Vote	36
3.9	Financial Statements; No Undisclosed Liabilities	37
3.10	Absence of Certain Events	38
3.11	Legal Proceedings	40

i

3.11 Legal Proceedings

3.12	Compliance with Laws	41
3.13	Compliance with Privacy and Data Security Laws	43
3.14	Title to Properties	43
3.15	Employee and Benefit Matters	44
3.16	6 Contracts	49
3.17	Officer and Director Indemnification Obligations	51
3.18	Intellectual Property	51
3.19	FDA Matters	56
3.20	Reimbursement	57
3.21	Environmental Matters	57
3.22	Tax Liabilities	58
3.23	Insurance	61
3.24	Brokers	61
3.25	Transactions with Affiliates	61
	6 Relationships	62
3.27	' Payments; Foreign Corrupt Practices Act; U.S. Export and Sanctions Laws	62
3.28	Manufacturing and Marketing Rights	62
3.29	Change of Control Agreements	63
3.30	Accounts Receivable; Bank Accounts	63
3.31	Restrictions on Business Activities	63
3.32	Customers; Distributors	64
3.33	Representations Complete	64
ARTICL	E IV REPRESENTATIONS AND WARRANTIES OF PARENT AND ACQUISITION SUB	64
4.1	Organization, Good Standing and Qualification	64
4.2	Ownership of Acquisition Sub; No Prior Activities	65
4.3	Authorization; Binding Obligation	65
4.4	Consents	65
4.5	No Violation	66
4.6	Legal Proceedings	66
4.7	Brokers	66
4.8	Funds	66
ARTICL	E V COVENANTS	67
5.1	Financial Statements	67
5.2	Board Observer Rights	67

ii

5.9 5.10 5.11 5.12	Distribution of Loan Amount Issuance of Additional Indebtedness Issuance of Additional Shares Approvals and Consents; Further Assurances	67 68 68 68 68 68 70 70 70 71 71 71
ARTICL	E VI ADDITIONAL PRE-CLOSING COVENANTS	72
6.2 6.3 6.4 6.5 6.6 6.7 6.8 6.9 6.10	Insurance Employee Benefits Tax Matters Allocation Certificate	72 74 75 76 77 77 78 80 81 81 81
ARTICL	E VII CONDITIONS PRECEDENT TO THE MERGER	82
	Conditions to Obligation of Each Party to Effect the Merger Additional Conditions to Obligations of Parent and Acquisition Sub Additional Conditions to Obligations of the Company	82 82 84
ARTICL	E VIII TERMINATION AND EXPENSES	85
	Effect of Termination	85 86 86
ARTICL	JE IX REPRESENTATIVE	87
9.1 9.2	Authorization of the Representative; Inability to Perform Representative Reimbursement Set-Aside	87 88

9.2 Representative Reimbursement Set-Aside

iii

9.3	Compensation; Exculpation	89
ARTICLE	X INDEMNIFICATION	89
10.1	General Indemnification	89
10.2	Limitations of Indemnification Obligations	91
10.3	Survival and Expiration of Representations and Warranties	92
10.4	Indemnification Procedures	93
10.5	Insurance	94
10.6	No Subrogation	94
10.7	Merger Consideration Adjustment	95
ARTICLE	XI MISCELLANEOUS	95
11.1	Entire Agreement	95
11.2	Amendment and Waiver	95
11.3	No Third-Party Beneficiaries	95
11.4	Assignment	96
11.5	Effect of Waivers	96
11.6	Governing Law; Venue	96
11.7	Equitable Relief	96
11.8	Interpretation	97
11.9	Severability	97
11.10) Notices	97
11.11	Representation by Counsel	98
	2 Construction	98
11.13	3 Calculation of Merger Consideration	98
	Materiality	99
	5 Disclosure Schedules	99
11.16	5 Headings	99
	7 Counterparts	99
	3 Certain Waiver	99
11.19) Conflict Waiver	100

Exhibits	
Exhibit A	[Reserved]
Exhibit B	[Reserved]
Exhibit C	Certificate of Merger
Exhibit D	Paying Agent Agreement
Exhibit E	Letter of Transmittal
Exhibit F	Option Cancellation Agreement
Exhibit G	Escrow Agreement

iv

Schedules Article III Schedule 6.1

Schedule 6.8 Schedule 7.2(c)

Company Disclosure Schedules Permissible Activities Between Signing and Closing Allocation Certificate Company Third Party Consents Required for Closing

AMENDED AND RESTATED

AGREEMENT AND PLAN OF MERGER

This AMENDED AND RESTATED AGREEMENT AND PLAN OF MERGER (the "<u>Agreement</u>"), dated as of February 2, 2014 (the "<u>Agreement</u> <u>Date</u>"), is made by and among **Myriad Genetics, Inc.**, a Delaware corporation ("<u>Parent</u>"); **Myriad Crescendo, Inc.**, a Delaware corporation and whollyowned subsidiary of Parent ("<u>Acquisition Sub</u>"); **Crescendo Bioscience, Inc.**, a Delaware corporation (the "<u>Company</u>"); and the Representative (as defined below) and, upon obtaining the Stockholder Approval, shall become effective and supersede the existing Agreement and Plan of Merger by and among Parent, Acquisition Sub, the Company and the Representative dated as of September 8, 2011 (the "<u>Prior Merger Agreement</u>"), pursuant to which Parent holds an option to acquire the Company and entered into the initial Loan Documents (as defined below) in connection therewith. If the Stockholder Approval shall not have been obtained by 11:59 p.m. Pacific Standard Time on February 7, 2014, or such other date and time as mutually agreed by Parent and the Company, then this Agreement shall be null and void and the Prior Merger Agreement, as modified by that certain letter agreement dated January 13, 2014 between Parent and the Company (the "<u>Letter Amendment</u>"), shall continue in full force and effect.

WHEREAS, in furtherance of such acquisition, the Boards of Directors of Parent, Acquisition Sub and the Company have each approved and declared advisable and in the best interests of their respective stockholders this Agreement and the merger of Acquisition Sub with and into the Company (the "<u>Merger</u>"), in accordance with the General Corporation Law of the State of Delaware (the "<u>DGCL</u>"), upon the terms and subject to the conditions set forth herein, which Merger will result in, among other things, Parent acquiring the Company, with the Company becoming a wholly-owned subsidiary of Parent;

WHEREAS, immediately following execution of this Agreement, the Company will distribute an action by written consent in lieu of a meeting pursuant to which stockholders of the Company representing the affirmative vote of the capital stock of the Company required to approve and authorize the Company's execution and delivery of this Agreement and the Merger will be asked to adopt this Agreement and approve the Merger; and

NOW, THEREFORE, in consideration of the foregoing and the mutual representations, warranties, covenants and agreements herein contained, and intending to be legally bound hereby, the parties hereto agree as follows:

ARTICLE I DEFINITIONS

1.1 Certain Defined Terms.

As used in this Agreement, the following terms shall have the following meanings:

"Action" means any action, suit, litigation, arbitration, claim, demand, complaint, criminal prosecution, governmental or other administrative proceeding, investigation (of which the Company has received written notice), hearing, charge whether at law or at equity, civil or criminal before or by any Court or Governmental Authority.

"Affiliate" means, with respect to any Person, a Person that directly or indirectly, through one or more intermediaries, controls, is controlled by, or is under common control with, the first mentioned Person; and "control" (including the terms "<u>controlled by</u>" and "<u>under common control with</u>") means the possession, directly or indirectly, of the power to direct or cause the direction of the management or policies of a Person, whether through the ownership of stock or other securities, as trustee or executor, by contract or otherwise.

"Business Day" means any day other than a Saturday, Sunday or day on which banks are permitted to close in the State of California or the State of Utah.

"Certificate of Incorporation" means, with respect to any corporation, those instruments that at the time constitute its corporate charter as filed or recorded under the general corporation Law of the jurisdiction of its incorporation, including the articles or certificate of incorporation or organization, and all amendments thereto, as the same may have been restated, and all amendments thereto (including any articles or certificates of merger or consolidation, certificate of correction or certificates of designation or similar instruments which effect any such amendment) which became effective after the most recent such restatement.

"Change of Control Payments" means any amounts (including all related Taxes payable by the Company on such amounts) which become payable by the Company or the Surviving Corporation (other than payments of Merger Consideration pursuant to Section 2.10, 2.12, 2.13 or 2.14) to any current or former employee, officer, director, consultant, independent contractor or other Person on, prior to or after the Closing as a result of or in connection with the consummation of the Merger, whether pursuant to any plan or policy of the Company or any individual employment, severance, change-of-control or other contract or agreement, including, for the avoidance of doubt, all Taxes payable by the Company or the Surviving Corporation with respect to the cancellation of Company Options pursuant to this Agreement; <u>provided</u>, <u>however</u>, that (i) any payments made by the Surviving Corporation based upon agreements made by the Surviving Corporation after the Effective Time that are not pursuant to any contract or agreement between the Company and the recipient in existence prior to the Effective Time, (ii) any payments made as a result of termination of employment or service by Parent or the Surviving Corporation has the President's direct reports as of the Agreement Date, but only to the extent that the total of any such payment amount to any given individual is equal to or less than one (1) times such individual's salary and benefits which such individual was receiving as of the most recently completed calendar year end of the Company prior to the date of termination of employment or service, in the case of each of subclauses (i), (ii) and (ii) of this proviso, shall not be considered Change of Control Payments. For the avoidance of doubt, any payments made as a result of termination of employment or service by Parent or the Surviving Corporation less than one hundred and twenty

(120) days following the Effective Time when made to (i) Persons other than the President of the Company or any of the President's direct reports as of the Agreement Date or (ii) if made to the President of the Company or any of the President's direct reports as of the Agreement Date, then any amount to any such individual greater than one (1) times such individual's salary and benefits which such individual was receiving as of the most recently completed calendar year end of the Company prior to the date of termination of employment or service, shall be treated as Change of Control Payments.

"Closing Date Working Capital" means an amount equal to (i) the Working Capital of the Company as of the Closing Date prepared in accordance with GAAP as of the Closing Date; plus (ii) the aggregate amount of current liabilities incurred by the Company between the Measurement Date and the Closing Date on the acquisition of any asset that is not included in Closing Date Working Capital as a current asset, which current liabilities shall not be added to Closing Date Working Capital if they exceed \$100,000 for any single item or \$250,000 in the aggregate for all such items without the prior written consent of Parent, which consent shall not be unreasonably withheld.

"Closing Option/Warrant Amount" means the sum of the aggregate exercise prices for all In-the-Money Company Options and In-the-Money Company Warrants outstanding and unexercised as of the Effective Time.

"Closing Payment" means (i) \$270,000,000, <u>plus</u> (ii) the total amount of cash and cash equivalents held by the Company on the Closing Date (prior to giving effect to the Closing), as set forth in the Preliminary Closing Balance Sheet, plus (iii) the aggregate cost of Meso Supplies, plus (iv) the amount of the Closing Option/Warrant Amount, <u>plus</u> (v) the Estimated Working Capital Surplus, if any, and minus (vi) the Closing Payment Reduction Amount minus (vii) the Escrow Consideration minus (viii) the Representative Reimbursement Set-Aside.

"Closing Payment Reduction Amount" means the sum (without duplication of components thereof) of (i) the aggregate amount of all outstanding Company Indebtedness (including, without limitation, the Loan Amount), (ii) the aggregate amount of all Change of Control Payments payable by the Company at or concurrent with the Closing, (iii) the aggregate amount of all unpaid Transaction Expenses for services performed on or prior to December 31, 2013, (iv) the aggregate amount of all unpaid IPO Expenses, (v) the aggregate amount of all unpaid Fiscal Year 2013 Bonus Payments, (vi) the aggregate amount of all unpaid expenses payable to Parent or its counsel under the Loan Documents, and (vii) the Estimated Working Capital Deficit, if any.

"Code" means the Internal Revenue Code of 1986, as amended, and all Regulations promulgated thereunder.

"**Commercial Product**" means any Company Product being commercialized, or planned to be commercialized within six months following the Agreement Date, the material aspects of which, have been identified as of the Agreement Date.

"Common Stock" means the Company's shares of common stock, par value \$0.01 per share.

"Company Board" means the Board of Directors of the Company.

"Company Equity Incentive Plans" means the Company's 2005 Stock Option Plan and 2007 Stock Plan and any other equity compensation plan, agreement or arrangement of the Company adopted from time to time and in effect immediately prior to the Effective Time.

"Company Financial Statements" mean, collectively, the Audited Financial Statements and the Interim Financial Statements.

"**Company Holders**" mean those persons (other than the holders of Dissenting Shares) who, immediately prior to the Effective Time, were holders of Common Stock, Company Preferred Stock, Company Warrants, Company Options and any other equity securities of the Company and whose interests therein, as a result of the Merger, are converted into rights to receive Merger Consideration.

"Company Indebtedness" means all Indebtedness of the Company as of the Effective Time.

"**Company Intellectual Property**" means any Intellectual Property owned by, under obligation of assignment to, or licensed to the Company or any of its Subsidiaries, but excluding any generally available, non-customized, third party software licensed to the Company that does not require aggregate payments in any given year in excess of \$40,000 in license, maintenance, royalty and/or other fees.

"Company Material Adverse Change" means any change, circumstance, effect, event, condition, occurrence or fact that (individually or in the aggregate) (i) has a material and adverse change to or effect on the business, operations, assets, liabilities, financial condition, or results of operations of the Company and its Subsidiaries, taken as a whole, or (ii) prevents or materially impedes the ability of the Company to perform its obligations pursuant to this Agreement and to consummate the transactions contemplated hereby, including the Merger, but excluding (in the case of both clause (i) and clause (ii)) any Excluded Company Material Adverse Change.

"**Company Permits**" mean all material permits, licenses, variances, exemptions, orders, franchises and approvals of all Governmental Authorities necessary for the lawful conduct of its business.

"**Company Personnel**" means any current director, officer, employee, or individual acting as an independent contractor or consultant of the Company or its Subsidiaries.

"**Company Product**" means any diagnostic test or clinical laboratory service being researched, developed or offered by the Company or its Subsidiaries, alone or in collaboration with another Person, as of the Agreement Date.

"Company Registered Intellectual Property" means all of the Registered Intellectual Property that is owned by, exclusively or non-exclusively licensed by, under obligation of assignment to, or filed in the name of the Company or any of its Subsidiaries.

"Company Shares" mean, collectively, the shares of Company Preferred Stock and Common Stock.

"**Consents**" mean all authorizations, consents, orders or approvals of, or registrations, declarations or filings with, or expiration of waiting periods imposed by, any Governmental Authority, in each case, that are necessary in order to consummate the transactions contemplated by this Agreement and any Related Agreement, and all consents and approvals of third parties necessary to prevent any conflict with, violation or breach of, or default under, the Material Contracts, as the case may be.

"Contract" or "contract" means any written or oral contract, agreement, license, lease or other binding instrument, agreement or arrangement, and all written amendments, modifications and supplements thereto.

"Court" means any court or arbitration tribunal of the United States, any domestic state, or any foreign country, and any political subdivision thereof.

"Deductible" means \$1,350,000.

"Development Product" means any Company Product that is not a Commercial Product but the material aspects of which have been identified as of the Agreement Date.

"**Dissenting Share Payments**" means (x) any payment in respect of Dissenting Shares in excess of the Merger Consideration that otherwise would have been payable in respect of such shares in accordance with this Agreement and (y) any costs or expenses (including reasonable attorneys' fees, costs and expenses in connection with any action or proceeding or in connection with any investigation) in respect of any Dissenting Shares.

"Environmental Law" means any Laws relating to Hazardous Materials, pollution, protection of human health, safety or the environment or natural resources, including Laws related to the presence of, or any Remedial Action taken in relation to, a Hazardous Material in the soil or other subsurface media, air or any body of water, including but not limited to, any ground water, surface water or aquifer, including the Comprehensive Environmental Response, Compensation and Liability Act (42 U.S.C. § 9601 et seq.), the Hazardous Materials Transportation Act (49 U.S.C. App. § 1801 et seq.), the Resource Conservation and Recovery Act (42 U.S.C. § 6901 et seq.), the Clean Water Act (33 U.S.C. § 1251 et seq.), the Clean Air Act (42 U.S.C. § 7401 et seq.) the Toxic Substances Control Act (15 U.S.C. § 2601 et seq.), the Federal Insecticide, Fungicide, and Rodenticide Act (7 U.S.C. § 136 et seq.), and other similar state and local statutes, as each has been amended and the Regulations promulgated or issued pursuant thereto.

"ERISA" means the Employee Retirement Income Security Act of 1974, as amended.

"Escrow Consideration" means an amount in cash equal to Twenty Million Dollars (\$20,000,000).

"Escrow Termination Date" means the date that is sixty (60) days following completion of the audit of the Company's financial statements, by auditors selected by Parent, for the fiscal year of the Company ending December 31, 2014; <u>provided</u> that, notwithstanding the status of such audit, the Escrow Termination Date shall not be less than twelve (12) months nor more than (15) months following the Closing Date.

"European Union" or "EU" means the countries of the European Union, as the European Union is then constituted.

"Excluded Company Material Adverse Change" means any Company Material Adverse Change that is caused by, arises out of or relates to (i) changes in general economic conditions or securities or financial markets generally, (ii) changes in general political conditions, including the effect of any outbreak of hostilities, acts of war, sabotage or terrorism or military actions or any escalation or material worsening of any such hostilities, acts of war, sabotage or terrorism or military actions existing or underway as of the Agreement Date, (iii) changes in accounting requirements or principles imposed upon the Company or its business or any change in GAAP, or the interpretation thereof after the Agreement Date, (iv) changes that generally affect the industry in which the Company operates, (v) changes in Laws or other binding directives issued by any Governmental Authority after the Agreement Date, (vi) the effect of any failure by the Company to meet internal forecasts or projections (it being understood, however, that the facts or circumstances giving rise to any such failure to meet internal forecasts or projections may be taken into account in determining whether there has been a Company Material Adverse Change), (vii) the execution and delivery of this Agreement and the other Related Agreements, or the announcement, disclosure or pendency of the transactions contemplated by this Agreement or any other Related Agreement, so long as in each of the cases set forth in clauses (i) through (v) inclusive, the Company is not disproportionately affected by such changes as compared to other companies in the Company's industry.

"Fiscal Year 2013 Bonus Payments" means any amounts payable to Company employees as a bonus for services rendered during 2013, whether pursuant to any plan or policy of the Company or any individual employment or other agreement, including all related Taxes or social charges payable by the Company or the Surviving Corporation on such amounts.

"FDA" means the U.S. Food and Drug Administration, or any successor thereto.

"Foreign Privacy Laws" means (i) the Directive 95/46/EC of the European Parliament and of the Council of the European Union of 24 October 1995 on the protection of individuals with regard to the collection, use, disclosure, and processing of personal data and on the free movement of such data, (ii) the corresponding national rules, regulations, codes, orders, decrees and rulings thereunder of the member states of the European Union, and (iii) any rules, regulations, codes, orders, decree, and rulings thereunder related to privacy, data protection or data transfer issues implemented in other countries (including without limitation Canada).

"**Fully-Diluted Common Share Number**" means the sum of the total number of (i) shares of Common Stock, (ii) shares of Common Stock issuable upon exercise of any unexercised vested In-the-Money Company Options or In-the-Money Company Warrants exercisable for shares of Common Stock, (iii) shares of Common Stock issuable upon conversion

of any convertible notes that are convertible into shares of Common Stock and for which the holders thereof have not declared the outstanding balance due and payable in cash as of the Effective Time, (iv) to the extent a conversion will be effected in connection with the transactions contemplated by this Agreement, shares of Company Preferred Stock that will be converted into shares of Common Stock (including any shares of Company Preferred Stock that will be issued upon exercise of In-the-Money Company Warrants or conversion of convertible notes for shares of Company Preferred Stock and, prior to the Effective Time, converted into shares of Common Stock), calculated on an as converted to Common Stock basis, and (v) shares of the Company's Series D Preferred Stock (the "<u>Series D Stock</u>"), calculated on an as converted to Common Stock basis, in each case as issued and outstanding immediately prior to the Effective Time.

"Fully-Diluted Preferred Shares" means (i) shares of Company Preferred Stock, (ii) shares of Company Preferred Stock issuable upon exercise of Inthe-Money Company Warrants exercisable for shares of Company Preferred Stock, and (iii) shares of Company Preferred Stock issuable upon conversion of any convertible notes that are convertible into shares of Company Preferred Stock and for which the holders thereof have not declared the outstanding balance due and payable in cash as of the Effective Time, in each case as issued and outstanding immediately prior to the Effective Time.

"GAAP" means U.S. generally accepted accounting principles consistently applied and maintained throughout the periods indicated.

"Government Payment Programs" means Medicare, TRICARE, Medicaid, Workers' Compensation, and any and all health maintenance organizations, preferred provider organizations, health benefit plans, health insurance plans, and other third party reimbursement and payment programs administered by a Governmental Authority.

"Governmental Authority" means any governmental agency, authority, department, commission, board, bureau, Court, tribunal, judicial or arbitral body or instrumentality of the United States, the European Union, any domestic state, any foreign country, or any federation or union of any of the forgoing, and any political subdivision or agency thereof, and includes any authority having governmental or quasi-governmental powers, including any administrative agency or commission.

"Hazardous Material" means any substance, material, chemical, pollutant, contaminant or waste that is regulated, classified, or otherwise characterized under or pursuant to or forming the basis of liability under any Environmental Law, including petroleum (including crude oil or any fraction thereof) and petroleum products, asbestos in any form, urea formaldehyde, percholorate, and polychlorinated biphenyls.

"**Health Care Laws**" means the following Laws and their implementing Regulations: (i) the Federal Health Care Program Anti-Kickback Statute, 42 U.S. C. § 1320a-7b(b), and any anti-kickback Law of a state; (ii) the Federal False Claims Act, 31 U.S.C. § 3729, and any false claim or fraud Law of a state; (iii) the Federal Civil Monetary Penalties Law, 42 U.S.C. § 1320a-7a; (iv) the Federal Ethics in Patient Referrals Act, 42 U.S.C. § 1395nn et seq., and any physician self-referral Law of a state; (v) applicable privacy and security Laws regarding protected health

information under HIPAA, and all applicable state and federal Laws regarding the privacy and security of protected health information under HIPAA and other confidential patient information and personal data; (vi) the Medicare Act, 42 U.S.C. § 1395 et seq.; (vii) 42 U.S.C. §1320a-7; (viii) 42 U.S.C. §1320c-5; and (ix) the Clinical Laboratory Improvement Amendments ("<u>CLIA</u>"), 42 U.S.C. § 263a, and any applicable state laboratory licensure Law.

"HIPAA" means the Health Insurance Portability and Accountability Act of 1996, as amended by the Health Information Technology for Economic and Clinical Health Act, and its implementing Regulations.

"Indebtedness" means any Liabilities (i) for borrowed money (other than deposits, advances or excess payments accepted in connection with the sale of products or services in the ordinary course of business) or extensions of credit (including bank advances and overdrafts); (ii) evidenced by bonds, debentures, notes or similar instruments, including the Loan Amount, but excluding any dividends payable on Company Preferred Stock that are paid to the holders thereof as part of the Merger Consideration and the amount due under any convertible notes of the Company outstanding as of the Effective Time for which the holders thereof have not declared the outstanding balance due and payable in cash as of the Effective Time and which will be paid as part of the Merger Consideration, (iii) under leases required to be accounted for as capital leases under GAAP, (iv) to pay the deferred purchase price of property or services, except trade accounts payable arising in the ordinary course of business, (v) of others secured by a Lien on any asset of the Company, whether or not such Liabilities are assumed, (vi) contingent or otherwise, directly or indirectly guaranteeing any obligations of any other Person, all obligations to reimburse the issuer in respect of letters of credit or under performance or surety bonds, or other similar obligations, (vii) in respect of bankers' acceptances and under reverse repurchase agreements, (viii) in respect of futures contracts, swaps, other financial contracts and other similar obligations (determined on a net basis as if such contract or obligation was being terminated early on such date), (ix) for lease incentive obligations, or (x) other than trade accounts payable arising in the ordinary course of business and included in the calculation of Working Capital, any other Liabilities of the Company in an amount greater than One Hundred Thousand Dollars (\$100,000) or that are payable in more than one year; including, for the avoidance of doubt, all interest, premium and pre-payment penalties pa

"Intellectual Property" means any or all of the following and all rights in, arising out of, or associated therewith: (i) all United States, international and foreign patents and applications therefor and all reissues, divisions, divisionals, renewals, extensions, provisionals, continuations and continuations-inpart thereof, and all patents, applications, documents and filings claiming priority to or serving as a basis for priority thereof; (ii) all inventions (whether or not patentable), invention disclosures, improvements, trade secrets, proprietary information, know how, computer software programs (in both source code and object code form), technology, business methods, and technical data, and tangible or intangible proprietary information; (iii) all copyrights, copyrights registrations and applications therefor, and all other rights corresponding thereto throughout the world; (iv) all industrial designs and any registrations and applications therefor throughout the world; (v) all trade names, logos, common law trademarks and service marks, trademark and service mark registrations and applications therefor throughout the world; (vi) all databases and data collections and all rights therein throughout the world; (vii) all moral and economic rights of authors and inventors, however denominated, throughout the world; (viii) all Web addresses, sites and domain names and numbers; and (ix) any similar or equivalent rights to any of the foregoing anywhere in the world.

"**IPO Expenses**" means all fees, costs and expenses incurred by or on behalf of the Company, whether or not invoiced or payable as of the Closing, arising out of, relating to, in connection with, or incidental to, the Company's proposed initial public offering of its Common Stock, including fees and disbursements of advisors, consultants, investment bankers and other financial advisors, brokers and finders, lawyers and accountants, to the extent such fees, costs and expenses relate to services rendered in connection with the Company's proposed initial public offering of its Common Stock.

"Judgment" means any judgment, order, decree, writ, injunction, ruling or any settlement under the jurisdiction of any Court or Governmental Authority.

"Knowledge of the Company" and all permutations thereof means the actual knowledge, after review of this Agreement and the Company Disclosure Schedule, of the Chief Executive Officer, the Chief Medical Officer, the Vice President, Bioinformatics, the Chief Technology Officer, the Chief Commercial Officer, the Vice President, Business Development, the Chief Financial Officer, the Chief Information Officer, General Counsel or Chief Legal Officer, director of intellectual property or chief IP counsel, and any other person who after the Agreement Date holds any of the foregoing offices at the Company, or the person(s) at the Company whose primary responsibilities include similar functions.

"Law" means any law, statute, code, common law, licensing requirement, ordinance, and Regulation of any Governmental Authority.

"Liabilities" means any damages, debts, obligations and other liabilities (whether known or unknown, absolute or contingent, liquidated or unliquidated, due or to become due, accrued or not accrued, asserted or unasserted or otherwise), losses, claims, royalties, Taxes, interest obligations, deficiencies, Judgments, assessments, fines, fees, penalties, expenses (including amounts paid in settlement, interest, Court costs, costs of investigators, reasonable fees and expenses of attorneys, accountants, financial advisors, consultants and other experts, and other expenses of litigation).

"Lien" means any mortgage, pledge, license, security interest, attachment, easement, encumbrance, lien (statutory or otherwise), option, conditional sale agreement, right of first refusal or right of first offer (including any agreement to give any of the foregoing).

"Loan Amount" means the aggregate amount outstanding under the Loan Documents as of the Closing.

"**Loan Documents**" means that certain Loan and Security Agreement, by and between Parent and the Company, dated as of September 8, 2011, as amended most recently on January 14, 2014 and as may be further amended prior to Closing as contemplated by Section 5.13 of this Agreement, together with all agreements, certificates and other documents delivered in connection therewith.

"made available," as used in Article III in this Agreement with respect to documents and other diligence materials made available to Parent, means delivered to Parent or posted in the electronic data room for this transaction prior to January 28, 2014.

"Merger Consideration" means the sum of the Closing Payment and the portion of the Escrow Consideration (if any), the Working Capital Surplus (if any), and the Representative Reimbursement Set-Aside (if any) distributed to the Company Holders.

"Meso Supplies" means plates and diluents used in the manufacturing process to support finished product for Crescendo's CLIA laboratory purchased from Meso Scale Diagnostics, LLC since January 12, 2014 and not consumed as of the Closing Date.

"Neutral Auditor" means a nationally recognized accounting firm of independent auditors mutually agreed upon by Parent and Representative that has not performed work for (other than as a neutral auditor), and is otherwise independent of, each of Parent, the Company and the Representative.

"**Pay-Off Letters**" means the letters, and any updates thereto, to be sent by each lender of the Company and its Subsidiaries to the Company and Parent prior to Closing, other than with respect to the Loan Amount, which letters shall among other things specify the aggregate amount of Indebtedness that will be outstanding as of the Effective Time under each note, loan agreement or other similar instrument and wire transfer information for each such lender to be paid at Closing.

"**Per Share Common Closing Payment**" means the amount equal to the quotient obtained by dividing (x) the greater of (i) zero and (ii) the amount of the Closing Payment minus the aggregate amount of the Per Share Preferred Closing Payments payable in respect of all Fully-Diluted Preferred Shares, by (y) the Fully-Diluted Common Share Number, as shown in the Allocation Certificate.

"**Per Share Common Escrow Payment**" means the amount equal to the quotient obtained by dividing (x) the greater of (i) zero and (ii) the amount of the Escrow Consideration (and any interest thereon) that becomes distributable to the Company Holders pursuant to the terms and conditions of the Escrow Agreement (if any) minus the aggregate amount of Per Share Preferred Escrow Payments, by (y) the Fully-Diluted Common Share Number, as shown in the Allocation Certificate.

"**Per Share Common Reimbursement Payment**" means the amount equal to the quotient obtained by dividing (x) the greater of (i) zero and (ii) the amount of the Representative Reimbursement Set-Aside that becomes distributable to the Company Holders (if any) minus the aggregate amount of Per Share Preferred Reimbursement Payments, by (y) the Fully-Diluted Common Share Number, as shown in the Allocation Certificate.

"**Per Share Common Working Capital Payment**" means the amount equal to the quotient obtained by dividing (x) the greater of (i) zero and (ii) the amount of the Working Capital Surplus (if any) minus the aggregate amount of Per Share Preferred Working Capital Payments, by (y) the Fully-Diluted Common Share Number, as shown in the Allocation Certificate.

"**Per Share Preferred Closing Payment**" means, with respect to each Fully-Diluted Preferred Share, the portion of the Closing Payment allocable to such share in preference to holders of Common Stock, in accordance with the Company's Certificate of Incorporation as in effect immediately prior to the Effective Time, as shown in the Allocation Certificate.

"**Per Share Preferred Escrow Payment**" means, with respect to each Fully-Diluted Preferred Share, the portion (if any) of the Escrow Consideration (and any interest thereon) that becomes distributable to the Company Holders and is allocable to such share in preference to holders of Common Stock, in accordance with the Company's Certificate of Incorporation as in effect immediately prior to the Effective Time, as shown in the Allocation Certificate.

"**Per Share Preferred Reimbursement Payment**" means, with respect to each Fully-Diluted Preferred Share, the portion (if any) of the Representative Reimbursement Set-Aside that becomes distributable to the Company Holders and is allocable to such share in preference to holders of Common Stock, in accordance with the Company's Certificate of Incorporation as in effect immediately prior to the Effective Time, as shown in the Allocation Certificate.

"**Per Share Preferred Working Capital Payment**" means, with respect to each Fully-Diluted Preferred Share, the portion (if any) of the Working Capital Surplus allocable to such share in preference to holders of Common Stock, in accordance with the Company's Certificate of Incorporation as in effect immediately prior to the Effective Time, as shown in the Allocation Certificate.

"**Permitted Liens**" means (i) statutory Liens for Taxes, assessments and other governmental charges not yet subject to penalties for non-payment or which are being contested in good faith by appropriate proceedings, (ii) statutory or common law Liens to secure landlords, sublandlords, licensors or sublicensors under leases or rental agreements, (iii) deposits or pledges made in connection with, or to secure payment of, workers' compensation, unemployment insurance, old age pension or other social security programs mandated under applicable Laws, (iv) statutory or common law Liens in favor of carriers, warehousemen, mechanics, workmen, repairmen and materialmen to secure claims for labor, materials or supplies and other like Liens, (v) restrictions on transfer of securities imposed by applicable state and federal securities Laws, (vi) any other encumbrance affecting any asset which does not materially impede or otherwise materially affect the ownership, operation, use or value of such asset, or (vii) Liens resulting from a filing by a lessor as a precautionary filing for a true lease.

"Person" means an individual, corporation, partnership, association, trust, unincorporated organization, limited liability company or other legal entity.

"**Personal Information**" means any information related to an identified or identifiable natural person that does not meet the definition of de-identified as defined by section 164.514(b)(2) of HIPAA or as defined by Foreign Privacy Laws.

"Plan Affiliate" means any Person with which the Company constitutes all or part of a controlled group of corporations, a group of trades or businesses under common control or an affiliated service group, as each of those terms are defined in Section 414 of the Code.

"**Pre-Closing Tax Period**" means any Tax period ending on or prior to the Closing Date, and for any period that includes (but does not end on) the Closing Date (a "<u>Straddle Period</u>"), the portion of the Straddle Period that ends on the Closing Date, with Taxes for such Straddle Period allocated to the Pre-Closing Tax Period portion thereof in accordance with Section 6.3(g).

"Privacy Laws" means all U.S. Privacy Laws and Foreign Privacy Laws.

"**Publicly Available Software**" means each of (i) any software that contains, or is derived in any manner (in whole or in part) from, any software that is distributed as free software, open source software (<u>e.g.</u>, GNU General Public License, Apache Software License, MIT License), or pursuant to similar licensing and distribution models and (ii) any software that requires as a condition of use, modification, hosting, and/or distribution of such software, or of other software used or developed with, incorporated into, derived from, or distributed with such software, that such software or other software (A) be disclosed or distributed in source code form; (B) be licensed for the purpose of making derivative works; (C) be redistributed, hosted or otherwise made available at no or minimal charge; or (D) be licensed, sold or otherwise made available on terms that (x) limit in any manner the ability to charge license fees or otherwise seek compensation in connection with marketing, licensing or distribution of such software or other software or (y) grant the right to decompile, disassemble, reverse engineer or otherwise derive the source code or underlying structure of such software or other software.

"**Registered Intellectual Property**" means all United States, international and foreign: (i) patents and patent applications (including provisional applications and design patents and applications) and all reissues, divisions, divisionals, renewals, extensions (including but not limited to supplemental protection certificates), counterparts, continuations and continuations-in-part thereof, and all patents, applications, documents and filings claiming priority thereto or serving as a basis for priority thereof; (ii) registered trademarks, service marks, applications to register trademarks, applications to register trademarks, applications for copyright registrations, or other registrations or applications related to trademarks; (iii) registered copyrights and applications for copyright registration; (iv) domain name registrations and Internet number assignments; and (v) any other Intellectual Property that is the subject of an application, certificate, filing, registration or other document issued, filed with, or recorded by any Governmental Authority.

"Regulation" means any rule or regulation of any Governmental Authority.

"**Regulatory Authority**" means the FDA and comparable national, supra-national, regional, state or local regulatory agencies to the extent they have authority over the ultimate approval and regulation of the sale of a Company Product.

"**Release**" means any release, spill, emission, leaking, pumping, pouring, injection, deposit, dumping, emptying, disposal, discharge, escaping or leaching into the environment.

"**Remedial Action**" means all actions including any capital expenditures undertaken to (i) clean up, remove, treat or in any other way address any Hazardous Material; (ii) prevent the Release or threat of Release, or minimize the further Release of any Hazardous Material; (iii) perform pre-remedial studies and investigations or post-remedial monitoring and care; or (iv) correct a condition of noncompliance with Environmental Laws.

"Representative" means MDV IX, L.P. or its successor appointed pursuant to Section 9.1(c) of this Agreement.

"**Representative Reimbursement Set-Aside**" means an amount in cash equal to \$200,000 or such larger amount as may be approved prior to the Closing by the Company Holders who, as of the Agreement Date, hold at least a majority of the outstanding Company Shares (on a fully-diluted basis) held by all Company Holders.

"Revenue" means the consolidated revenue of the Company and its Subsidiaries, calculated in accordance with GAAP.

"Straddle Period" is defined in the definition of "Pre-Closing Tax Period" above.

"Subsidiary" means, with respect to any Person, any corporation, partnership, joint venture, limited liability company, trust or other legal entity of which such Person (either alone or through or together with any other Subsidiary) owns, directly or indirectly, at least a majority of the stock or other equity interests in such entity.

"**Tax Authority**" means any Governmental Authority or any quasi-governmental or private body having jurisdiction over the assessment, determination, collection or imposition of any Tax.

"**Taxes**", and with correlative meaning, "Taxable" and "Taxing", mean all taxes and governmental impositions of any kind in the nature of (or similar to) taxes, payable to any federal, state, local or foreign taxing authority or other Governmental Authority, including, but not limited to, those on or measured by or referred to as income, franchise, profits, gross receipts, capital, net worth, ad valorem, custom duties, alternative or add-on minimum taxes, estimated, environmental, disability, registration, value added, sales, use, service, real or personal property, capital stock, license, payroll, withholding, employment, social security, workers' compensation, unemployment compensation, utility, severance, production, excise, stamp, occupation, premiums, windfall profits, transfer and gains taxes, and interest, penalties and additions to tax imposed with respect thereto.

"**Tax Return**" means any report, return, documents, declaration or other information (and any supporting schedules, estimates, elections, statements or attachments thereto) required to be supplied to any Tax Authority or jurisdiction with respect to Taxes (including any returns or reports filed on a consolidated, unitary, or combined basis).

"**Transaction Expenses**" means all fees, costs and expenses incurred by or on behalf of the Company, whether or not invoiced or payable as of the Closing, arising out of, relating to, in connection with, or incidental to, the discussion, evaluation, negotiation, documentation, preparation, implementation, consummation and performance of the Prior Merger Agreement and this Agreement and the transactions contemplated hereby, including costs associated with securing required consents, and fees and disbursements of advisors, consultants, investment bankers and other financial advisors, brokers and finders, lawyers and accountants, to the extent such fees, costs and expenses relate to services rendered in connection with the consummation of the transactions contemplated by this Agreement at or prior to the Effective Time.

"U.S. Privacy Laws" means any U.S. federal, state, regional or local rules, regulations, codes, orders, decrees, and rulings or any Governmental Authority obtaining authority from any of the foregoing that relate to privacy, security, data protection or data transfer issues, including all implementing laws, ordinances or regulations.

"Voting Holders" means the holders of shares of Preferred Stock and Common Stock outstanding as of the Agreement Date.

"Working Capital" means an amount, which may be positive or negative, equal to: (i) the sum of the Company's current assets, which shall include accounts receivable (net of reserves), deposits on leased property, inventory (net of reserves), prepaid expenses and other current assets but which shall not include the Meso Supplies; less (ii) the sum of the Company's current liabilities, including accrued expenses and accounts payable (including unpaid Transaction Expenses for services performed after December 31, 2013); in each case, calculated in accordance with GAAP; provided, however, that, for purposes of this definition of Working Capital, whether or not in accordance with GAAP, (A) Tax assets, including deferred Tax assets, shall not be taken into account under clause (i), (B) cash and cash equivalents shall not be taken into account under clause (i), and (C) "current liabilities" shall exclude (1) unpaid Company Indebtedness (including, without limitation, the Loan Amount), (2) the amount of any Change of Control Payments, (3) the amount of unpaid Transaction Expenses for services performed on or prior to December 31, 2013, (4) the amount of unpaid IPO Expenses, (5) the amount of unpaid Fiscal Year 2013 Bonus Payments, and (6) the amount of unpaid expenses payable to Parent or its counsel under the Loan Documents, in each case to the extent that such amounts are included in the calculation of the Closing Payment Reduction Amount, (7) payments made by Parent or the Surviving Corporation following the Closing that would be Change of Control Payments but for a specific exclusion contained in the definition thereof, and (8) any liabilities that would otherwise be included as current liabilities with respect to outstanding Company Preferred Stock or Warrants exercisable for shares of Company Preferred Stock which shares of Preferred Stock or Warrants will be cancelled in exchange for the payment of Merger Consideration pursuant to Section 2.10(b) or Section 2.14 of this Agreement.

"Working Capital Target" means an amount equal to the Working Capital of the Company as of December 31, 2013 calculated on a basis consistent with the Company's preparation of its consolidated balance sheets and income statements for each month-end (the "Measurement Date").

1.2 Additional Defined Terms.

The following terms shall have the meanings set forth on the pages of this Agreement indicated below:

280G Approval 280G Payments Acquisition Proposal

14

6.3(c)

6.3(c)

5.11

Acquisition Sub	Preamble
Agreed Claims	10.4(g)
Agreement	Preamble
Agreement Date	Preamble
Allocation Certificate	6.8
Antitrust Filings	6.2(a)
Audited Balance Sheet	3.9(b)
Audited Balance Sheet Date	3.9(b)
Audited Financial Statements	3.9(b)
California Stockholder Approval	3.8
Certificate of Merger	2.4
Claims	10.4
CLIA	10.4
Closing	2.4
Closing Date	2.4
Closing Working Capital Estimate	2.4 2.17(a)
Company	Preamble
Company Board Approval	3.5
Company Certificate	
	2.10(e)
Company Disclosure Schedule	3.0
Company Disclosure Schedules	3.0
Company Option	2.13(a)
Company Preferred Stock	2.10(b)
Company Representative	5.11
Company Software	3.18(1)
Company Third Party Consents	6.3 <i>(b)</i>
Company Warrant	2.14(a)
Company Warrant Documents	2.12(a)
Confidentiality Agreement	5.9
Corporate Stockholder Approval	3.8
D&O Cap	6.4(b)
Delivered Tax Returns	3.22(a)
DGCL	Preamble
Disclosure Statement	5.8(a)
Dissenting Shares	2.15
Dissenting Stockholders	2.15
DOJ	6.2(a)
Effective Time	2.4
Employee Pension Benefit Plan	3.15(h)
Employee Plan	3.15(h)
Employee Welfare Benefit Plan	3.15(h)
Environmental Approvals	3.21(e)
Environmental Reports	3.21(f)
Escrow Agent	2.16
Escrow Agreement	2.16

Estimated Working Capital Surplus	2.17(b)
FDA Requirements	3.19(a)
Final Working Capital	2.17(e)
Firm	11.9
FTC	6.2(a)
Fundamental Representations	10.2(a)(ii)
HSR Act	6.2(a)
Indemnified D&Os	6.4(a)
Indemnified Parties	10.1(a)
Interim Balance Sheet	3.9(b)
Interim Balance Sheet Date	3.9(b)
Interim Financial Statements	3.9(b)
In-the-Money Company Option	2.13(a)
In-the-Money Company Warrant	2.14(a)
Leased Property	3.14(a)
Letter Amendment	Premable
Letter of Transmittal	2.12(a)
License Agreement	3.16(a)(xi)
License Agreements	3.16(a)(xi)
Majority Effective Time Holders	9.1(c)
Measurement Date	1.1
Merger	Preamble
Option Cancellation Agreement	2.13(c)
Organizational Documents	3.3
Out-of-the-Money Company Option	2.13(b)
Out-of-the-Money Company Warrant	2.14(b)
Outside Date	8.1(b)
Parent	Preamble
Parent Closing Balance Sheet	2.17(c)
Parent Representatives	5.9
Parent Working Capital Determination	2.17(c)
Parent Working Capital Statement	2.17(c)
Paying Agent	2.11(a)
Paying Agent Agreement	2.11(a)
Payment Fund	2.11(a)
Pre-Closing Period	6.1
Preliminary Closing Balance Sheet	2.17(a)
Preliminary Closing Statement	2.17(a)
Prior Merger Agreement	Preamble
Related Agreements	3.5(a)
Seller Group	11.19
Series D Stock	1.1
Special Representations Specified Representations	10.2(a)(iii) 10.2(a)(iii)
Specified Representations	10.2(a)(iii)
Stark Law Stackholder Approval	3.12(b)
Stockholder Approval	3.8

Stockholder Meeting Straddle Period Subsidiary Securities Survival Date Third Party Claim Working Capital Deficit Working Capital Objection Period Working Capital Objection Statement	5.8(b) 1.1 3.2(b) 10.3(a) 10.4(b) 2.17(f) 2.17(e) 2.17(e) 2.17(f)
Working Capital Surplus	2.17(e) 2.17(f)

2.1 [<u>Reserved</u>].

2.2 [<u>Reserved</u>].

2.3 <u>The Merger</u>. Upon the terms, and subject to the conditions, set forth in this Agreement, and in accordance with the DGCL, Acquisition Sub shall be merged with and into the Company at the Effective Time. From and after the Effective Time, the separate corporate existence of Acquisition Sub shall cease and the Company, as the surviving corporation in the Merger, shall continue its existence under the Laws of the State of Delaware as a wholly-owned Subsidiary of Parent. The Company, as the surviving corporation after the Merger, is hereinafter sometimes referred to as the "Surviving Corporation."

2.4 <u>Effective Time</u>. Subject to the terms and conditions hereof, the closing of the Merger and the transactions contemplated by this Agreement (the "<u>Closing</u>") will take place following the date on which all of the conditions set forth in Article VII have been satisfied or, if permissible, waived, at the offices of Mintz, Levin, Cohn, Ferris, Glovsky and Popeo, P.C., One Financial Center, Boston, Massachusetts 02111, at 10:00 a.m. (Pacific Time) or remotely via the exchange of executed documents and other closing deliverables, unless another time or date is agreed to in writing by the Company and Parent; <u>provided</u>, <u>however</u>, in the event that the Closing would be consummated on any date which is during the last five Business Days of any month following satisfaction or waiver of the conditions set forth in Article VII, Parent may elect to have the Closing take place on the last calendar day of such month in which such conditions are satisfied or waived or such other date as mutually agreed by Parent and the Company. The date on which the Closing actually occurs is referred to herein as the "<u>Closing Date</u>"). On the Closing Date, subject to the terms and conditions set forth in this Agreement, the parties shall cause the Merger to be consummated by filing with the Secretary of State of the State of Delaware a certificate of merger (the "<u>Certificate of Merger</u>") in substantially the form of **Exhibit C** attached hereto and executed in accordance with the relevant provisions of the DGCL (the date and time of the acceptance of such filing, or such later date and time as may be specified in the Certificate of Merger by mutual agreement of Parent and the Company, being the "<u>Effective Time</u>").

2.5 Effect of the Merger. The Merger shall have the effects set forth in Section 259 of the DGCL.

2.6 <u>Certificate of Incorporation and Bylaws of the Surviving Corporation</u>. At the Effective Time and without further action on the part of the parties hereto, the Certificate of Incorporation and the bylaws of the Surviving Corporation shall be amended to read in their entirety to contain the provisions set forth in the Certificate of Incorporation and bylaws of Acquisition Sub, as in effect immediately prior to the Effective Time, in each case, until thereafter amended as provided by the DGCL.

2.7 <u>Directors and Officers</u>. From and after the Effective Time, (a) the directors of Acquisition Sub immediately prior to the Effective Time shall be the initial directors of the Surviving Corporation, and (b) the officers of Acquisition Sub immediately prior to the Effective Time shall be the initial officers of the Surviving Corporation, in each case, to hold office in accordance with the Certificate of Incorporation and the bylaws of the Surviving Corporation until their respective successors are duly elected or appointed and qualified or until their earlier death, resignation or removal in accordance with the Surviving Corporation's Certificate of Incorporation and bylaws or the terms of any Contract pursuant to which they may be serving as such.

2.8 <u>Taking of Necessary Action</u>; <u>Further Action</u>. If, at any time after the Effective Time, any further action is necessary or desirable to carry out the purposes of this Agreement and to vest the Surviving Corporation with full right, title and possession to all assets, property, rights, privileges, powers and franchises of the Company and Acquisition Sub, the officers and directors of the Company and Acquisition Sub are fully authorized in the name of their respective corporations or otherwise to take all such lawful and necessary action.

2.9 [<u>Reserved</u>].

2.10 Effect of Merger on Capital Stock.

(a) <u>Effect of Merger on Capital Stock of Acquisition Sub</u>. At the Effective Time, each share of Acquisition Sub's common stock, par value \$.001 per share, that is issued and outstanding immediately prior to the Effective Time, shall, by virtue of the Merger and without any action on the part of Parent, Acquisition Sub, the Company or the Company Holders, be converted automatically into and exchanged for one share of common stock of the Surviving Corporation.

(b) <u>Conversion of Preferred Stock of the Company</u>. At the Effective Time, by virtue of the Merger and without any action on the part of the parties hereto, the Company Board or the holders of the Company's capital stock, each share of preferred stock of the Company (the "<u>Company Preferred Stock</u>") issued and outstanding immediately prior to the Effective Time, including, without limitation, all shares of Company Preferred Stock issuable upon exercise of Company Warrants for Company Preferred Stock and all shares of Company Preferred Stock issuable upon conversion of any convertible notes that are convertible into Company Preferred Stock and for which the holders thereof have not declared the outstanding balance due and payable in cash as of the Effective Time (whereupon such convertible notes shall be cancelled at the Effective Time), and excluding the Dissenting Shares, shall, by virtue of the Merger and without any action on the part of the holder thereof, be canceled and extinguished and converted automatically, at the Effective Time, into the right to receive, subject to Section 2.10(f) and 2.12 of this Agreement and without interest thereon:

(i) the Per Share Preferred Closing Payment associated with such class or series of Company Preferred Stock <u>plus</u>, in the case of each share of Series D Stock, the Per Share Common Closing Payment, payable in cash as provided in Section 2.11;

(ii) the Per Share Preferred Working Capital Payment (if any) plus, in the case of each share of Series D Stock, the Per Share Common Closing Working Capital Payment (if any), at such time and only to the extent distributable to the Company Holders pursuant and subject to the provisions of Section 2.17, payable in cash as provided in Section 2.11;

(iii) the Per Share Preferred Escrow Payment (if any) <u>plus</u>, in the case of each share of Series D Stock, the Per Share Common Escrow Payment (if any), at such time and only to the extent it becomes distributable to the Company Holders pursuant to the terms and conditions of the Escrow Agreement; and

(iv) the Per Share Preferred Reimbursement Payment (if any) <u>plus</u>, in the case of each share of Series D Stock, the Per Share Common Reimbursement Payment (if any), at such time and only to the extent it becomes distributable to the Company Holders pursuant to the terms and conditions of Section 9.2.

(c) <u>Conversion of Common Stock of the Company</u>. At the Effective Time, by virtue of the Merger and without any action on the part of the parties hereto, the Company Board or the holders of the Company's capital stock, each share of Common Stock that is issued and outstanding immediately prior to the Effective Time, including, without limitation, all shares of Common Stock issuable upon exercise of any Company Options or Company Warrants for Common Stock and all shares of Common Stock issuable upon conversion of any convertible notes that are convertible into Common Stock and for which the holders thereof have not declared the outstanding balance due and payable in cash as of the Effective Time (whereupon such convertible notes shall be cancelled at the Effective Time), and excluding the Dissenting Shares, shall, by virtue of the Merger and without any action on the part of the holder thereof, be canceled and extinguished and converted automatically, at the Effective Time, into the right to receive, subject to Section 2.10(f) and 2.12 of this Agreement and without interest thereon:

(i) the Per Share Common Closing Payment (if any) payable in cash as provided in Section 2.11;

(ii) the Per Share Common Working Capital Payment (if any) at such time and only to the extent it becomes distributable to the Company Holders pursuant and subject to the provisions of Section 2.17, payable in cash as provided in Section 2.11;

(iii) the Per Share Common Escrow Payment (if any) payable in cash at such time and only to the extent it becomes distributable to the Company Holders, pursuant and subject to the terms and conditions of the Escrow Agreement; and

(iv) the Per Share Common Reimbursement Payment (if any) at such time and only to the extent it becomes distributable to the Company Holders pursuant to the terms and conditions of Section 9.2.

(d) <u>Treasury Stock</u>; <u>Parent Stock</u>. All shares of capital stock of the Company held in the treasury of the Company or held by Parent immediately prior to the Effective Time shall be canceled and extinguished without any consideration therefor, and no payment shall be made with respect thereto.

(e) <u>Cancellation and Retirement of Capital Stock</u>. At the Effective Time, each Company Share converted into the right to receive cash in accordance with this Section 2.10 shall no longer be outstanding and shall automatically be canceled and retired and shall cease to exist, and each holder of a certificate that immediately prior to the Effective Time represented any such shares (a "<u>Company Certificate</u>") shall cease to have any rights with respect thereto, except the right to receive payment in accordance with Section 2.10.

(f) <u>Tax Withholding Obligation</u>. Each of Parent, the Paying Agent, the Escrow Agent and the Surviving Corporation reserves the right and is hereby authorized to make any withholdings required by applicable Tax Laws and to appropriately reduce any Merger Consideration payable to any Company Holder by the amount of any withholdings or payments that are required to be made by Parent, the Paying Agent, the Escrow Agent or the Surviving Corporation from or on behalf of such Company Holder (including, without limitation, withholding obligations of the Company arising from the termination and settlement of any Company Options pursuant to Section 2.13). To the extent that amounts are so withheld by Parent, the Paying Agent, the Escrow Agent or the Surviving Corporation, such amounts shall be treated for all purposes of this Agreement as having been paid to the Company Holder in respect of which such withholdings were made by Parent, the Paying Agent, the Escrow Agent or the Surviving Corporation, as the case may be.

2.11 Actions at Closing on the Closing Date. Parent shall take the following actions at the Closing:

(a) In connection with the Closing, Parent, the Representative and JPMorganChase Bank, N.A. or another Person mutually satisfactory to Parent and the Company (the "<u>Paying Agent</u>") shall have executed and delivered a paying agent agreement, in substantially the form attached hereto as **Exhibit D** (the "<u>Paying Agent Agreement</u>"), pursuant to which the Paying Agent shall distribute the Closing Payment and Working Capital Surplus (if any) to the Company Holders; <u>provided, however</u>, that any portion of the Merger Consideration that is to be paid to holders of Company Options in accordance with Section 2.13 shall not be deposited with the Paying Agent and shall instead be paid by the Surviving Corporation, out of funds made available therefor by Parent, to such holders through the next regularly scheduled payroll of the Surviving Corporation (or Parent, if the holders have been transitioned to Parent's payroll). At the Closing, Parent shall deposit with the Paying Agent the Closing Payment less the Closing Option/Warrant Amount and less the amount to be deposited with the Surviving Corporation as set forth in the immediately preceding sentence, by wire transfer of immediately available funds, for exchange in accordance with Section 2.12 (the "<u>Payment Fund</u>");

(b) Parent shall make a cash payment to the Representative, by wire transfer of immediately available funds to an account designated by the Representative prior to the Closing Date, in an amount equal to the Representative Reimbursement Set-Aside (as contemplated by Section 9.2);

(c) Parent shall deposit into the Escrow Fund (as contemplated by Section 2.16) with the Escrow Agent, by wire transfer of immediately available funds, an amount in cash equal to the Escrow Consideration, to be held in trust by the Escrow Agent pursuant to the Escrow Agreement; and

(d) Parent shall pay and discharge directly the aggregate amount of all unpaid (i) Company Indebtedness, including, without limitation, the Loan Amount, and (ii) Change of Control Payments (in each case to the extent that such amounts are being deducted from the Closing Payment and not otherwise being distributed pursuant to this Section 2.11), subject to receipt by Parent of Pay-Off Letters, releases and other documentation reasonably satisfactory to Parent reflecting such payment and discharge.

2.12 Exchange of Certificates and Company Warrants.

(a) As soon as reasonably practicable after the Effective Time (but in any event within three (3) Business Days after the Effective Time), Parent and the Surviving Corporation shall use commercially reasonable efforts to cause the Paying Agent to send to each Company Holder entitled to receive Merger Consideration in respect of their Company Shares or Company Warrants a letter of transmittal in the form of Exhibit E hereto (the "Letter of Transmittal") containing instructions for use in effecting the surrender of the Company Certificates and documentation for Company Warrants (the "Company Warrant Documents") in exchange for such Company Holder's portion of the Merger Consideration. Upon surrender of a Company Certificate or Company Warrant Document for cancellation (or an affidavit of lost Company Certificate or Company Warrant Document as contemplated by the Letter of Transmittal) to the Paying Agent, together with such letter of transmittal, duly executed and completed in accordance with the instructions thereto, and such other documents as may reasonably be required by the Paying Agent, the holder of such Company Certificate or Company Warrant Document shall be entitled to receive in exchange a check or wire transfer in the amount (after giving effect to any required tax withholding pursuant to Section 2.10(f)) of such Company Holder's portion of the Merger Consideration that the holder is entitled to receive under Sections 2.10 and 2.11, and the Company Certificate or Company Warrant Document so surrendered shall immediately be canceled. No interest will be paid or accrued with respect to any Merger Consideration deliverable upon due surrender of the Company Certificates or Company Warrant Document (or delivery of an affidavit of lost Company Certificate or Company Warrant Document as contemplated by the Letter of Transmittal). In the event of a transfer of ownership of Company Shares or Company Warrants that is not documented in the transfer records of the Company, payment may be made to a transferee if, and only if, the Company Certificate or Company Warrant Document representing such Company Shares or Company Warrants (or an affidavit of lost Company Certificate or Company Warrant Document as contemplated by the Letter of Transmittal) is presented to the Paying Agent, accompanied by all documents required to evidence and effect such transfer and by evidence that any applicable stock transfer Taxes have been paid. Until surrendered as contemplated by this Section 2.12, each Company Certificate or Company Warrant Document (other than Company Certificates representing Dissenting Shares) shall be deemed at any time after the Effective Time for all purposes to represent only the right to receive upon such surrender any payments which the holder thereof has the right to receive in respect of such Company Certificate or Company

Warrant Document pursuant to this Article II. In the case of Company Certificates representing Dissenting Shares, each Company Certificate representing Dissenting Shares shall be deemed at any time after the Effective Time for all purposes to represent only the right to receive the fair value of such Dissenting Shares pursuant to the DGCL.

(b) <u>No Further Ownership Rights; Transfer Books</u>. The payment of any consideration upon the surrender for exchange of Company Shares in accordance with the terms hereof shall be deemed to have been issued and made in full satisfaction of all rights pertaining to Company Shares as of the Effective Time, and (except with respect to Dissenting Shares) following the Effective Time, there shall be no further transfers on the stock transfer books of the Surviving Corporation of the Company Shares that were outstanding immediately prior to the Effective Time. If, after the Effective Time, Company Certificates are presented to the Surviving Corporation for any reason, they shall be canceled and exchanged as provided in this Section 2.12, subject to applicable Law in the case of the Company Certificates representing Dissenting Shares. From and after the Effective Time, holders of Company Certificates shall cease to have any rights as stockholders of the Company, except as provided by Law.

(c) <u>Lost, Stolen or Destroyed Certificates</u>. If any Company Certificates shall have been lost, stolen or destroyed, then payment shall be made in accordance with this Section 2.12 in exchange for such lost, stolen or destroyed the Company Certificates, upon the delivery to the Paying Agent of an affidavit of that fact by the Person claiming such Company Certificate to be lost, stolen or destroyed and an indemnity in a form reasonably satisfactory to Parent and the Paying Agent against any claim that may be made against the Paying Agent or Parent or otherwise with respect to such Company Certificate, including, if reasonably required by the Paying Agent, delivery of a bond.

(d) <u>No Liability</u>. Neither Parent, the Company, the Surviving Corporation nor the Paying Agent shall be liable to any former Company Holder for any stock or cash held by Parent, the Surviving Corporation or the Paying Agent for payment pursuant to this Section 2.12 properly delivered to a public official pursuant to any applicable abandoned property, escheat or similar Law.

(e) <u>Return of Closing Payment; Termination of Payment Fund</u>. Any portion of the Merger Consideration made available to the Paying Agent pursuant to the Paying Agent Agreement that remains undistributed to holders of the Company Certificates for one (1) year after the Effective Time shall be delivered by the Paying Agent to Parent, and any holders of the Company Certificates who have not theretofore complied with Section 2.11 for payments to holders of Company Certificates shall thereafter only look to Parent for payment of such Merger Consideration.

2.13 Options.

(a) At the Effective Time, each option to purchase shares of Common Stock that was issued under any Company Equity Incentive Plan or otherwise (each, a "<u>Company Option</u>") and is outstanding immediately prior to the Effective Time shall be cancelled. Each such Company Option that (x) is vested in accordance with the provisions of the Company Equity Incentive Plan and/or applicable option agreement

under which such Company Option was granted (whether by its terms or accelerated immediately prior to the Effective Time) and (y) has an exercise price per share that is less than the Per Share Common Closing Payment (each such Company Option, an "<u>In-the-Money Company Option</u>") shall be terminated in exchange for the right to receive, with respect to each share of Common Stock underlying such Company Option:

(i) an amount equal to (A) the Per Share Common Closing Payment minus (B) the exercise price applicable to such underlying share of Common Stock under such Company Option, payable in cash as provided in Section 2.11;

(ii) an amount equal to the Per Share Common Working Capital Payment (if any), at such time and only to the extent it becomes distributable to the Company Holders pursuant and subject to the provisions of Section 2.17, payable in cash as provided in Section 2.11;

(iii) an amount equal to the Per Share Common Escrow Payment (if any), at such time and only to the extent it becomes distributable to the Company Holders, pursuant and subject to the terms and conditions of the Escrow Agreement; and

(iv) an amount equal to the Per Share Common Reimbursement Payment (if any), at such time and only to the extent it becomes distributable to the Company Holders pursuant and subject to the terms and conditions of Section 9.2.

(b) At the Effective Time, each Company Option that (x) is vested in accordance with the provisions of the Company Equity Incentive Plan and/or applicable option agreement under which such Company Option was granted (whether by its terms or accelerated immediately prior to the Effective Time) and (y) has an exercise price per share that is equal to or greater than the Per Share Common Closing Payment (each such Company Option, an "<u>Out-of-the-Money Company Option</u>") shall be terminated in exchange for the right to receive, with respect to each share of Common Stock underlying such Option, the same amounts set forth in Section 2.13(a); <u>provided</u>, that:

(i) no payment shall be made unless such payment, when aggregated with all prior payments of Merger Consideration distributed to the Company Holders, would have caused such Out-of-the-Money Company Option to be an In-the-Money Company Option if such payment had been made at Closing;

(ii) the exercise price applicable to the underlying share of Common Stock under such Company Option shall be deducted from the first payment made under this Section 2.13(b); and

(iii) in the event that the holder of an Out-of-the-Money Company Option is entitled to receive any payment as contemplated by this Section 2.13(b), the Representative shall, prior to the payment of such amount, (x) update the Allocation Certificate to include the amounts payable to the holder of such Out-of-the-Money Company Option and to adjust the payments to be made to all other Company Holders, to the extent necessary, and (y) deliver joint written instructions with Parent to the Escrow Agent to update the schedule of escrow participants attached to the Escrow Agreement.

(c) Each holder of Company Options, as a condition to receiving any payment under Section 2.13, shall execute and deliver to the Company prior to the Effective Time, an option cancellation agreement in substantially the form attached hereto as **Exhibit F** (an "<u>Option Cancellation Agreement</u>"), to be effective upon the Effective Time.

(d) Each of Parent, the Escrow Agent and the Surviving Corporation reserves the right and is hereby authorized to make any withholdings required by applicable Tax Laws as set forth in Section 2.10(f) hereof.

(e) At the Effective Time, each outstanding Company Option that is not vested shall be terminated, without any consideration therefor.

(f) The Company agrees that the Company Board (or, if appropriate, any committee administering the Company Equity Incentive Plans) shall adopt such resolutions or take such other actions (including obtaining any required consents but not including the payment of any cash or non-cash consideration, without Parent's prior written consent) as may be required to effect the transactions described in this Section 2.13 as of the Effective Time, including, without limitation, providing any required notices to holders of all Company Options and obtaining Option Cancellation Agreements. The Company shall terminate any Company Options that are not vested and the Company Equity Incentive Plans, in each case effective immediately prior to the Effective Time.

2.14 Warrants.

(a) At the Effective Time, each warrant to purchase Common Stock or Company Preferred Stock (each, a "<u>Company Warrant</u>") that is outstanding immediately prior to the Effective Time shall be cancelled. Each such Company Warrant that (x) is vested in accordance with the terms of the applicable warrant agreement (whether by its terms or accelerated immediately prior to the Effective Time) and (y) has an exercise price per share that is less than either the Per Share Preferred Closing Payment for the applicable series of Company Preferred Stock, in the case of Company Warrants exercisable for shares of Company Preferred Stock, or the Per Share Common Closing Payment, in the case of Company Warrants exercisable for shares of Common Stock, in each case assuming such Company Warrants had been exercised immediately prior to the Effective Time (each such Company Warrant, an "<u>In-the-Money</u> <u>Company Warrant</u>"), shall be terminated in exchange for the right to receive, with respect to each share of Company Preferred Stock or Common Stock, as applicable, underlying such Company Warrant:

(i) (A) with respect to Company Warrants exercisable for Company Preferred Stock, an amount equal to (x) the number of shares of Company Preferred Stock subject to such Company Warrant that are vested immediately prior to the Effective Time, multiplied by (y) the amount by which the Per Share Preferred Closing Payment for the

applicable series of Company Preferred Stock exceeds the exercise price per share of the applicable series of Company Preferred Stock that is subject to such Company Warrant; and (B) with respect to Company Warrants exercisable for Common Stock, an amount equal to (x) the number of shares of Common Stock subject to such Company Warrant that are vested immediately prior to the Effective Time, multiplied by (y) the amount by which the Per Share Common Closing Payment exceeds the exercise price per share of Common Stock that is subject to such Company Warrant, in each case payable in cash as provided in Section 2.11;

(ii) (A) with respect to Company Warrants that are exercisable for Preferred Stock, an amount equal to the Per Share Preferred Working Capital Payment (if any); and (B) with respect to Company Warrants that are exercisable for Common Stock, an amount equal to the Per Share Common Working Capital Payment (if any); in each case at such time and only to the extent it becomes distributable to the Company Holders pursuant and subject to the provisions of Section 2.17, payable in cash as provided in Section 2.11;

(iii) (A) with respect to Company Warrants that are exercisable for Preferred Stock, an amount equal to the Per Share Preferred Escrow Payment (if any); and (B) with respect to Company Warrants that are exercisable for Common Stock, an amount equal to the Per Share Common Escrow Payment (if any); in each case at such time and only to the extent it becomes distributable to the Company Holders, pursuant and subject to the terms and conditions of the Escrow Agreement; and

(iv) (A) with respect to Company Warrants that are exercisable for Preferred Stock, an amount equal to the Per Share Preferred Reimbursement Payment (if any); and (B) with respect to Company Warrants that are exercisable for Common Stock, an amount equal to the Per Share Common Reimbursement Payment (if any); in each case at such time and only to the extent it becomes distributable to the Company Holders pursuant and subject to the terms and conditions of Section 9.2.

(b) At the Effective Time, each Company Warrant that (x) is vested in accordance with the terms of the applicable warrant agreement (whether by its terms or accelerated immediately prior to the Effective Time) and (y) has an exercise price per share that is equal to or greater than the Per Share Preferred Closing Payment for the applicable series of Company Preferred Stock, in the case of Company Warrants exercisable for shares of Company Preferred Stock, or the Per Share Common Closing Payment, in the case of Company Warrants exercisable for shares of Company Stock, in each case assuming such Company Warrants had been exercised immediately prior to the Effective Time (each such Company Warrant, an "<u>Out-of-the-Money</u> <u>Company Warrant</u>") shall be terminated in exchange for the right to receive, with respect to each share of Company Preferred Stock or Common Stock, as applicable, underlying such Warrant, the same amounts set forth in Section 2.14(a); provided, that:

(i) no payment shall be made unless such payment, when aggregated with all prior payments of Merger Consideration distributed to the Company Holders, would have caused such Out-of-the-Money Company Warrant to be an In-the-Money Company Warrant if such payment had been made at Closing;

(ii) the exercise price applicable to the underlying share of Common Stock or Preferred Stock, as applicable, under such Out-of-the-Money Company Warrant shall be deducted from the first payment made under this Section 2.14(b); and

(iii) in the event that the holder of an Out-of-the-Money Company Warrant is entitled to receive any payment as contemplated by this Section 2.14(b), the Representative shall, prior to the payment of such amount, (x) update the Allocation Certificate to include the amounts payable to the holder of such Out-of-the-Money Company Warrant and to adjust the payments to be made to all other Company Holders, to the extent necessary, and (y) deliver joint written instructions with Parent to the Escrow Agent to update the schedule of escrow participants attached to the Escrow Agreement.

(c) At the Effective Time, each outstanding Company Warrant that is not vested shall be terminated, without any consideration therefor.

(d) The Company agrees that the Company Board shall adopt such resolutions or take such other actions (including obtaining any required consents but not including the payment of any cash or non-cash consideration, without Parent's prior written consent) as may be required to effect the transactions described in this Section 2.14 as of the Effective Time, including, without limitation, providing any required notices to holders of all Company Warrants.

2.15 <u>Appraisal Rights</u>. Notwithstanding anything in this Agreement to the contrary, Company Shares that are issued and outstanding immediately prior to the Effective Time and are held by the holders of Company Shares who have not voted in favor of the Merger, consented thereto in writing or otherwise contractually waived their rights to appraisal and who have complied with all of the relevant provisions of Section 262 of the DGCL (the "<u>Dissenting Shares</u>") shall not be converted into or be exchangeable for the right to receive their applicable portion of the Merger Consideration, if any, unless and until the holders of such Dissenting Shares (the "<u>Dissenting Stockholders</u>") shall have failed to perfect or shall have effectively withdrawn or lost their rights to appraisal under the DGCL. The Company shall give Parent (a) prompt notice of any written demands for appraisal of any Company Shares, attempted withdrawals of such demands and any other instruments, documents or notices received by the Company relating to Dissenting Stockholders' rights of appraisal, and (b) the opportunity to direct all negotiations and proceedings with respect to demands for appraisal under the DGCL. Neither the Company nor the Surviving Corporation shall, except with the prior written consent of Parent, voluntarily make any payment with respect to, or settle or offer to settle, any such demand for payment. If any Dissenting Stockholder shall fail to perfect or shall have effectively withdrawn or lost the right to receive the Merger Consideration payable pursuant to Sections 2.10 and 2.11.

2.16 <u>Escrow</u>. In connection with the Closing, Parent, the Representative and JPMorganChase Bank, N.A. or another Person mutually satisfactory to Parent and the Company (the "<u>Escrow Agent</u>") shall have executed and delivered an escrow agreement, in substantially the form attached hereto as **Exhibit G** (the "<u>Escrow</u>

Agreement"), pursuant to which the Escrow Agent shall hold an escrow fund in an amount equal to the Escrow Consideration, together with interest (the "Escrow Fund"), until the Escrow Termination Date for the purposes of securing the payment of any Working Capital Deficit pursuant to Section 2.17 and any indemnification obligations of the Company Holders pursuant to Article X. Parent and the Representative shall provide a joint written instruction to the Escrow Agent of the date of the Escrow Termination Date under Section 2.2(b)(1) of the Escrow Agreement. The Parties will treat any escrow payment as taxable to a Company Holder only upon the release of the funds to such Company Holder and not earlier. Upon the termination of the Escrow Fund in accordance with the Escrow Agreement, the Escrow Consideration (less any Working Capital Deficit paid from the Escrow Fund pursuant to Section 2.17 and any indemnification obligations of the Company Holders paid from the Escrow Fund pursuant to Article X or continue to be held by the Escrow Agent pursuant to the terms of the Escrow Agreement) shall be distributed to the Company Holders in accordance with Sections 2.10, 2.13 and 2.14 and the Escrow Agreement, and Parent and the Representative shall provide a joint written instruction to the Escrow Agent under Section 2.2(e) of the Escrow Agreement. The adoption of this Agreement and the approval of the transactions contemplated hereby, including the Merger, by the stockholders of the Company shall constitute approval of the Escrow Agreement and all arrangements related thereto, including the depositing of the Escrow Consideration into the Escrow Fund. Any interest accruing with respect to the Escrow Fund shall accrue to the benefit of the Company Holders. The right of any Company Holder to receive its portion of the Escrow Fund, if any, (i) is an integral part of the consideration provided for in this Agreement, (ii) does not give the Company Holders dividend rights, voting rights, liquidation rights, preemptive rights or other rights of holders of capital stock of the Company, (iii) shall not be evidenced by a certificate or other instrument, (iv) shall not be assignable or otherwise transferable by such Company Holder, except by will, upon death or by operation of Law, and (v) does not represent any right other than the right to receive the consideration set forth in this Section 2.16. Any attempted transfer of the right to any portion of the Escrow Consideration by any holder thereof (other than as specifically permitted by the immediately preceding sentence) shall be null and void.

2.17 Working Capital Adjustment.

(a) <u>Calculation of Closing Working Capital Estimate</u>. Not later than five (5) Business Days prior to the Closing Date, the Company shall prepare and deliver, in addition to the Allocation Certificate required by Section 6.8, to Parent a projected consolidated balance sheet of the Company and its Subsidiaries as of 11:59 pm Pacific Time as of the day immediately prior to the Closing Date (the "<u>Preliminary Closing Balance Sheet</u>"). The Preliminary Closing Balance Sheet shall be prepared in accordance with GAAP (subject to normal and recurring year-end adjustments) and on a basis consistent with and using the same principles, practices and policies as those used in the preparation of and substantially in the form of the Company Financial Statements. The Preliminary Closing Balance Sheet shall be accompanied by a statement (the "<u>Preliminary Closing Statement</u>"), certified by the Chief Financial Officer of the Company as being accurate, containing all items required to be included pursuant to this Agreement and prepared in accordance with this Agreement, setting forth in reasonable detail the calculations showing the basis for the determination of the amounts set

forth in the Preliminary Closing Balance Sheet and the Company's good faith best estimate of (i) the Closing Date Working Capital calculated on the basis of the Preliminary Closing Balance Sheet, including each respective component thereof (the "<u>Closing Working Capital Estimate</u>"), and (ii) without duplication of amounts taken into consideration in calculating the Closing Working Capital Estimate: (1) the amount of cash and cash equivalents as of such date, (2) the Closing Option/Warrant Amount, including each respective component thereof, (3) the Company Indebtedness, including each respective component thereof, (4) the Change of Control Payments, including each respective component thereof, (5) the aggregate cost of Meso Supplies, (6) the amount of unpaid Transaction Expenses for services performed on or prior to December 31, 2013, including each respective component thereof, (7) the amount of unpaid Transaction Expenses for services performed after December 31, 2013, including each respective component thereof, (8) the amount of unpaid IPO Expenses, including each respective component thereof, (9) the amount of unpaid Fiscal Year 2013 Bonus Payments, including each respective component thereof, and (10) the amount of unpaid expenses payable to Parent or its counsel under the Loan Documents. All legal expenses of the Company shall be treated as Transaction Expenses or IPO Expenses except for such expenses as are set forth in reasonable detail on a schedule provided to Parent with the Preliminary Closing Statement. Along with the Preliminary Closing Balance Sheet and the Preliminary Closing Statement, the Company shall deliver to the Parent backup materials and schedules reasonably requested by Parent.

(b) <u>Adjustment to Closing Payment</u>. If the Closing Working Capital Estimate is greater than the Working Capital Target, then the difference (the "<u>Estimated Working Capital Surplus</u>") shall be included as an addition to the calculation of the Closing Payment, and if the Closing Working Capital Estimate is less than the Working Capital Target, then the difference (the "Estimated Working Capital Deficit") shall be included as an addition in the calculation of the Closing Payment Reduction Amount; <u>provided</u>, <u>however</u>, in no event shall any adjustment be made unless the difference is greater than the amount determined by multiplying the Working Capital Target by seven and one half percent (7.5%).

(c) The Parent's Right to Require a Post-Closing Adjustment. Parent shall be entitled, within sixty (60) days after the Closing, to prepare and deliver to the Representative a consolidated balance sheet of the Company and its Subsidiaries as of 11:59 pm Pacific Time as of the day immediately prior to the Closing Date (the "Parent Closing Balance Sheet"). The Parent Closing Balance Sheet shall be prepared in accordance with GAAP (subject to normal and recurring year-end adjustments) and on a basis consistent with and using the same principles, practices and policies as those used in the preparation of and substantially in the form of the Company Financial Statements. The Parent Closing Balance Sheet shall be accompanied by a statement (the "Parent Working Capital Statement"), certified by the Chief Financial Officer of Parent, setting forth Parent's calculation of the Closing Date Working Capital, including each respective component thereof (the "Parent Working Capital Determination").

(d) <u>Parent's Failure to Deliver the Parent Closing Balance Sheet</u>. If Parent does not deliver the Parent Closing Balance Sheet and Parent Working Capital Statement in accordance with Section 2.17(c) within sixty (60) days after the Closing, or if during such 60-day period Parent delivers to the Representative written notice that Parent accepts the Estimated Working Capital Surplus or Estimated Working Capital Deficit, as applicable, then such amounts shall be considered final, conclusive and binding.

(e) Objection by Representative. If Parent delivers a Parent Closing Balance Sheet and Parent Working Capital Statement in accordance with Section 2.17(c) within sixty (60) days after the Closing, (i) Parent and the Representative shall reasonably cooperate with and assist each other in resolving any items disputed by the Representative in good faith, including by making available and granting reasonable access (during normal business hours and subject to Parent's reasonable security measures and insurance requirements) to the Representative and its agents, including an independent certified public accountant designated by the Representative and reasonably acceptable to Parent, whose fees and disbursements shall be paid by the Company Holders (initially out of the Representative Reimbursement Set-Aside), to records and employees of the Surviving Corporation (provided that the Representative and his agents is bound by a duty of a confidentiality reasonably satisfactory to Parent and the Representative or his agents is not, in Parent's reasonable judgment, intrusive or disruptive), and (ii) the Representative shall have a period of thirty (30) days after delivery of the Parent Closing Balance Sheet and Parent Working Capital Statement (the "Working Capital Objection Period"), to deliver to Parent a statement (the "Working Capital Objection Statement") setting forth any objections that the Representative may have to the Parent Working Capital Determination, including a reasonably detailed explanation of the basis for each such objection along with reasonably detailed supporting calculations. If the Representative does not deliver to Parent a Working Capital Objection Statement by the end of the Working Capital Objection Period, or if during the Working Capital Objection Period the Representative delivers to Parent written notice that the Representative accepts the Parent Closing Balance Sheet, the Parent Working Capital Statement, the Parent Working Capital Determination, then the Parent Closing Balance Sheet, the Parent Working Capital Statement, the Parent Working Capital Determination shall be considered final, conclusive and binding. If the Representative does deliver a Working Capital Objection Statement by the end of the Working Capital Objection Period, the Representative and Parent shall attempt in good faith to resolve any disputed items. If the Representative and Parent are unable to resolve all or any of the disputed items within forty five (45) days after delivery of the Working Capital Objection Statement, then the remaining disputed items shall be submitted to the Neutral Auditor. The Neutral Auditor shall be instructed that it shall, based solely on its determination of the disputed items, prepare and deliver to the Representative and Parent the Neutral Auditor's determination of the Closing Date Working Capital, as soon as reasonably practicable (and in any event within sixty (60) days after its engagement), and such Neutral Auditor's determination of the Closing Date Working Capital shall be final, conclusive and binding. Parent and the Representative (out of the Representative Reimbursement Set-Aside) shall equally share the fees and disbursements of the Neutral Auditor. Each of Parent and the Representative shall cooperate with and assist the Neutral Auditor to resolve disputed items, including by making available and granting reasonable access to records and employees. The Closing Date Working Capital as finally determined by (A) failure of Parent to deliver a Parent Closing Balance Sheet and Parent Working Capital Statement, or (B) failure of the Representative to deliver a Working Capital Objection Statement, or (C) agreement

of the Representative and Parent (including pursuant to a notice by the Representative that the Parent Working Capital Statement is acceptable), or (D) the Neutral Auditor, shall be the "<u>Final Working Capital</u>". The date of the determination of the Final Working Capital shall be referred to herein as the "<u>Working Capital Determination Date</u>."

(f) <u>Final Closing Payment Adjustment</u>. If the difference of (A) the Final Working Capital and (B) the Closing Working Capital Estimate is not zero, then (1) to the extent such difference is negative (the absolute value of such difference, the "<u>Working Capital Deficit</u>"), the Representative and Parent shall immediately deliver joint written instructions to the Escrow Agent to pay and deliver the Working Capital Deficit to Parent out of the Escrow Fund, and (2) to the extent that such difference is positive (such difference, the "<u>Working Capital Surplus</u>"), Parent shall, within five (5) Business Days after the Working Capital Determination Date, pay and deliver to the Paying Agent, and shall (i) cause the Paying Agent to distribute to the applicable Company Holders pursuant to the Paying Agent Agreement an amount equal to such Working Capital Surplus to be distributed to the Company Holders as contemplated by Sections 2.10, 2.12 and 2.13; <u>provided</u>, <u>however</u>, that payments to holders of Company Options shall be paid by the Surviving Corporation in accordance with Section 2.11.

ARTICLE III REPRESENTATIONS AND WARRANTIES OF THE COMPANY

The Company makes the following representations and warranties to Parent and Acquisition Sub as of the Agreement Date and as of the Closing Date (except, in each case, to the extent such representations and warranties speak expressly as of an earlier date) as follows; <u>provided</u>, <u>however</u>, that (i) for purposes of determining the truth and correctness of the representations and warranties of the Company as of the Agreement Date, the representations and warranties shall be deemed qualified only by such exceptions as are disclosed in schedule of exceptions set forth in Schedule III to this Agreement (the "<u>Company Disclosure Schedule</u>" or "<u>Company Disclosure Schedules</u>"), and (ii) each exception disclosed in any section, subsection or clause of the Company Disclosure Schedules shall be deemed disclosed with respect to the specific section, subsection or clause of the Agreement to which it is referenced and such other representations and warranties of whether such representation or warranty contains an express reference to the Company Disclosure Schedules) to the extent that it is reasonably apparent from a reading of such disclosure item that it would also apply to such other representation or warranty:

3.1 <u>Organization, Good Standing, Qualification and Power</u>. The Company (a) is a corporation duly organized, validly existing and in good standing under the Laws of the State of Delaware, (b) has all requisite corporate power and authority to own, lease and operate its properties and to carry on its business as it is now being conducted, and (c) is duly qualified or licensed and in corporate and tax good standing to do business in each jurisdiction listed in Section 3.1 of the Company Disclosure Schedule, which constitute all of the jurisdictions in which the character of the properties owned, leased or operated by it or the nature of its activities makes such qualification necessary, except such jurisdictions where the failure to be so qualified or licensed or in good standing would not reasonably be expected to result, individually or in the aggregate, in a Company Material Adverse Change.

3.2 Subsidiaries.

(a) Except as set forth in Section 3.2(a) of the Company Disclosure Schedule, the Company does not own or control, and has never owned or controlled, directly or indirectly, any interest in any other corporation, partnership, limited liability company, association or other business entity. Each Subsidiary is duly organized, validly existing and in good standing (to the extent applicable) under the Laws of its jurisdiction of formation. Each Subsidiary has all requisite corporate power and authority to own, lease and operate its properties and to carry on its business as currently conducted. Each Subsidiary is duly qualified or licensed to do business and is in good standing (to the extent applicable) as a foreign organization in each jurisdiction listed in Section 3.2(a) of the Company Disclosure Schedule, which constitute all of the jurisdictions in which the conduct of its business or the ownership, leasing, holding or use of its properties makes such qualification necessary, except such jurisdictions where the failure to be so qualified or licensed or in good standing would not reasonably be expected to result, individually or in the aggregate, in a Company Material Adverse Change. The Company has made available to Parent an accurate and complete copy of each Subsidiary's Organizational Documents, each as amended or restated as of the date of the Company Disclosure Schedule. None of the Company's Subsidiaries has violated its Organizational Documents in any material respect. Section 3.2(a) of the Company Disclosure Schedule lists, with respect to each Subsidiary, every jurisdiction in which such Subsidiary has facilities, maintains an office or has a current employee, consultant or contractor. None of the Company's Subsidiaries or their respective predecessors have conducted any business under or otherwise used for any purpose in any jurisdiction any fictitious name, assumed name, "d/b/a", trade name or other name.

(b) The authorized capitalization of each Subsidiary, including the identity of each holder of any outstanding equity interest therein, is set forth in Section 3.2(b) of the Company Disclosure Schedule. All of the outstanding capital stock of, or other equity or ownership interests in, each Subsidiary is owned by the Company, directly or indirectly, free and clear of any Lien and free of any other limitation or restriction (including any restriction on the right to vote, sell or otherwise dispose of such capital stock or other equity or ownership interests (other than any restrictions imposed by Law)). There are no outstanding (i) shares of capital stock of the Company, or any other outstanding voting securities or other equity or ownership interests of the Company, or securities of any of the Company's Subsidiaries convertible into or exercisable or exchangeable for shares of capital stock or other voting securities or equity or ownership interests in any Subsidiary ("<u>Subsidiary Securities</u>") or (ii) any option, warrant, subscription right, preemptive right, other right, proxy, put, call, demand, plan, commitment, agreement, understanding or arrangement of any kind relating to such Subsidiary Securities, whether issued or unissued, or any other security convertible into or exchangeable for any such security. There are no outstanding obligations of the Company or any of its Subsidiaries to repurchase, redeem or otherwise acquire any outstanding Subsidiary Securities. All of the outstanding Subsidiary Securities have been duly authorized and are validly issued, fully paid and non-assessable.

3.3 <u>Organizational Documents</u>. The Company has delivered as of the Agreement Date to Parent a complete and correct copy of each of its Certificate of Incorporation and by-laws, each as amended or restated as of the date of the Company Disclosure Schedule (collectively, "<u>Organizational Documents</u>"). Except as set forth in Section 3.3 of the Company Disclosure Schedule, the Organizational Documents have not been amended or modified and no provision thereof has been waived since the Agreement Date. Each of the Organizational Documents is in full force and effect on the Agreement Date and the Closing Date, as applicable. The Company has not violated its Organizational Documents in any material respect. The minute books of the Company made available to Parent are the only minute books of the Company and contain accurate summaries, in all material respects, of all material meetings of directors (or committees thereof) and stockholders or actions by written consent since the time of incorporation of the Company. Section 3.3 of the Company Disclosure Schedule lists as of the Agreement Date every state or foreign jurisdiction in which the Company has facilities, maintains an office or has a current employee, consultant or contractor. Neither the Company nor its predecessors has conducted any business under or otherwise used for any purpose in any jurisdiction any fictitious name, assumed name, "d/b/a", trade name or other name.

3.4 Capitalization.

(a) The capitalization of the Company, including the total number of shares of each class or series of capital stock that the Company has authority to issue and the number of issued and outstanding shares of Common Stock and of each class and series of Company Preferred Stock as of the Agreement Date is set forth in Section 3.4 of the Company Disclosure Schedule.

(b) No Company Shares are held in the treasury of the Company. Section 3.4(b) of the Company Disclosure Schedule sets forth, as of the Agreement Date, a complete and accurate list of the number of shares of Common Stock and Company Preferred Stock that are reserved for future issuance upon exercise of Company Warrants. Section 3.4(b) of the Company Disclosure Schedule sets forth, as of the Agreement Date, a complete and accurate list of the number of shares under the Company Equity Incentive Plans, including the number of shares subject to Company Options that are currently outstanding and that remain available for grant. Section 3.4(b) of the Company Disclosure Schedule sets forth, as of the Agreement Date, a complete and accurate list of all holders of Company Shares, together with the number of shares by class held by each such holder on an issued and as-converted basis.

(c) Section 3.4(c) of the Company Disclosure Schedule sets forth, as of the Agreement Date, a complete and accurate list of (i) the name of each holder of outstanding Company Options, indicating, with respect to each Company Option, the number of shares of Common Stock for which such Company Option is exercisable, the exercise price, the date of grant or issuance, the vesting schedule for such Company Option and the number of shares vested and unvested under such Options; and (ii) all holders of outstanding Company Warrants, indicating, with respect to each Company Warrant, the number of shares and series of capital stock for which such Company Warrant is exercisable (including as they may have been increased under the terms of such Warrant), the exercise price, the date of issuance and the

expiration date thereof. The Company has provided to Parent complete and accurate copies of all Contracts evidencing Company Warrants and all Contracts evidencing Company Options (or a form thereof to the extent based on a form). No Company Option is exercisable for any class or series of capital stock of the Company other than Common Stock.

(d) Except as described in Section 3.4(b) or Section 3.4(c) hereof or in Section 3.4(d) of the Company Disclosure Schedule, there are no other outstanding securities, options (whether vested or unvested), warrants, calls, rights, commitments or agreements to which the Company is a party or by which it is bound obligating the Company to issue, deliver or sell, or cause to be issued, delivered or sold, additional shares of capital stock or other securities of the Company. Except as described in Section 3.4(d) of the Company Disclosure Schedule, there are no outstanding contractual obligations of the Company to repurchase, redeem or otherwise acquire any shares of capital stock (or options or warrants to acquire any such shares) or other securities or equity interests of the Company. All outstanding shares of capital stock of the Company are, and all shares which may be issued upon the exercise of Company Options and Company Warrants will be, when issued in accordance with their terms, duly authorized, validly issued, fully paid and nonassessable and not subject to any preemptive or similar rights. To the Company's Knowledge, all of the issued and outstanding shares of capital stock of the Company have been offered, issued and sold by the Company in compliance with all applicable federal and state securities Laws.

(e) Except as set forth in Section 3.4(e) of the Company Disclosure Schedule and the Voting Agreement, there are no voting trusts, proxies or other agreements with respect to the Company Shares to which the Company is a party or, to the Knowledge of the Company, any of the Company Holders, is a party or by which any of them is bound with respect to the issuance, holding, acquisition, voting, registration, sale, transfer (including agreements relating to rights of first refusal, "co-sale" rights or "drag-along" rights) or other disposition of any Company Shares. The execution, delivery and performance of this Agreement and the consummation of the transactions contemplated hereby by the Company does not implicate any rights or obligations under any of the voting trusts, proxies or other agreements set forth in Section 3.4(e) of the Company Disclosure Schedule that have not been complied with or waived. The Company Holders with respect to any of the foregoing have been or will be properly given or shall have been properly waived any required notice prior to the Merger.

3.5 Authorization; Binding Obligation.

(a) The Company has all requisite corporate power and authority to enter into and deliver this Agreement, the Certificate of Merger, and each of the other agreements, certificates or documents required to be executed in accordance herewith (collectively, the "<u>Related Agreements</u>") to which it is or will be a party, to perform its obligations hereunder and thereunder and to consummate the transactions contemplated hereby and thereby. Subject to Stockholder Approval and the filing and recordation of the Certificate of Merger in accordance with the DGCL, the execution and delivery by the Company of this Agreement, the performance of its obligations hereunder, and the consummation by the Company of the Merger and the other transactions contemplated hereby have been duly and validly authorized by all necessary corporate action on the part of the Company Board (<u>"Company Board Approval</u>") and, other than the Stockholder Approval, no other corporate proceedings on the part of the Company are necessary to authorize this Agreement or to consummate the Merger or any other transactions contemplated hereby. As of the Closing Date, the Company Will have delivered to Parent certified copies of the Company Board Approval and the Stockholder Approval and as of such date neither the Company Board Approval nor the Stockholder Approval will have been revoked, rescinded or amended.

(b) This Agreement has been, and each of the Related Agreements to which the Company is a party will be at the Closing, duly and validly executed and delivered by the Company, and, assuming that this Agreement and the Related Agreements to which the Company is a party constitute the valid and binding agreement of the other parties hereto and thereto (other than the Company), this Agreement constitutes, and in the case of the Related Agreements they will at Closing constitute, legal, valid and binding obligations of the Company enforceable against the Company in accordance with their respective terms, subject to applicable bankruptcy, insolvency, reorganization, moratorium, fraudulent conveyance and similar Laws affecting the enforcement of creditors' rights and remedies generally, and subject, as to enforceability, to general principles of equity, including principles of commercial reasonableness, good faith and fair dealing (regardless of whether enforcement is sought in a proceeding at Law or in equity); provided, however, that the Certificate of Merger will not be effective until accepted by the Secretary of State of the State of Delaware.

3.6 Consents.

(a) The execution and delivery by the Company of this Agreement and any Related Agreement required by this Agreement to be executed and delivered by the Company do not, and the performance of this Agreement and any Related Agreement required by this Agreement to be executed and delivered by the Company shall not, require the Company or any of its Subsidiaries to obtain any Consent or approval of any Person or observe any waiting period imposed by, or make any filing with or notification to, any Governmental Authority, except for (a) the Stockholder Approval, (b) the filing of the Certificate of Merger in accordance with the DGCL, (c) the filing of a pre-merger notification and report form under the HSR Act and the filings and receipt, termination or expiration, as applicable, of such other approvals or waiting periods required under any other applicable competition, merger control, antitrust or similar Law, (d) such Consents and approvals set forth in Section 3.6(a) of the Company Disclosure Schedules (other than any Consents or approvals under Contracts that are not Material Contracts, for which no representation or warranty is made under this Section 3.6(a) as of the Agreement Date and which need not be listed therein).

(b) Section 3.6(b) of the Company Disclosure Schedule sets forth all notices to, and all Consents, waivers and approvals of, parties to any Material Contract to which the Company or any of its Subsidiaries is a party or by which they or their properties are bound that are required thereunder in connection with the transactions contemplated hereby, or for any such Material Contract to remain in full force and effect without limitation, modification or alteration (including payment of any additional amounts or consideration other than ongoing fees, royalties or payments which the Company or any of its Subsidiaries, as the case may be, would otherwise be required to pay pursuant to the terms of such Material Contract had the transactions contemplated hereby not occurred) after the Effective Time, so as to preserve all rights of, and benefits to, the Company and its Subsidiaries, as the case may be, under such Material Contract from and after the Effective Time.

(c) The Company owns or possesses all right, title, and interest in and has been issued or granted all of the Company Permits. Section 3.6(c) of the Company Disclosure Schedule sets forth a true and complete list of all Company Permits, each of which is valid and in full force and effect, and the Company is in compliance in all material respects with the terms and conditions of all Company Permits and, during the past three (3) years, has not received any written notice that the Company is in material violation of any of the terms and conditions of any Company Permit.

3.7 No Violation. Except as set forth in Section 3.7 of the Company Disclosure Schedule, the execution and delivery by the Company of this Agreement and any Related Agreement required by this Agreement to be executed and delivered by the Company do not, and the performance of this Agreement and any Related Agreement required by this Agreement to be executed and delivered by the Company, will not, (a) violate or contravene any provision of the Company's Organizational Documents, (b) violate or contravene any Law or Judgment, in each case, applicable to the Company or any of its Subsidiaries, or by which its properties is bound or affected, or (c) conflict with or result in a breach of or constitute a default (or an event that with notice or lapse of time or both would become a default) by, or impair the Company's or any of its Subsidiaries' rights or alter the rights or obligations of any third party under, or give to others any rights of termination, amendment, acceleration or cancellation of, or result in the imposition or creation of a Lien on any of the properties or assets (tangible or intangible) of the Company or any of its Subsidiaries pursuant to, any note, bond, mortgage, indenture, Contract, Company Permit, Payment Program or other material instrument or obligation to which the Company or any of its Subsidiaries is a party or by which the Company or any of its Subsidiaries or their respective properties are bound or affected, except in the case of clauses (b) and (c) for such violations, conflicts, contraventions, breaches, defaults or impairments that would not reasonably be expected to result, individually or in the aggregate, in a Company Material Adverse Change.

3.8 <u>Required Vote</u>. The Company Board has, at a meeting duly called and held or by written consent, (a) unanimously approved and declared this Agreement and the Related Agreements advisable, (b) determined that the transactions contemplated hereby and thereby are advisable, fair to and in the best interests of the Voting Holders, (c) resolved to recommend

adoption of this Agreement, the Related Agreements, the Merger and the other transactions contemplated hereby and thereby to the Voting Holders, and (d) directed that this Agreement be submitted to the Voting Holders for their approval and authorization. The affirmative vote of (i) at least 77% of all outstanding shares of Company Preferred Stock, voting together as a single class on an as-converted to Common Stock basis, and (ii) at least a majority of all outstanding shares of Common Stock and Company Preferred Stock, voting together as a single class on an as-converted to Common Stock basis (collectively, the "<u>Corporate Stockholder Approval</u>") are the only votes of the holders of any class or series of capital stock of the Company necessary to approve and authorize the Company's execution and delivery of this Agreement, the Merger and the other transactions contemplated hereby and thereby; provided, that the affirmative vote of at least a majority of all outstanding shares of Common Stock may also be required by the California Corporations Code, if applicable (the "<u>California Stockholder Approval</u>" and together with the Corporate Stockholder Approval, the "<u>Stockholder Approval</u>").

3.9 Financial Statements; No Undisclosed Liabilities.

(a) [Reserved].

(b) Section 3.9(b) of the Company Disclosure Schedules sets forth the (i) audited consolidated balance sheet (the "Audited Balance Sheet") of the Company and its Subsidiaries as of December 31, 2012 (the "Audited Balance Sheet Date"), and the related audited consolidated statements of income, cash flow and stockholders' equity for the twelve-month period then ended (such financial statements, collectively, the "Audited Financial Statements"), certified by the Company's independent public accountants and accompanied by a copy of such auditor's report; and (ii) unaudited consolidated balance sheet (the "Interim Balance Sheet") of the Company and its Subsidiaries as of December 31, 2013 (such date, the "Interim Balance Sheet Date") and the related unaudited consolidated statements of income, cash flows and stockholders' equity for the twelve-month period ended December 31, 2013 (such financial statements, the "Interim Financial Statements"). The Audited Financial Statements and Interim Financial Statements fairly present the financial condition of the Company and its Subsidiaries as of the dates indicated and the results of operations of the Company and its Subsidiaries for the respective periods indicated, and have been prepared in accordance with GAAP, with the exception that the Interim Financial Statements do not include footnotes and remain subject to changes resulting from normal year-end audit adjustments which adjustments are not expected to be material. As of the Agreement Date, the Company does not have any Liabilities, whether or not required to be disclosed by GAAP, which are individually or in the aggregate material to the business, (ii) Liabilities incurred since the date of the Interim Balance Sheet in the ordinary course of business, (iii) Liabilities that are not required to be presented in unaudited interim financial statements of the Company prepared in conformity with GAAP, or (iv) as set forth in Section 3.9(b) of the Company Disclosure Schedules.

(c) The Company has in place systems and processes that in the Company's reasonable judgment provide reasonable assurances that (i) transactions are executed in accordance with management's general or specific authorizations, (ii) transactions are recorded as necessary to permit preparation of financial statements in conformity with GAAP, (iii) access to assets is permitted only in accordance with management's general or specific authorizations, and (iv) the recorded accountability for assets is compared with the existing assets at reasonable intervals and appropriate action is taken with respect to any difference. To the Knowledge of the Company, there have been no instances of fraud by the Company or any of its Subsidiaries, whether or not material, that occurred during any period covered by the Company Financial Statements.

(d) To the Knowledge of the Company, no employee has provided or threatened to provide information to any Governmental Authority regarding the commission of any crime by the Company or any of its Subsidiaries or the violation of any Law applicable to the Company, any of its Subsidiaries or any part of their respective operations.

3.10 <u>Absence of Certain Events</u>. Except as set forth in Section 3.10 of the Company Disclosure Schedule, since the Audited Balance Sheet Date through the Agreement Date, the Company has conducted its business in the ordinary and usual course and in a manner consistent with past practices. Except as set forth in Section 3.10 of the Company Disclosure Schedule or as specifically contemplated by this Agreement, since the Balance Sheet Date through the Agreement Date, there has not been, occurred or arisen:

(a) any sale, assignment, lease, license or other disposition, of any material asset or property of the Company or any of its Subsidiaries, or creation of any Lien in such assets or properties, except in the ordinary course of business and consistent with past practices;

(b) destruction of, damage to or loss of any material assets of the Company or any of its Subsidiaries (whether or not covered by insurance), other than ordinary wear and tear of such assets;

(c) any entry into or any termination, modification or amendment of any Material Contract to which the Company or any of its Subsidiaries was or is a party, except for new Contracts, terminations, modifications or amendments made in the ordinary course of business consistent with past practice;

(d) any individual transaction (excluding regularly scheduled payments of payroll) by the Company or any of its Subsidiaries in an amount in excess of \$500,000;

(e) any increase in the salary or other compensation payable or to become payable by the Company or any of its Subsidiaries to any employee, consultant, or contractor of the Company or any of its Subsidiaries (including any repricing of any right to acquire Company Shares or any amendment to or the acceleration of any vesting terms related to any award with respect to Company Shares held by such individuals); any entry into an employment agreement or offer letter (or material amendment to an existing employment agreement or offer letter); any

payment or agreement to pay any bonus, severance or termination benefits, or special remuneration to any employee; any making or grant of any increase in benefits under any Employee Plan; or any adoption of any change of control agreement or severance plan or new Employee Plan (or material amendment to any such existing agreement or plan); in each of the foregoing cases other than (i) annual salary increases which do not exceed \$20,000 for any such individual, (ii) in the ordinary course of business consistent with prior practice, or (iii) as necessary to comply with any applicable minimum wage Law;

(f) any issuance by the Company or any of its Subsidiaries of any shares of capital stock or any security, right, option or warrant convertible into or exercisable or exchangeable for any shares of capital stock other than (i) Company Options issued under the Company Equity Incentive Plans, (ii) shares issued upon exercise of any Company Options issued under the Company Equity Incentive Plans, and (iii) shares issued upon exercise of any Company Warrants outstanding as of the Agreement Date;

(g) any adjustment, split, combination or reclassification of any of the Company's or any of its Subsidiaries' shares of capital stock or other equity interests, or declaration, setting aside or payment of any dividend or other distribution in respect of the capital stock;

(h) any amendment or change to the Organizational Documents of the Company or any of its Subsidiaries;

(i) any Indebtedness incurred by the Company or any of its Subsidiaries, or any loans made or agreed to be made by or to the Company or any of its Subsidiaries, guarantee by the Company or any of its Subsidiaries of any Indebtedness, issuance or sale of any debt securities of the Company or any of its Subsidiaries or purchase of or guaranteeing of any debt securities of others, except for advances of customary travel and expense advances to executive officers or directors of the Company or any of its Subsidiaries in the ordinary course of business and consistent with past practice;

(j) any work stoppage, labor strike or other labor trouble, or any action, suit, claim, demand, labor dispute or grievance relating to any labor, employment and/or workplace safety and health matter involving the Company or any of its Subsidiaries, including, but not limited to, any claim or charge of wrongful discharge, discrimination or harassment, wage and hour violations, or other unlawful labor and/or employment practices or actions;

(k) any change in accounting methods or practices by the Company or any of its Subsidiaries;

(l) any commencement or, to the Company's Knowledge, threat of commencement of any lawsuit or proceeding against or, to the Company's Knowledge, investigation of the Company or any of its Subsidiaries, or commencement or threat of any commencement of any litigation by the Company or any of its Subsidiaries, or settlement of any lawsuit, proceeding or investigation (regardless of the party initiating the same);

(m) terminations and/or layoffs of employees, consultants or contractors who receive more than \$150,000 per year;

(n) waiver or release of any right or claim that is material to the Company or any of its Subsidiaries, including any write-off or other compromise of any account receivable of the Company or any of its Subsidiaries, except in the ordinary course of business consistent with past practice;

(o) (i) license, transfer or sale by the Company or any of its Subsidiaries of any rights to any Company Intellectual Property or the entering into of any license agreement (other than non-exclusive end-user license agreements entered into by the Company or any of its Subsidiaries in the ordinary course of business consistent with past practice that do not include any rights with respect to source code), distribution agreement, reseller agreement, security agreement, assignment or other conveyance or option for the foregoing with respect to any Company Intellectual Property with any Person, (ii) purchase, license or acquisition of any Intellectual Property or the entering into of any license agreement, distribution agreement, reseller agreement, security agreement, assignment or other conveyance or option for the foregoing with respect to any Intellectual Property of any Person (other than generally available, non-customized software, in each case with no recurring license fee greater than \$12,000 per year), (ii) material change in pricing, milestones or royalties set or charged by the Company or any of its Subsidiaries to its customers or licensees or in pricing, milestones or royalties set or charged by Persons who have licensed Intellectual Property to the Company or any of its Subsidiaries (other than generally available, non-customized software, in each case with no recurring into, or amendment of, any agreement with respect to the development of any Intellectual Property with any Person;

(p) any agreement, understanding, authorization or proposal for the Company or any of its Subsidiaries to take any of the actions specified in this Section 3.10 (other than negotiations with Parent and its representatives regarding the transactions contemplated hereby); or

(q) any change, circumstance, development, state of facts, event or effect that has resulted in or would reasonably be expected to result in a Company Material Adverse Change.

3.11 Legal Proceedings. Except as set forth in Section 3.11 of the Company Disclosure Schedule, there is no Action that is pending or, to the Company's Knowledge, has been threatened in writing by or against the Company or any of its Subsidiaries, any of their respective properties or assets, or to the Knowledge of the Company, any Company Personnel with respect to their service to the Company or any Subsidiary or that would reasonably be expected to interfere with their ability to perform their duties, nor, to the Company's Knowledge, is there any reasonable basis therefor or any facts, threats, claims or allegations that would reasonably be expected to result in any such threatened or pending Action. Neither the Company nor any of its Subsidiaries has received any claim, complaint, incident, report, threat or notice of any Action or any facts, threats, claims or allegations that would reasonably be expected to result in any such threatened or pending Action. Neither the Company nor any of its Subsidiaries has received any claim, complaint, incident, report, threat or notice of any Action or any facts, threats, claims or allegations that would reasonably be expected to result in any such threat or notice of any Action or any facts, threats, claims or allegations that would reasonably be expected to result in any such the Company or any of its Subsidiaries or action pending by the Company or any of its Subsidiaries, or which the Company or any of its Subsidiaries intend to initiate, against any other Person. Section 3.11 of the Company Disclosure Schedule lists each Action that has been commenced since July 31, 2008, by or against the Company or any of its Subsidiaries as of the Agreement Date. None of the Company, its Subsidiaries or their respective properties is subject to any Judgment that materially impairs the Company's or any of its Subsidiaries' ability to operate.

3.12 Compliance with Laws.

(a) Except as set forth in Section 3.12 of the Company Disclosure Schedule, each of the Company and its Subsidiaries has been and is in compliance in all material respects with, has not violated in any material respects and is not in violation in any material respect of, and has not received any written notice of non-compliance or violation or alleged non-compliance or violation with respect to, all applicable Laws, including all applicable Health Care Laws and Judgments of any Governmental Authority. Without limiting the foregoing, neither the Company nor any of its Subsidiaries, nor any current or former director, officer or employee of the Company or any of its Subsidiaries (with respect to their service to the Company or any Subsidiary) has, during the past five (5) years, received or been subject to any Action by any Governmental Authority or other Person alleging any violations of Laws, and no Action alleging any violation of any Laws by the Company or any of its Subsidiaries is, to the Knowledge of the Company, threatened in writing against the Company or any of its Subsidiaries have not, during the preceding five (5) years, conducted any internal investigation, inquiry, or review in connection with which the Company or any of its Subsidiaries retained outside legal counsel for the purpose of conducting or assisting with such investigation, inquiry, or review with respect to any actual, potential, or alleged violation of any Law by the Company. The Company and its Subsidiaries have not received any citation, directive, letter, or other written or oral communication that any Governmental Authority has at any time during the last five (5) years challenged or questioned the legal right of the Company and its Subsidiaries to market, offer, or sell its services in the present manner or style thereof.

(b) Neither the Company nor any of its Subsidiaries has submitted, or caused to be submitted, any claim to any Payment Program in connection with any referrals that violated, in any material respect, any applicable self-referral Law, including the Federal Ethics in Patient Referrals Act, 42 U.S.C. § 1395nn (known as the "<u>Stark Law</u>"), or any applicable state self-referral Law.

(c) Neither the Company nor any of its Subsidiaries has Knowingly or willfully solicited, received, paid, or offered to pay any remuneration, directly or indirectly, overtly or covertly, in cash or kind for the purpose of making or receiving any referral that violated, in any material respect, any applicable anti-kickback Law, including the Federal Health Care Program Anti-Kickback Statute, 42 U.S.C. § 1320a-7b(b), or any applicable state anti-kickback Law.

(d) Neither the Company nor any of its Subsidiaries has Knowingly submitted or Knowingly caused to be submitted any claim for payment to any Payment Program in violation, in any material respect, of any Laws relating to false claims or fraud, including the Federal False Claims Act, 31 U.S.C. § 3729, or any applicable state false claim or fraud Law.

(e) Neither the Company nor any of its Subsidiaries has Knowingly committed any act, or failed to commit any act, where such act or failure to act would subject the Company or such Subsidiary to exclusion from any federal or state health care program pursuant to 42 U.S.C. §1320a-7.

(f) Neither the Company nor any of its Subsidiaries has Knowingly submitted, or Knowingly caused to be submitted, any claim or request for payment or transferred any remuneration to any individual eligible for benefits from any federal or state Payment Program in a manner that would reasonably be expected to subject the Company or such Subsidiary to civil money penalties under 42 U.S.C. §1320a-7a.

(g) Each employee or agent of the Company and any of its Subsidiaries required to be licensed by an applicable Governmental Authority, professional body, and/or medical body has such licenses, such licenses are in full force and effect, and, to the Company's Knowledge, there are no facts or circumstances that would reasonably be expected to result in any such licenses being suspended, revoked, or otherwise to lapse.

(h) Neither the Company, nor any of its Subsidiaries, nor, to the Company's Knowledge, any of their respective current or former directors, officers, or employees has been excluded, suspended, debarred, or otherwise sanctioned by any Governmental Authority, including, without limitation, the Office of Inspector General for the Department of Health and Human Services or the General Services Administration, in a manner that would adversely affect the Company or any of its Subsidiaries.

(i) To the Company's Knowledge, neither the Company nor any of its Subsidiaries has arranged or contracted, by employment or otherwise, with an individual or entity that has been excluded, suspended, debarred, or otherwise sanctioned by any Governmental Authority, including, without limitation, the Office of Inspector General for the Department of Health and Human Services or the General Services Administration.

(j) No notice of violation, citation, suspension, revocation, limitation, warning, audit finding, request, or other written or oral communication issued by a Governmental Authority has been received by the Company, any of its Subsidiaries, or, to the Company's Knowledge, any of their respective current or former directors, officers, or employees in a manner that would adversely affect the Company or any of its Subsidiaries that has not been resolved to the applicable Governmental Authority's satisfaction, without the imposition of any fines or penalties.

3.13 Compliance with Privacy and Data Security Laws.

(a) <u>Health Insurance Portability and Accountability Act of 1996</u>. Each of the Company and its Subsidiaries: (i) has assessed the applicability of HIPAA, and its implementing regulations to the Company and such Subsidiary, including any fully insured and self-insured health plans that the Company or such Subsidiary sponsor or have sponsored or contribute to or have contributed to and in view of any health care provider activities, if any, in which the Company or such Subsidiary engage; (ii) is a "covered entity" under HIPAA; and (iii) has entered into the business associate, data use, or other agreements with HIPPA-covered entities under which agreements it has agreed to safeguard or otherwise restrict the use or disclosure of any protected health information of such a HIPAA-covered entity.

(b) <u>Privacy and Security Policies and Collection of Personal Information</u>. Each of the Company and its Subsidiaries has at all times complied in all material respects and is currently in compliance in all material respects with any applicable privacy and security policies it has established. The current privacy and security policies of the Company and its Subsidiaries are in all material respects in compliance with all applicable Laws and Regulations. No Personal Information has been (i) collected by the Company or its Subsidiaries in material violation of any Privacy Laws, or (ii) transferred or disclosed by the Company or its Subsidiaries to third parties in material violation of any Privacy Laws. There are no written notices or Actions pending, or, to the Company or its Subsidiaries, by any Governmental Authority, or private parties, involving Personal Information held or stored by the Company or its Subsidiaries.

(c) <u>Data Breaches</u>. Neither the Company nor any of its Subsidiaries has received any written notice regarding any violation in any material respect of any Privacy Laws, and, to the Company's Knowledge, neither the Company nor any of its Subsidiaries has had any data breach involving Personal Information or, if it was made aware of a data breach, has complied with all data breach notification and related obligations and has taken corrective action reasonably designed to prevent recurrence of such a data breach.

(d) <u>Compliance</u>. The Company and each of its Subsidiaries are currently and have been at all times in compliance in all material respects with all Privacy Laws. No Person has filed a HIPAA-related Action with the Company or any of its Subsidiaries or, to the Company's Knowledge, any Governmental Authority relating to the Company or any of its Subsidiaries.

(e) <u>No Actions</u>. No Action has been filed or commenced against the Company or its Subsidiaries, nor to the Company's Knowledge, threatened against the Company or its Subsidiaries, relating to Privacy Laws; nor, to the Company's Knowledge, has the Company or its Subsidiaries incurred any Liabilities under any Privacy Laws.

3.14 Title to Properties.

(a) Neither the Company nor any of its Subsidiaries owns any real property. Section 3.14(a) of the Company Disclosure Schedule identifies all lease agreements pursuant to which any real property is leased to the Company or any of its Subsidiaries (the "Leased Property"). The Company has delivered to Parent complete

and accurate copies of all such leases, master leases, ground leases, subleases, and any amendments, subordination, non-disturbance and attornment agreements, operating agreements and any other documents relating thereto. With respect to each Leased Property, except as set forth in Section 3.14(a) of the Company Disclosure Schedule, neither the Company nor any of its Subsidiaries is and, to the Knowledge of the Company, no other party to the lease relating to such Leased Property is, in breach or violation of, or in default under, such lease, (ii) no event, occurrence, condition or act has occurred, is pending or, to the Knowledge of the Company, is threatened in writing, which, with the giving of notice, lapse of time, or the happening of any further event, occurrence, condition or act, would constitute a breach or default by the Company or any of its Subsidiaries or, to the Knowledge of the Company, any other party to such lease, under such lease, or give rise to a right of termination, cancellation or to loss of a material benefit under, or to increased, additional, accelerated or guaranteed rights or entitlements of any Person under any such leases, (iii) there are no disputes, oral agreements or forbearance programs in effect as to the lease relating to such Leased Property, (iv) no third party consents are required from landlords, master landlords, ground lessors, lenders or otherwise as a result of any transfers or assignments that occur or are deemed to occur by operation of the Merger for any of the Leased Property, and (v) the uses of the Leased Property are permitted under and are not in violation of applicable zoning or land use Laws.

(b) The Company or one of its Subsidiaries is the true and lawful owner or lessee of and has good and valid title to, or a valid leasehold interest in, all personal property (tangible or intangible) that it purports to own, including, without limitation, all those reflected on the Audited Balance Sheet or thereafter acquired, except that which has been sold or otherwise disposed of for fair value in the ordinary course of business since the date of the Audited Balance Sheet and not in violation of this Agreement, in each case free and clear of all Liens (other than Permitted Liens). All such properties and assets that are material to the Company are in good condition and repair, reasonable wear-and-tear excepted, and are, and as of the Closing Date will be, adequate and sufficient to carry on the business of the Company and its Subsidiaries and suitable for their present use.

(c) Section 3.14(c) of the Company Disclosure Schedule sets forth all Liens securing Indebtedness of the Company or any of its Subsidiaries.

3.15 Employee and Benefit Matters.

(a) The Company has made available to Parent a true and complete list of the names, positions and rates of compensation of all current officers, directors, employees and consultants of the Company and its Subsidiaries, showing each such person's name, positions, and annualized remuneration and bonuses and has made available to Parent a list of all fringe benefits generally made available to such officers, directors and employees, in each case, for the current fiscal year and the most recently completed fiscal year. Except as indicated in Section 3.15(a) of the Company Disclosure Schedule, (i) all employees are employed on an "at-will" basis and their employment can be terminated at any time for any reason without any amounts being owed to such individual other than with respect to wages, compensation and benefits accrued before the termination, (ii) the Company's and its Subsidiaries relationships with all individuals who act on their own as contractors, consultants or other service providers to the Company and its Subsidiaries can be terminated at any time for any reason without any amounts

being owed to such individual other than with respect to compensation or payments accrued before the termination, and (iii) no employee is on disability or other leave of absence, other than standard leaves taken in the ordinary course of the Company's and its Subsidiaries' operations. Except as set forth in Section 3.15(a) of the Company Disclosure Schedule, neither the Company nor any of its Subsidiaries has been notified by any employee that such employee intends to terminate such employee's employment with the Company or its Subsidiaries, including in connection with or as a result, in part or in whole, of the transactions contemplated hereby or any other sale of the Company. The Company and each of its Subsidiaries has complied in all material respects with all applicable Laws and agreements respecting employment, employment practices, employee benefits, terms and conditions of employment, immigration matters, labor matters, and wages and hours, in each case, with respect to its employees and to the Knowledge of the Company, there are no allegations to the contrary. The Company and each of its Subsidiaries has complied in all material respects with all applicable Laws governing the employment of non- U.S. nationals in the United States, including the Immigration and Nationality Act 8 U.S.C. Sections 1101 et seq. and its implementing Regulations. Except as set forth in Section 3.15(a) of the Company Disclosure Schedule, neither the Company nor any of its Subsidiaries has sponsored any employee for, or otherwise engaged any current employee working pursuant to, a non-immigrant visa.

(b) To the Knowledge of the Company: (i) no current employee, consultant or contractor is a party to or is bound by any employment agreement, patent disclosure agreement, non-competition agreement, any other restrictive covenant or other agreement with any Person, or subject to any judgment, decree or order of any court or administrative agency, any of which would reasonably be expected to affect (A) the performance by such Person of any of his or her duties or responsibilities for the Company or its Subsidiaries, or (B) the Company's or its Subsidiaries' business or operations; (ii) no current employee, consultant or contractor of the Company or its Subsidiaries is in violation of any term of any employment agreement, patent disclosure agreement, non-competition agreement, or any other restrictive covenant to a former employer or entity relating to the right of any such employee, contractor or consultant to be employed or retained by the Company or its Subsidiaries; and (iii) the Company and each of its Subsidiaries is not and has not ever been engaged in any dispute or litigation with any employee, consultant or contractor regarding intellectual property matters.

(c) The Company and each of its Subsidiaries is not engaged or has not ever been engaged in any unfair labor practice of any nature, that, if adversely determined, would result in liability in any material respect to the Company or its Subsidiaries. There has never been any slowdown, work stoppage, labor dispute or union organizing activity, or any similar activity or dispute, affecting the Company or its Subsidiaries or any of their respective employees. There is not now pending and, to the Knowledge of the Company, no Person has threatened to commence, any such slowdown, work stoppage, labor dispute, union organizing activity or any similar activity or dispute.

(d) All employees of the Company and its Subsidiaries have been, and currently are, properly classified under the Fair Labor Standards Act of 1938, as amended, and under any similar Law of any state or other jurisdiction applicable to such employees. Any Persons now or heretofore engaged by the Company or its Subsidiaries as consultants or contract labor or independent contractors have been properly classified as such, are not entitled to any

compensation or benefits to which regular, full-time employees are or were at the relevant time entitled, were and have been engaged in accordance with all applicable Laws, and have been treated accordingly and as required for all Tax purposes. The Company and each of its Subsidiaries is not delinquent to, and has not failed to pay, any of its employees, consultants or contractors for any wages (including overtime, meal breaks or waiting time penalties), salaries, commissions, accrued and unused vacation to which they would be entitled under applicable Law, if any, bonuses, benefits or other compensation for any services performed by them or amounts required to be reimbursed to such individuals. The Company and each of its Subsidiaries is not liable for any payment to any trust or other fund or to any Governmental Authority, with respect to unemployment compensation benefits, social security or other benefits or obligations for employees (other than routine payments to be made in the normal course of business and consistent with past practice). Neither the Company nor any of its Subsidiaries has material liability arising out of the treatment of any service provider as a consultant or independent contractor and not as an employee.

(e) There are no demands or claims pending or, to the Knowledge of the Company, threatened, before any Governmental Authority by any employees, consultants or contractors for compensation, pending severance benefits, vacation time, unpaid meal or rest breaks, vacation pay or pension benefits, or, to the Knowledge of the Company, any other claim threatened or pending before any Governmental Authority (or any state "referral agency") from any employee, consultants or contractors or any other Person arising out of the Company's or its Subsidiaries' status as employer or joint employer, whether in the form of claims for employment discrimination, harassment, retaliation, unfair labor practices, grievances, wrongful discharge, wage and hour violations, workplace health and safety violations, breach of contract, unfair business practice, tort, unfair competition or otherwise. In addition, there are no pending or, to the Knowledge of the Company, threatened claims or actions against the Company or its Subsidiaries under any workers compensation policy or long-term disability policy, nor to the Knowledge of the Company, is there any reasonable basis therefor. The Company and each of its Subsidiaries has complied with and is in compliance with, in each case in all material respects, all applicable workers compensation Laws.

(f) The Company and each of its Subsidiaries has not implemented any plant or office closing, transfer of employees or layoff of employees that (without regard to any actions taken by the Parent after the Closing) is or could reasonably be expected to be in violation of the Worker Adjustment and Retraining Notification Act or similar Laws.

(g) Except for Material Contracts or agreements set forth in Section 3.16(a) of the Company Disclosure Schedule, neither the Company nor any of its Subsidiaries is a party to any (i) outstanding Contracts with employees, agents, consultants, advisers, salesmen, sales representatives, distributors, sales agents or dealers (other than offer letters and letter agreements entered into in the ordinary course of business consistent with past practice with any such Persons who are terminable "at will" without Liability to the Company or any of its Subsidiaries and which letters and agreements do not contain post-termination severance provisions), or (ii) collective bargaining agreements or Contracts with any labor union or other representative of employees and is not liable with respect to any employee benefits provided for by any such agreement. Moreover, the Company and each of its Subsidiaries does not know of any activities or proceedings of any labor union to organize any employees. The Company and each of its

Subsidiaries has made available to Parent copies of all such Contracts and agreements and such copies are true and correct. No strike, union organizational activity or formal charge or complaint of employment discrimination is currently pending or, to the Knowledge of the Company, threatened against the Company or any of its Subsidiaries.

(h) Except as set forth in Section 3.15(h) of the Company Disclosure Schedule, neither the Company, its Subsidiaries nor any Plan Affiliate has maintained, sponsored, adopted, made contributions to or obligated itself to make contributions to or to pay any benefits or grant rights under or with respect to or has any liability with respect to any "Employee Pension Benefit Plan" (as defined in Section 3(2) of ERISA), "Employee Welfare Benefit Plan" (as defined in Section 3(1) of ERISA), "multi-employer plan" (as defined in Section 3(37) of ERISA), plan of deferred compensation, medical plan, life insurance plan, long-term disability plan, dental plan or other plan providing for the welfare of any of the Company's, any Subsidiary's or any Plan Affiliate's employees or former employees or beneficiaries thereof, personnel policy (including but not limited to vacation time, holiday pay, bonus programs, moving expense reimbursement programs and sick leave), excess benefit plan, bonus or incentive plan (including but not limited to stock options, restricted stock, stock bonus, or other equity based plans or profit sharing, savings and deferred bonus plans), salary reduction agreement, change-of-control agreement, employment agreement, consulting agreement or Law, which could give rise to or result in the Company, any of its Subsidiaries or such Plan Affiliate having any material debt, liability, claim or obligation of any kind or nature, whether accrued, absolute, contingent, direct or indirect (each, an "Employee Plan"). Each Employee Plan has been maintained, funded and administered in compliance in all material respects with its terms and in compliance in all material respects with all applicable Laws.

(i) The Company and each of its Subsidiaries is in compliance in all material respects with all applicable Laws relating to wages, hours, labor (including work conditions, safety regulations and employee representatives), plant closings and layoffs, and collective bargaining agreements.

(j) The Company has made available to the Parent (i) copies of all plan documents, including any amendments, for each Employee Plan; (ii) copies of all summary plan descriptions, if applicable, for each Employee Plan; (iii) the three most recent Form 5500-series annual reports for each Employee Plan, with all required schedules and audited financial statements; (iv) the most recent favorable determination letter or opinion for each Employee Plan that is intended to be a "qualified plan" under Section 401(a) of the Code; (v) copies of each trust, contract, insurance policy or other funding document with respect to any Employee Plan.

(k) No Employee Plan constitutes, and neither the Company, any of its Subsidiaries nor any Plan Affiliate is contributing to or is required to contribute to a multi-employer plan, a multiple employer plan as defined under Section 413(c) of the Code or an Employee Pension Benefit Plan that is subject to Title IV of ERISA. Neither the Company, any of its Subsidiaries nor any Plan Affiliate has any liability, contingent or otherwise, under the Code or Title IV of ERISA, including any "withdrawal liability" as defined under Section 4201 et seq. of ERISA, whether to the Pension Benefit Guaranty Corporation, the Internal Revenue Service, or any other Person.

(l) Each Employee Plan which is intended to be "qualified" within the meaning of Section 401(a) of the Code has received a favorable determination letter or opinion letter from the Internal Revenue Service, and, to the Knowledge of the Company, no event has occurred and no condition exists which would reasonably be expected to result in the revocation of any such determination.

(m) No Employee Plan is self-funded or provides, and there is no verbal or written agreement, promise or representation that obligates the Company, any of its Subsidiaries or any Plan Affiliate to provide, welfare, death, health care, medical or similar benefits, beyond termination of employment, service or retirement other than (i) coverage mandated by Law or (ii) death or retirement benefits under any Employee Plan that is intended to be qualified under Section 401(a) of the Code. Each Employee Plan that is a "group health plan" (within the meaning of Section 5000(b)(1) of the Code) has been operated in compliance in all material respects with all laws applicable to such plan, its terms, and with the group health plan continuation coverage requirements of the Consolidated Omnibus Budget Reconciliation Act (COBRA), Section 4980D of the Code and Sections 701 through 707 of ERISA, Title XXII of the Public Health Service Act and the provisions of the Social Security Act and the Patient Protection and Affordable Care Act.

(n) Except as set forth in Section 3.15(n) of the Company Disclosure Schedule, the execution and performance of this Agreement or of any ancillary agreements, either alone or upon the occurrence of any additional or subsequent event, will not directly or indirectly (i) result in any payment (whether of severance pay or otherwise) becoming due from the Company or any of its Subsidiaries to any current or former officer, employee, director or consultant (or dependents of such Persons) of the Company or any of its Subsidiaries, (ii) accelerate the time of payment or vesting or increase the amount of compensation due from the Company or any of its Subsidiaries to any current or former officer, employee, director or consultant (or dependents of such Persons) of the Company or any of its Subsidiaries to any current or former officer, employee, director or consultant (or dependents of such Persons) of the Company or any of its Subsidiaries to any current or former officer, employee, director or consultant (or dependents of such Persons) of the Company or any of its Subsidiaries to any current or former officer, employee, director or consultant (or dependents of such Persons) of the Company or any of its Subsidiaries, or (iii) constitute a triggering event that will or may result in any "parachute payment" (as such term is defined in Section 280G of the Code).

(o) All contributions or premium payments required to be made prior to the Effective Time with respect to any Employee Plan have been or will be timely made and all contributions or premium payments not required to be made prior to the Effective Time with respect to any Employee Plan have been made or properly accrued in the Financial Statements.

(p) The Company and each of its Subsidiaries has reserved all rights necessary to amend or terminate each of the Employee Plans without the consent of any other Person, and there are no restrictions (including any verbal or written representations to the contrary) on the ability of the Company, any of its Subsidiaries or any Plan Affiliate to amend, terminate or transfer any Employee Plan.

(q) There have been no "prohibited transactions" (as defined under Section 4975 of the Code and Section 406 of ERISA) with respect to any Employee Plan. No "fiduciary" as defined in Section 3(21) of the Code has any liability for breach of fiduciary duty with respect to the administration of or investment of the assets of any Employee Plan. There are no claims or proceedings pending, or, to the Knowledge of the Company, threatened or reasonably anticipated (other than routine claims for benefits), against any Employee Plan or against the Company or any Plan Affiliate. There are no audits, inquiries or proceedings pending or, to the Knowledge of the Company, threatened by the IRS, DOL or any other Governmental Authority, with respect to any Employee Plan.

3.16 Contracts.

(a) Section 3.16 of the Company Disclosure Schedule sets forth, as of the Agreement Date, a correct and complete list of each effective Contract to which the Company or any of its Subsidiaries is a party and which constitutes:

(i) an employment Contract (other than offer letters and letter agreements entered into in the ordinary course of business consistent with past practice with any such individuals who are terminable "at will" without Liability to the Company or any of its Subsidiaries and which letters and agreements do not contain post-termination severance provisions), a consulting Contract providing annual compensation in excess of \$100,000 and which is not terminable on less than ninety (90) days written notice without Liability to the Company or any of its Subsidiaries, or any employee collective bargaining agreement or other contract with any labor union;

(ii) a Contract (other than trade debt incurred in the ordinary course of business) under which the Company or any of its Subsidiaries has borrowed any money from, or issued any note, bond, debenture or other evidence of Indebtedness to, any Person;

(iii) a non-competition, non-solicitation or exclusive dealing arrangement or any other agreement or obligation which purports to limit or restrict in any respect (A) the ability of the Company or any of its Subsidiaries to solicit customers or employees or (B) the manner in which, or the localities in which, all or any portion of the business and operations of the Company or its Subsidiaries or, following consummation of the transactions contemplated by this Agreement, the business and operations of Parent and its Affiliates, is conducted;

(iv) a Contract that (A) involves future expenditures or projected receipts by the Company or any of its Subsidiaries of more than \$250,000 in any one-year period or is otherwise material to the operation of the business of the Company or its Subsidiaries and (B) is terminable by the other party or parties upon a change in control of the Company;

(v) a Contract granting a Lien (other than Permitted Liens) upon any material property or asset of the Company or any of its Subsidiaries;

(vi) a lease, sublease or similar Contract with any Person under which the Company or any of its Subsidiaries is a lessor or sublessor of, or makes available for use to any Person (other than the Company or such Subsidiary), (A) any Leased Property or (B) any portion of any premises otherwise occupied by the Company or any of its Subsidiaries;

(vii) a lease or similar Contract with any Person under which (A) the Company or any of its Subsidiaries is lessee of, or holds or uses, any machinery, equipment, vehicle or other tangible personal property owned by any Person (other than any Contracts that individually do not involve the payment by or to the Company or any of its Subsidiaries of more than \$100,000 in any twelve-month period and in the aggregate do not involve the payment by or to the Company or any of its Subsidiaries of more than \$300,000 in any twelve-month period) or (B) the Company or any of its Subsidiaries is a lessor or sublessor of, or makes available for use by any Person, any tangible personal property owned or leased by the Company or any of its Subsidiaries;

(viii) a Contract which is a joint venture or partnership agreement, or research or development collaboration or similar arrangement;

(ix) a Contract providing for the acquisition or disposition after the Agreement Date of any of the Company's or any of its Subsidiaries'

material assets;

(x) any agreement that contains a "most favored nation" clause or other term providing preferential pricing to a third party;

(xi) a Contract granting a third party any license to any Company Intellectual Property, or pursuant to which the Company or any of its Subsidiaries have been granted by a third party any license to any Intellectual Property, or any other license, option or other Contract relating in whole or in part to the Company Intellectual Property or the Intellectual Property of any other Person (each, a "<u>License Agreement</u>" and collectively, the "<u>License</u> <u>Agreements</u>"), other than (A) agreements between the Company and its employees in the Company's standard form thereof and (B) any generally available, non-customized, third party software licensed to the Company which does not require aggregate payments in any given year in excess of \$40,000 in license, maintenance, royalty and/or other fees;

(xii) a Contract with a health maintenance organization or health benefit plan, health insurance plan, or other third party reimbursement or payment program administered or otherwise operated by any Person other than a Governmental Authority (excluding any employee benefit plans which the Company has put in place for the benefit of its employees); or

(xiii) any other Contract that (A) involves future expenditures or projected receipts by the Company or any of its Subsidiaries of more than \$250,000 in any one-year period, (B) is not terminable on less than 180 days' notice without Liability to the Company or any Subsidiary, or (C) is otherwise material to the operation of the business of the Company or any of its Subsidiaries (the Contracts set forth in the foregoing clauses (i) through (xiii) are collectively referred to herein as the "Material Contracts").

(b) Assuming the due authorization, execution and delivery by the other parties thereto, each Material Contract is a valid and binding obligation of the Company or its Subsidiary and, to the Knowledge of the Company, of each of the other parties thereto. True and complete copies of each Material Contract have been delivered to Parent. Each Material Contract is in full force and effect, and none of the Company, any of its Subsidiaries nor, to the

Knowledge of the Company, any other party thereto, is in default or breach in any material respect under the terms of any such Material Contract. No Material Contract requires prepayments, additional payments or increased payments by the Company or any of its Subsidiaries as a result of consummation of the transactions contemplated by this Agreement. Except as set forth in Section 3.16(b) of the Company Disclosure Schedule, no notice, waiver, consent or approval is required (or the lack of which would give rise to a right of termination, cancellation or acceleration of, or entitle any party to accelerate, whether after the giving of notice or lapse of time or both, any obligation under the Material Contracts) under or relating to any Material Contract in connection with the execution, delivery and performance of this Agreement or the consummation of the Merger or any of the other transactions contemplated hereby, and, immediately following the Effective Time, each Material Contract will continue to be in full force and effect, and valid, binding and enforceable in accordance with its terms.

3.17 Officer and Director Indemnification Obligations. Section 3.17 of the Company Disclosure Schedule sets forth a full and complete list of all indemnification obligations of the Company or any of its Subsidiaries to any of their respective officers, directors or employees.

3.18 Intellectual Property.

(a) Section 3.18(a) of the Company Disclosure Schedule lists and separately identifies, as of the Agreement Date, all Company Registered Intellectual Property (excluding any Company Intellectual Property that is only non-exclusively licensed to the Company and its Subsidiaries) setting forth, for each item, the full legal name of the owner of record, applicable jurisdiction, status, application or registration number, and date of application, registration or issuance, as applicable, and including the following additional information: (i) for each item of Company Registered Intellectual Property that is licensed in from another Person, identification of the applicable License Agreement and each of the parties thereto, (ii) for each pending or registered trademark, trade name or service mark, the class of goods covered, (iii) for each URL or domain name, any renewal date and the name of the registry, (iv) for each registered mask work, the date of first commercial exploitation), and (v) for each pending or registered copyright registration, the author(s) of the subject work of authorship.

(b) Except as set forth in Section 3.18(b) of the Company Disclosure Schedule, the Company and each of its Subsidiaries, and to the Knowledge of the Company, each of its licensors, has complied with all the requirements of all United States and foreign patent offices and all other applicable Governmental Authorities to maintain the Company Patents in full force and effect, including payment of all required fees when due to such offices or agencies. Other than prior art references cited in the applicable patent office file history with respect to a particular Company Patent (a complete copy of such file histories which the Company has previously delivered to Parent), to the Knowledge of the Company, there are no prior art references or prior public uses, sales, offers for sale or disclosures or other facts or circumstances that could invalidate any of the Company Patents or any claim thereof, or of any conduct the result of which could render any Company Patent or any claim thereof invalid or unenforceable.

(c) Except as disclosed in Section 3.18(c) of the Company Disclosure Schedule, to the Company's Knowledge, the original, first and joint inventors of the subject matter claimed in the patents and patent applications included in the Company Registered Intellectual Property (the "<u>Company</u> <u>Patents</u>") are properly named in the Company Patents. To the Knowledge of the Company, the applicable statutes governing marking of products covered by the Company Patents have been and are being fully complied with; in particular, to the Knowledge of the Company, all Company Products have been properly marked and no Company Products have been improperly marked.

(d) Except as disclosed in Section 3.18(d) of the Company Disclosure Schedule, each item of Company Intellectual Property is either: (i) owned exclusively by the Company or one of its Subsidiaries free and clear of any Liens, or (ii) rightfully used and authorized for use by the Company and its Subsidiaries and their permitted successors pursuant to a valid and enforceable written license identified on Section 3.16(a)(xi) of the Company Disclosure Schedule. Except as disclosed in Section 3.18(d) of the Company Disclosure Schedule, the Company and its Subsidiaries have all rights in the Company Intellectual Property necessary thereunder to carry out the Company's and its Subsidiaries' current and planned activities with respect to the Commercial Products and the Development Products. Except as disclosed in Section 3.18(d) of the exclusive right and ability to develop, manufacture, sell and commercialize the Commercial Products and the Development Products as developed, planned, manufactured, sold and commercialized by the Company and its Subsidiaries as of the Agreement Date and as of the Closing Date.

(e) The Company and each of its Subsidiaries, and to the Knowledge of the Company, each of their respective licensors and licensees are in compliance with and have not committed any uncured breach, violation or default under, or received written notice that they have breached, violated or defaulted under, any of the terms or conditions of any license, sublicense or other agreement to which the Company or any of its Subsidiaries is a party or is otherwise bound relating to any of the Company Intellectual Property, nor to the Company's Knowledge has there been any event or occurrence that would reasonably be expected to constitute such a breach, violation or default (with or without the lapse of time, giving of notice or both). Each such agreement in full force and effect, and to the Company's Knowledge, no Person obligated to the Company or any of its Subsidiaries pursuant to any such agreement in default thereunder. Except as set forth on Section 3.18(e) of the Company Disclosure Schedule, immediately following the Closing Date, the Surviving Corporation will be permitted to continue to exercise all of the Company's and its Subsidiaries' rights under such contracts, licenses and agreement not occurred and without the payment of any additional amounts or consideration other than the amount of fees, royalties or payments which the Company or any of its Subsidiaries is obligated to provide any consideration (whether financial or otherwise) to any third Person, nor is any third Person otherwise entitled to any consideration, with respect to any exercise of rights by the Company or any of its Subsidiaries or the Surviving Corporation, as successor to the Company or any of its Subsidiaries, in the Company Intellectual Property (other than (A) sales commissions paid to employees pursuant to Company's standard commission plan and (B) license fees).

(f) To the Knowledge of the Company, neither the Company's and its Subsidiaries' business, nor the manufacture, use, or sale of any Commercial Product infringes or misappropriates, or will infringe or misappropriate, any other Person's Intellectual Property. Neither the Company nor any of its Subsidiaries is making any unauthorized use of any confidential information or trade secrets of any third party, including without limitation, any customer of the Company or any of its Subsidiaries, or any past or present employee of the Company or any of its Subsidiaries. The Company and its Subsidiaries have the right to use, without infringing the rights of others, all customer lists, designs, manufacturing or other processes, computer software, systems, data compilations, research results and other information required for or incident to its products or its business as presently conducted. Except as set forth on Section 3.18(f) of the Company Disclosure Schedule, no claims (i) challenging the validity, enforceability, effectiveness or ownership by the Company, any of its Subsidiaries or any of their respective licensors of any of the Company Intellectual Property or (ii) to the effect that the use, reproduction, modification, manufacture, distribution, licensing, sublicensing, sale, or any other exercise of rights in or under any Company Intellectual Property by the Company and its Subsidiaries or by any licensee of the Company or any of its Subsidiaries, or manufacture, use or sale of any Company Product or engagement in any activity, infringes or will infringe on any intellectual property or other proprietary or personal right of any Person have been asserted against the Company or any of its Subsidiaries or, to the Company's Knowledge, are threatened by any Person nor, to the Company's Knowledge, does there exist any basis for such a claim. There are no legal or governmental proceedings, including interference, re-examination, reissue, opposition, nullity, or cancellation proceedings pending that relate to any of the Company Registered Intellectual Property that is owned or exclusively licensed to the Company or any of its Subsidiaries, nor to the Company's Knowledge are there any such proceedings that relate to any of the Company Registered Intellectual Property that is non-exclusively licensed to the Company or any of its Subsidiaries, other than review of pending patent and trademark applications, and to the Company's Knowledge no such proceedings are threatened or contemplated by any Governmental Entity or any other Person. To the Company's Knowledge, all Company Registered Intellectual Property is valid and subsisting. To the Company's Knowledge, there is no unauthorized use, infringement, or misappropriation of any Company Intellectual Property by any third party or employee, consultant or contractor.

(g) Except as set forth on Section 3.18(g) of the Company Disclosure Schedule, each Person who is or was an employee or independent contractor of any Company or Subsidiary and who is or was involved in the creation or development of any Company Intellectual Property owned by such Company or Subsidiary has entered into a written agreement with such Company or Subsidiary assigning to the Company or Subsidiary all rights in any work performed by such Person and agreeing to maintain the confidentiality of all trade secrets, each of which agreements is in full force and effect.

(h) Except as set forth on Section 3.18(h) of the Company Disclosure Schedule, the Merger and other transactions contemplated by this Agreement will not alter, impair or otherwise affect any rights of the Company or its Subsidiaries in any Company Intellectual Property.

(i) Neither the Company nor any of its Subsidiaries has disclosed or delivered to any escrow agent or any other Person any of the source code relating to any Company Intellectual Property, and no other Person has the right, contingent or otherwise, to obtain access to or use any such source code. No event has occurred, and no circumstance or condition exists, that (with or without notice or lapse of time or both) will, or could reasonably be expected to, result in the delivery, license, or disclosure of any source code to any Person who is not, as of the Agreement Date, a current employee, consultant or contractor of the Company having a need for such access solely for the purpose of performing his or her duties to the Company or its Subsidiaries for the sole benefit of the Company or its Subsidiaries and for no other purpose.

(j) The Company and each of its Subsidiaries has taken commercially reasonable measures to protect its ownership of, and rights in, all Company Intellectual Property owned by the Company and any of its Subsidiaries in accordance with industry standard practices. Neither the Company nor any of its Subsidiaries has made any of its trade secrets available to any other Person except pursuant to written agreements restricting use solely for the benefit of the Company and requiring such Person to maintain the confidentiality of such information. Section 3.18(j) of the Company Disclosure Schedule lists all such agreements, and the Company has delivered true and complete copies of such agreements to Parent.

(k) To the Knowledge of the Company, the Company Intellectual Property does not contain any computer code designed to disrupt, disable or harm in any manner the operation of any software or hardware. To the Company's Knowledge, none of the Company Intellectual Property contains any unauthorized feature (including any worm, bomb, backdoor, clock, timer or other disabling device, code, design or routine) that causes the software or any portion thereof to be erased, inoperable or otherwise incapable of being used, either automatically, with the passage of time or upon command by any Person.

(1) Section 3.18(1) of the Company Disclosure Schedule sets forth as of the Agreement Date, a complete and accurate list of all third-party software (i) sold with, incorporated into, or distributed with or used in the development of any Commercial Product or Development Product or (ii) used or held for use by the Company or any of its Subsidiaries for any other purpose (excluding any generally available, non-customized, third party software licensed to the Company that does not require aggregate payments in any given year in excess of \$40,000 in license, maintenance, royalty or other fees), setting forth for each such item (A) all licenses and similar agreements pursuant to which the Company or any of its Subsidiaries holds rights thereto and (B) the Company Product(s) to which the item relates, if any. Except for the software that is licensed to the Company or its Subsidiaries as set forth in Section 3.16(xi) of the Company Disclosure Schedule (or not required to be listed pursuant to the exclusion in the preceding sentence), the Company exclusively owns the entire right, title, and interest in and to (including all Intellectual Property rights in) all of the material software that is used or held for use in connection with the business of the Company who have assigned all of their Intellectual Property rights therein (including all rights under copyright law and moral rights) to the Company pursuant to valid and enforceable agreement. No Persons (other than employees or consultants of the Company described in the immediately preceding sentence) have been involved in the creation, specification, development, testing, or authorship of any Company Software.

(m) Section 3.18(m) of the Company Disclosure Schedule sets forth as of the Agreement Date, a complete and accurate list of all computer software programs sold with, incorporated into, or distributed in connection with or used in the development, manufacture, use, or sale of any Commercial Product or Development Product that is, in whole or in part, subject to the provisions of any license to Publicly Available Software, setting forth for each such item (i) all licenses and similar agreements pursuant to which the Company or any of its Subsidiaries is subject or otherwise holds rights thereto, (ii) the Company Product(s) to which the item relates, and (iii) whether such item has been modified by or on behalf of the Company or any of its Subsidiaries. Except as set forth in Section 3.18(m) of the Company Disclosure Schedule, all Publicly Available Software used by the Company or any of its Subsidiaries has been used in its entirety and without modification. Neither the Company nor any of its Subsidiaries has incorporated into any Commercial Product or Development Product or otherwise accessed, used or distributed any Publicly Available Software, in whole or in part, in a manner that may (x) require, or condition the use, hosting or distribution of any Company Intellectual Property on the disclosure, licensing or distribution of any source code for any portion of such Company Intellectual Property, or (y) otherwise impose any limitation, restriction or condition on the right or ability of the Company or any of its Subsidiaries to use, host or distribute any Company Intellectual Property or Company Products or to undertake any activity, and neither the Company nor any of its Subsidiaries has any plans to do any of the foregoing.

(n) Except as set forth in Section 3.18(n) of the Company Disclosure Schedule, none of the Company's or any of its Subsidiaries' agreements (including any agreement for the performance of professional services by or on behalf of the Company or any of its Subsidiaries' with customers, agreements with merchants, agreements with outside consultants for the performance of professional services for or on behalf of the Company, any of its Subsidiaries or any of their respective customers, and any agreement or license with any end user or reseller of any Company Products), confers upon any Person other than the Company or any of its Subsidiaries any ownership right, exclusive license or other exclusive right with respect to any Company Intellectual Property developed in connection with such agreement or license.

(o) Section 3.18(o) of the Company Disclosure Schedule sets forth as of the Agreement Date, all agreements pursuant to which the Company or any of its Subsidiaries has any current development or other professional services obligations. Except as specified in Section 3.18(o) of the Company Disclosure Schedule, neither the Company nor any of its Subsidiaries has entered into any agreement to provide custom coding, new features or functionality or other software development with respect to any Company Product.

(p) Except as set forth in Section 3.18(p) of the Company Disclosure Schedule, neither the Company nor any of its Subsidiaries has (i) transferred ownership of, or granted any exclusive license with respect to, any Company Intellectual Property owned by the Company or any of its Subsidiaries to any other Person or (ii) granted any customer the right to use any Company Product or portion thereof on anything other than a non-exclusive basis or for anything other than such customer's internal business purposes.

(q) Except as set forth in Section 3.18(q) of the Company Disclosure Schedule, no funding, facilities or personnel of any educational institution or Governmental Authority were used, directly or indirectly, to develop or create, in whole or in part, any Company Intellectual Property, including any portion of a Company Product. Neither the Company nor any of its Subsidiaries is or has ever been a member or promoter of, or a contributor to, any industry standards body or similar organization that could compel the Company or such Subsidiary to grant or offer to any third Person any license or right to such Company Intellectual Property. Section 3.18(q) of the Company Disclosure Schedule sets forth a complete and accurate list of any and all grants and similar funding received by the Company or any of its Subsidiaries (including their respective predecessors), including the name of the granting authority.

(r) Except as set forth in Section 3.18(r) of the Company Disclosure Schedule, neither the Company nor any of its Subsidiaries has ever agreed to indemnify any Person for or against any interference, infringement, misappropriation, or other conflict with respect to any of the Company Intellectual Property or any Intellectual Property of any third party, excluding such provisions set forth in any license for generally available, non-customized, third party software licensed to the Company that does not require aggregate payments in any given year in excess of \$40,000 in license, maintenance, royalty and/or other fees.

3.19 FDA Matters.

(a) The Company and each of its Subsidiaries is in compliance in all material respects with all Laws and Regulations of the FDA and all other comparable state or non- U.S. Regulatory Authorities ("<u>FDA Requirements</u>"). Neither the Company nor any of its Subsidiaries has received any written notice from the FDA or any comparable state or non- U.S. Regulatory Authority alleging that the Company, any Subsidiary or any Company Product is not in compliance with FDA Requirements, nor is there any reasonable basis for such a notice. The Company has made available to Parent all correspondence with and records of communications with the FDA and other comparable state and non- U.S. Regulatory Authorities relating to the Company and its Subsidiaries, their business and the Company Products.

(b) Set forth in Section 3.19(b) of the Company Disclosure Schedule is a list of all clinical trials and studies being conducted by or on behalf of the Company or its Subsidiaries as of the Agreement Date that are subject to oversight by any Regulatory Authority. Research involving human subjects conducted by or for the Company or its Subsidiaries has been performed in compliance in all material respects with all applicable Laws and Regulations governing the protection of human subjects and the performance of such clinical trials.

(c) <u>Debarment proceedings by the FDA</u>. None of the Company, its employees nor any third party contractors are or have been listed on the debarment list maintained by the FDA pursuant to 21 U.S.C. §335(a) and §335(b) and published on the internet at the following address (or any successor address): <u>http://www.fda.gov/ora/compliance_ref/debar/default.htm</u>.

3.20 <u>Reimbursement</u>. All Government Payment Programs in which the Company or any of its Subsidiaries have contracted with or submitted claims to at any time during the three (3) years prior to the Agreement Date are listed in Section 3.20 of the Company Disclosure Schedule. Each of the Company and each of its Subsidiaries is a participating supplier or provider, in good standing, in each of the Government Payment Programs in which Company or such Subsidiary currently participates, and each of the Company and each of its Subsidiaries has complied during the last three (3) years and is currently in compliance in all material respects with all Laws applicable to such Government Payment Programs. All claims, returns, invoices and other forms submitted by the Company or any of its Subsidiaries with respect to any Government Payment Program have been true, complete, correct and accurate in all material respects during the last three (3) years. Neither the Company nor any of its Subsidiaries has submitted to any Government Payment Program any false or fraudulent claim for payment, and neither the Company nor any of its Subsidiaries has been, during the last three (3) years, and is not currently subject to any Action by any Government Payment Program.

3.21 Environmental Matters.

(a) There are no Actions or Judgments pending, or to the Knowledge of the Company, threatened by any Person or Governmental Authority against the Company or any of its Subsidiaries which assert any claim, violation of Environmental Law or noncompliance with any Environmental Approval (as defined below) or seek any Remedial Action in connection with any Environmental Law.

(b) Neither the Company nor any of its Subsidiaries has received any written notice to the effect that it is or may be liable to any Person as a result of the Release or threatened Release of a Hazardous Material, the performance of any Remedial Action, violation of Environmental Law or noncompliance with any Environmental Approval (as defined below).

(c) Each of the Company and its Subsidiaries is currently and has been in compliance, in all material respects, with all Environmental Laws. There are no facts, conditions, events or circumstances that would reasonably be expected to prevent or interfere with continued compliance in all material respects with all Environmental Laws.

(d) Except as set forth in Section 3.21(d) of the Company Disclosure Schedule, to the Knowledge of the Company, (i) no portion of any Leased Property or any real estate formerly owned, leased or operated by the Company or any of its Subsidiaries has been used for the handling, manufacturing, processing, generation, storage, treatment or disposal of Hazardous Materials, except as is customary for the operation of the business of the Company or any of its Subsidiaries as currently conducted and in compliance with applicable Environmental Laws; (ii) there have been no Releases or threatened Releases of Hazardous Materials on, in, at, to or from any Leased Property or any real property formerly owned, leased or operated by the Company or any of its Subsidiaries; (iii) any Hazardous Materials that have been handled, manufactured, processed, generated or treated by the Company or any of its Subsidiaries have been transported offsite and disposed of in accordance with applicable Environmental Laws; and (iv) there are no underground storage tanks or aboveground storage tanks present at an any Leased Property or at any real property formerly owned, leased or operated by the Company or any of its Subsidiaries.

(e) Each of the Company and its Subsidiaries has and maintains, in full force and effect, all Company Permits as are required under Environmental Laws or are otherwise necessary for the conduct of the business or operations of the Company or any of its Subsidiaries ("<u>Environmental Approvals</u>"), and the Company and each of its Subsidiaries are currently and have been in compliance in all material respects with all Environmental Approvals. Each of the Company and its Subsidiaries has made all filings and maintained all data, documentation and records required under all Environmental Laws and Environmental Approvals. Neither the Company nor any of its Subsidiaries is in default or violation, nor has any event occurred which could reasonably be expected to constitute a default or violation of any term, condition or provision of any Environmental Approval. No Environmental Approvals are required under any Environmental Laws to be transferred from the Company or any of its Subsidiaries in connection with this Agreement. Section 3.21(e) of the Company Disclosure Schedule contains a true and complete list of all Environmental Approvals now held by the Company or any of its Subsidiaries, and true and complete copies of such Environmental Approvals have been made available to Parent.

(f) The Company has made available to Parent true and complete copies of all environmental assessments, audits, and any other reports, studies, analyses, tests, or monitoring in the possession, custody or control of the Company or any of its Subsidiaries in connection with or pertaining to compliance with, or potential liability under, any Environmental Laws at the Leased Property and at property formerly owned, leased or operated by the Company or any of its Subsidiaries ("Environmental Reports").

3.22 Tax Liabilities.

(a) Each of the Company and its Subsidiaries has timely filed all Tax Returns it was required to have filed, each such Tax Return has been prepared in compliance in all material respects with applicable Law, and all such Tax Returns are true, correct and complete in all material respects. The Company has made available to Parent true, correct and complete copies of all Tax Returns with respect to income Taxes filed by or with respect to it or its Subsidiaries with respect to Tax periods ended on or after December 31, 2005 (the "Delivered Tax Returns"), and has made available to Parent all examination reports, and statements of deficiencies proposed or assessed against or agreed to by the Company and its Subsidiaries. The Company and its Subsidiaries have timely paid all Taxes that have become due and payable by the Company or its Subsidiaries, other than any Taxes for which adequate reserves in accordance with GAAP are reflected in the Company Financial Statements or, in the cases of Taxes accruing after the date of the Interim Balance Sheet, will be reflected in the Preliminary Closing Balance Sheet. The Company and its Subsidiaries have timely withheld and paid all Taxes required to have been withheld and paid in connection with amounts paid to any employee or other Person. The charges, accruals and reserves for current Taxes with respect to the Company or its Subsidiaries reflected in the Company Financial Statements are adequate to cover all Tax liabilities payable or anticipated to be payable in respect of all periods or portions thereof ending on or before the date of the Interim Balance Sheet and any Taxes of the Company or its Subsidiaries arising after such date and at or before the Effective Time have been or will be incurred in the ordinary course of the business of the Company and its Subsidiaries or as a result of transactions contemplated by this Agreement.

(b) There are no Liens for Taxes (other than current Taxes not yet due and payable) on any assets of the Company or its Subsidiaries and, to the Knowledge of the Company, there are no pending real estate Tax challenges by any Persons.

(c) Section 3.22(c) of the Company Disclosure Schedule (i) lists all federal, state, local, and foreign income Tax Returns filed with respect to the Company and its Subsidiaries; (ii) indicates those Tax Returns that have been audited; and (iii) indicates those Tax Returns and any other Tax obligations that currently are the subject of audit. There are no actions, suits, proceedings, audits, investigations or claims (including refund claims) now proposed or pending or, to the Knowledge of the Company, threatened against or with respect to the Company and its Subsidiaries concerning the Tax liability of the Company and its Subsidiaries, and no claim for assessment or collection of Taxes that previously has been asserted relating in whole or in part to the Company or its Subsidiaries remains unpaid. To the Knowledge of the Company, no issue has been raised in any examination by any Governmental Authority with respect to the Company which, by application of similar principles, reasonably would be expected to result in a proposed deficiency or increase in Taxes for any other period not so examined.

(d) No written claim has ever been made by a Tax Authority in a jurisdiction where the Company does not pay Tax or file Tax Returns that the Company is or may be subject to Taxes assessed by such jurisdiction.

(e) Except as set forth in Section 3.22(e) of the Company Disclosure Schedule, there are no outstanding agreements or waivers extending the statutory period of limitation applicable to any Tax assessment or deficiency with respect to the Company or its Subsidiaries, and the Company or its Subsidiaries have not requested any extension of time within which to file any Tax Return, which Tax Return has not yet been filed, other than extension requests with respect to the Company's and its Subsidiaries 2010 federal and applicable state income Tax Returns.

(f) The Company or its Subsidiaries have never been a member of any affiliated group of corporations (as defined in Section 1504 of the Code) or filed or been included in a combined, consolidated or unitary Tax Return. The Company or its Subsidiaries are neither a party to nor bound by any Tax sharing or allocation agreement. Neither the Company nor its Subsidiaries is presently liable, nor does the Company or its Subsidiaries have any potential liability, for the Taxes of another person (i) under Treasury Regulations Section 1.1502-6 (or any comparable provision of state, local or foreign Law), (ii) as transferee or successor, or (iii) by contract or indemnity or otherwise.

(g) The Company and its Subsidiaries will not be required, as a result of a change in method of accounting for any period ending on or before or including the Closing Date, to include any adjustment under Section 481(c) of the Code (or any similar or corresponding provision or requirement under any other Law) in Taxable income for any period (or portion thereof) ending after the Closing Date as a result of any (i) prepaid amount received on or prior to the Closing Date, or (ii) "closing agreement" described in Section 7121 of the Code (or any similar or corresponding provision of any other Law).

(h) The Company and its Subsidiaries do not operate or conduct business through any branch in any country other than the United States.

(i) Since the Company's formation, the Company has not constituted either a "distributing corporation" or a "controlled corporation" (within the meaning of Section 355(a)(1)(A) of the Code) in a distribution qualifying for tax free treatment, in whole or in part, under Section 355(a) of the Code.

(j) The Company or its Subsidiaries are not and have not been a "United States real property holding corporation" within the meaning of Section 897 of the Code at any time during the applicable period specified in Section 897(c)(1)(A)(i) of the Code.

(k) Except as set forth in Section 3.22(k) of the Company Disclosure Schedule, the Company has not made any payments, is not obligated to make any payments, and is not a party to any agreement that under any circumstances could obligate it to make any payments, that will not be deductible under Code Sections 162(m) or 280G or that could be subject to Code Section 4999, other than any such payments for which stockholder approval satisfying the requirements of Section 280G(b)(5) of the Code and the Treasury Regulations thereunder will be obtained prior to the Closing and are listed in Section 3.22(k) of the Company Disclosure Schedule.

(1) There are no outstanding rulings of, or requests for rulings by, any Tax Authority addressed to the Company or its Subsidiaries that are, or if issued would be, binding on the Company or its Subsidiaries.

(m) To the Knowledge of the Company, all persons who have purchased shares of the Company's stock that at the time of such purchase were subject to a substantial risk of forfeiture under Section 83 of the Code have timely filed elections under Section 83(b) of the Code and any analogous provisions of applicable foreign, state and local Law.

(n) The exercise price of each Company Option is not less than the fair market value of a share of Common Stock determined on the date of grant of such Option (and as of each later modification thereof, if any, within the meaning of Section 409A of the Code). Each grant or award issued under a Company Equity Incentive Plan, is exempt from Section 409A of the Code. Each plan, program, arrangement or agreement, including any Employee Plan, that constitutes in any part a nonqualified deferred compensation plan within the meaning of Section 409A of the Code is identified as such in Section 3.22(n) of the Company Disclosure Schedule. Each plan, program, arrangement or agreement identified or required to be identified in Section 3.22(n) of the Company Disclosure Schedule has been established, operated and maintained in accordance with Section 409A of the Code and applicable guidance thereunder, including but not limited the final regulations promulgated thereunder.

(o) The Company has not participated (i) in any "tax shelter" within the meaning of Section 6111 of the Code (as in effect prior to the enactment of P.L. 108-357) or any comparable Laws of jurisdictions other than the United States or (ii) in any "reportable transaction" within the meaning of Treasury Regulations Section 1.6011-4(b) or any comparable Laws of jurisdictions other than the United States.

(p) The Company and its Subsidiaries are in compliance with all terms and conditions of any Tax exemption, Tax holiday or other Tax credit or reduction agreement, approval or order of any Tax Authority, and the consummation of the Merger will not have any adverse effect on the validity and effectiveness of any such Tax exemption, Tax holiday or other Tax reduction agreement or order or otherwise result in the termination or recapture of any grant, Tax subsidy, Tax rate reduction, Tax credit, or other Tax incentive.

(q) Notwithstanding anything to the contrary herein, the Company makes no representations as to the amount of, or any limitations on the ability to use, its or its Subsidiaries' net operating losses or other tax attributes.

3.23 Insurance. Section 3.23 to the Company Disclosure Schedule sets forth as of the Agreement Date a complete list of all insurance policies and fidelity bonds covering the assets, business, equipment, properties, operations and employees, consultants and contractors of the Company or any of its Subsidiaries. Such policies are in full force and effect, all premiums due and payable under such policies and bonds have been paid (or if installment payments are due, will be paid if incurred prior to the Closing) and the Company and its Subsidiaries are otherwise in compliance in all material respects with the terms of such policies or bonds. The Company has not received any notice of cancellation or intent to cancel or increase premiums with respect to present insurance policies or bonds nor, to the Knowledge of the Company, is there any basis for any such action. There is no claim by the Company or any of its Subsidiaries pending under any of such insurance policies or bonds as to which coverage has been questioned, denied or disputed in writing or that the Company has a reason to believe will be denied or disputed by the underwriters of such policies or bonds. None of the Company or any of its Subsidiaries has ever maintained, established, sponsored, participated in or contributed to any self-insurance plan or program.

3.24 <u>Brokers</u>. None of the Company or any of its Subsidiaries has incurred, or will incur, directly or indirectly, any liability for any brokerage or finders' fees or agents' commissions or any similar charges in connection with this Agreement, nor employed or entered into any Contract with any investment banker, broker, finder, consultant or intermediary that would be entitled to any investment banking, brokerage, finder's or similar fee in connection with the Merger or the other transactions contemplated by this Agreement.

3.25 <u>Transactions with Affiliates</u>. To the Company's Knowledge, except as disclosed in Section 3.25 of the Company Disclosure Schedule, no officer, director or other Affiliate of the Company or any of its Subsidiaries has or has had since January 1, 2008, directly or indirectly, (i) an economic interest in any Person which furnished or sold, or furnishes or sells, services or products that the Company or any of its Subsidiaries, any goods or services or (iii) an economic interest in any Person that purchases from or sells or furnishes to, the Company or any of its Subsidiaries, any goods or services or (iii) except for any employment, severance, change-of-control or other employee compensation or benefit agreement, a beneficial interest in any agreement to which the Company or any of its Subsidiaries is a party or by which they or their properties or assets are bound; <u>provided</u>, <u>however</u>, that ownership of no more than five percent (5%) of the outstanding voting stock of a publicly traded corporation shall not be deemed an "economic interest in any Person" for purposes of this Section 3.25.

3.26 <u>Relationships</u>. Since January 1, 2011, neither the Company nor any of its Subsidiaries has received written notice from any customer or supplier having a material relationship with the Company or any of its Subsidiaries that such Person intends to terminate or substantially alter its existing business relationship with the Company or any of its Subsidiaries, nor has any licensor under a License Agreement with the Company or any of its Subsidiaries in writing of an intention to terminate or substantially alter the Company's or any of its Subsidiaries' rights under such License Agreement.

3.27 Payments; Foreign Corrupt Practices Act; U.S. Export and Sanctions Laws. Without limiting any provision of this Agreement:

(a) Neither the Company nor any of its Subsidiaries nor any manager, member, officer, agent, employee or other Person associated with or acting on behalf of the Company or any of its Subsidiaries has, directly or indirectly (i) paid or delivered or agreed to pay or deliver any fee, commission or other sum of money or item of property, however characterized, to any Person, government official or other party that is illegal or improper under any applicable Law, (ii) used any funds for any unlawful contribution, gift, entertainment or other unlawful expense relating to political activity, (iii) made any unlawful payment or offered anything of value to any foreign or domestic government official or employee or to any foreign or domestic political parties or campaigns, (iv) violated or is in violation of any provision of the United States Foreign Corrupt Practices Act of 1977 (15 United States Code Section 78dd-1, et seq.), as amended, or any applicable Law of similar effect, (v) made any unlawful bribe, rebate, payoff, influence payment, kickback or transfer of value to any other Person or (vi) established or maintained any fund that has not been recorded in the books and records of the Company or its Subsidiaries.

(b) The Company and each of its Subsidiaries has complied and is in compliance with applicable provisions of the United States export and sanctions laws, and regulations implemented thereunder, including the Arms Export Control Act (22 United States Code Section 2751 et seq.), as amended, the Export Administration Act (50 United States Code Section 2401 et seq.), as amended, the International Emergency Economic Powers Act (50 United States Code Section 17091 et seq.), as amended, and the various sanctions regulations administered by the Office of Foreign Assets Control of the Department of the Treasury of the United States, as amended. Without limiting the foregoing, neither the Company nor any of its Subsidiaries has made any investments or performed any Contracts in, or involving a Person from, Cuba, Iran, Sudan, Syria or Burma (Myanmar).

3.28 <u>Manufacturing and Marketing Rights</u>. Except as are set forth in Section 3.28 of the Company Disclosure Schedule, neither the Company nor any of its Subsidiaries has at any time granted rights to manufacture, produce, distribute, assemble, license, market, develop or sell any Company Products to any other person and is not bound by any agreement that affects the Company's or any of its Subsidiaries' exclusive right to manufacture, produce, distribute, assemble, license, market, develop or sell any Company Products, assemble, license, market, develop or sell any Company Products.

3.29 <u>Change of Control Agreements</u>. Section 3.29 of the Company Disclosure Schedule sets forth each plan, agreement or Company Employee Plan of the Company or any of its Subsidiaries with any Company Personnel (i) pursuant to which any amounts, including severance amounts, would become payable (whether currently or in the future) to any Person (including any employee, director or consultant) directly or indirectly as a result of the consummation of the transactions contemplated by this Agreement (either alone or upon the occurrence of any additional or subsequent event) and/or such Person's termination of employment or service at or prior to the Closing or (ii) which provides for the acceleration or early vesting of any right or benefit or lapse of any restriction directly or indirectly as a result of the consummation of the transactions contemplated by this Agreement (either alone or upon the acceleration or early vesting of any right or benefit or lapse of any restriction directly or indirectly as a result of the consummation of the transactions contemplated by this Agreement (either alone or upon the occurrence of any additional or subsequent event) and/or such Person's termination of employment or service at or prior to Closing.

3.30 Accounts Receivable; Bank Accounts.

(a) Section 3.30(a) of the Company Disclosure Schedule sets forth a list of all accounts receivable reflected in the Interim Balance Sheet, together with an aging schedule as of the Interim Balance Sheet Date, indicating the range of days elapsed since being invoiced. All of the accounts receivable of the Company and its Subsidiaries reflected in the Interim Balance Sheet (a) represent bona fide transactions that arose in the ordinary course of business and (b) are subject to no setoffs or counterclaims. The allowances for doubtful accounts set forth in the Interim Balance Sheet have been prepared in accordance with the past practices of the Company and its Subsidiaries. The receivables of the Company and its Subsidiaries arising after the date of the Interim Balance Sheet and prior to the Closing Date arose or will arise in the ordinary course of business consistent with past practice. No Person has any Lien (other than a Permitted Lien) on any account receivable, and, no written request or agreement for material deduction or material discount has been made with respect to any account receivable. Except to the extent reserved for in the aforementioned allowances for doubtful accounts, neither the Company nor any of its Subsidiaries has received written notice from any customer that such customer does not intend to pay any currently outstanding account receivable.

(b) Set forth in Section 3.30(b) of the Company Disclosure Schedule is a description of each account maintained by or for the benefit of the Company or any of its Subsidiaries at any bank or other financial institution including the authorized signatories of each account. There are no outstanding powers of attorney executed on behalf of the Company or any of its Subsidiaries.

3.31 <u>Restrictions on Business Activities</u>. There is no Judgment which specifically names the Company or any of its Subsidiaries that would prohibit, impair or otherwise limit: (a) any business practice of the Company or any of its Subsidiaries; (b) any acquisition of property (tangible or intangible) by the Company or any of its Subsidiaries; (c) the conduct of business by the Company or any of its Subsidiaries; or (d) the ability of the Company or any of its Subsidiaries to engage in any line of business or to compete or do business with any Person, in each case whether arising as a result of a change in control of the Company or any of its Subsidiaries. There is no agreement to which the Company or any of its Subsidiaries; (ii) the conduct of business in a particular or otherwise limit: (i) any acquisition of property (tangible or intangible) by the Company or any of its Subsidiaries; or field of use by the Company or any of its Subsidiaries; or (iii) the ability of the Company or any of its Subsidiaries to engage in any line of business in a particular geography or field of use by the Company or any of its Subsidiaries; or (iii) the ability of the Company or any of its Subsidiaries to engage in any line of business or to compete or do business with any Person, in each case whether arising as a result of a change in control of business or to compete or do business with any Person, in each case whether arising as a result of a change in any line of business or to compete or do business with any Person, in each case whether arising as a result of a change in control of the Company or any of its Subsidiaries.

3.32 <u>Customers; Distributors</u>. Section 3.32 of the Company Disclosure Schedule identifies, for the current fiscal year and the prior fiscal year ended prior to the Agreement Date: (a) the largest sources of the Company's revenue, which list shall account for at least seventy percent (70%) of such revenue and (b) ordering physicians and practitioners who are the largest volume purchasers of the Company Products, which list shall account for at least seventy percent (70%) of such volume. Neither the Company nor any of its Subsidiaries has received any written notice that any such customer (i) has ceased, or is planning to cease to use the current services of the Company or any of its Subsidiaries. To the Knowledge of the Company, no customer has otherwise threatened in writing to take any action described in the preceding sentence as a result of the consummation of the transactions contemplated hereby or any of the documents or instruments required hereby. To the Company Products indicating that any such distributor intends to cease acting as a distributor of such products or otherwise dealing with the Company and its Subsidiaries.

3.33 <u>Representations Complete</u>. To the Knowledge of the Company, none of the representations or warranties made by the Company in this Agreement or any Related Agreement, nor any statement made in the Company Disclosure Schedule or any certificate furnished by the Company pursuant to this Agreement (including the Company Financial Statements) or any Related Agreement, when taken together, contains any untrue statement of a material fact, or omits to state any material fact necessary in order to make the statements contained herein, in the light of the circumstances under which they were made, not misleading. There is no event, fact or condition that would reasonably be likely, individually or in the aggregate, to have a Company Material Adverse Effect that has not been set forth in this Agreement or in the Company Disclosure Schedules.

ARTICLE IV REPRESENTATIONS AND WARRANTIES OF PARENT AND ACQUISITION SUB

Parent and Acquisition Sub hereby jointly and severally represent and warrant to the Company and the Company Holders as of the Agreement Date and as of the Closing Date (except, in each case, to the extent such representations and warranties speak expressly as of an earlier date), as follows:

4.1 <u>Organization, Good Standing and Qualification</u>. Parent is a corporation duly organized, validly existing and in good standing under the Laws of the State of Delaware. Acquisition Sub is a corporation duly organized, validly existing and in good standing under the Laws of the State of Delaware. Each of Parent and Acquisition Sub has all requisite corporate power and authority to own, lease and operate its properties and assets and to carry on its business as now being conducted and is duly qualified or licensed and in corporate and tax good standing to do business in each jurisdiction in which the character of the properties owned, leased or operated by it or the nature of its activities makes such qualification necessary, except

such jurisdictions where the failure to be so qualified or licensed or in good standing would not reasonably be expected to result, individually or in the aggregate, in a material adverse change to or effect on the business, operations, assets, liabilities, financial condition or results of operations of the Parent and its Subsidiaries, taken as a whole.

4.2 <u>Ownership of Acquisition Sub; No Prior Activities</u>. Parent owns all of the issued and outstanding capital stock of Acquisition Sub, free and clear of all Liens. Acquisition Sub was formed solely for the purpose of engaging in the transactions contemplated by this Agreement and has engaged in no business activity other than as contemplated by this Agreement. Except for obligations or Liabilities incurred in connection with its incorporation and the transactions contemplated by this Agreement, Acquisition Sub has not and will not have incurred, directly or indirectly, through any Subsidiary or Affiliate, any obligations or Liabilities or engaged in any business activities of any type or kind whatsoever or entered into any agreements or arrangements with any Person. Parent has made available to the Company an accurate and complete copy of Acquisition Sub's Organizational Documents, each as amended or restated as of the Agreement Date and the Agreement Date.

4.3 <u>Authorization; Binding Obligation</u>. Each of Parent and Acquisition Sub has all requisite corporate power and authority to enter into this Agreement and the Related Agreements to which it is or will be a party, to perform its obligations hereunder and thereunder and to consummate the transactions contemplated hereby and thereby. The execution and delivery by Parent and Acquisition Sub of this Agreement, the performance of their obligations hereunder, and the consummation by Parent and Acquisition Sub of the Merger and the other transactions contemplated hereby have been duly and validly authorized by all necessary corporate action on the part of the boards of directors of each of Parent and Acquisition Sub and no other corporate proceedings on the part of Parent or Acquisition Sub are necessary to authorize this Agreement or to consummate the Merger or any other transactions contemplated hereby other than approval by Parent as the sole stockholder of Acquisition Sub and the filing of the Certificate of Merger in accordance with the DGCL. This Agreement has been, and each of the Related Agreements to which Parent or Acquisition Sub is a party will be at the Closing, duly and validly executed and delivered by each of Parent and Acquisition Sub, and, assuming the due authorization, execution and delivery by the other parties hereto and there (other than Parent and Acquisition Sub), this Agreement constitutes, and in the case of the Related Agreements they will at Closing constitute, legal, valid and binding obligations of Parent and Acquisition Sub, enforceable against each of Parent and Acquisition Sub in accordance with their respective terms, subject to applicable bankruptcy, insolvency, reorganization, moratorium, fraudulent conveyance and similar Laws affecting creditors' rights and remedies generally, and subject, as to enforceability, to general principles of equity, including principles of commercial reasonableness, good faith and fair dealing (regardless of whether enforcement is sought in a proceedin

4.4 <u>Consents</u>. The execution and delivery by each of Parent and Acquisition Sub of this Agreement and any Related Agreement required by this Agreement to be executed and delivered by each of Parent and Acquisition Sub do not, and the performance of this Agreement and any other instrument or document required by this Agreement to be executed and delivered by Parent and Acquisition Sub shall not, require Parent or Acquisition Sub to obtain any Consent

of any Person or approval of, observe any waiting period imposed by, or make any filing with or notification to, any Governmental Authority, except (a) for the filing of the Certificate of Merger in accordance with the DGCL, (b) the filing of a pre-merger notification and report form under the HSR Act and the filings and receipt, termination or expiration, as applicable, of such other approvals or waiting periods required under any other applicable competition, merger control, antitrust or similar Law, that if not obtained or made would not reasonably be expected to have a material adverse effect on the ability of Parent or Acquisition Sub to consummate the transactions contemplated by this Agreement, and (c) such other filings, authorizations, consents and approvals that if not obtained or made would not reasonably be expected to have a material adverse effect on the ability of Sub to consummate the transactions contemplated by this Agreement.

4.5 <u>No Violation</u>. The execution, delivery, compliance with and performance by Parent or Acquisition Sub of this Agreement and each of the Related Agreements delivered in connection therewith do not and will not (a) violate or contravene the Certificate of Incorporation or by-laws, each as amended, of Parent or Acquisition Sub, (b) violate or contravene any Law or Judgment to which Parent or Acquisition Sub is subject, or (c) conflict with or result in a breach of or constitute a default (or an event that with notice or lapse of time or both would become a default) by any Parent or Acquisition Sub under any material Contract to which Parent or Acquisition Sub is a party or by which Parent or Acquisition Sub or any of their respective assets or properties are bound or to which Parent or Acquisition Sub or any of their respective assets or properties are subject, except in the case of clauses (b) and (c) for such violations, conflicts, contraventions, breaches or defaults that would not reasonably be expected to have a material adverse effect on the ability of Parent or Acquisition Sub to consummate the transactions contemplated by this Agreement.

4.6 <u>Legal Proceedings</u>. There is no Judgment outstanding and there are no Actions pending or, to the knowledge of Parent, threatened by or against Parent or Acquisition Sub, whether at Law or in equity, or before or by any Governmental Authority, which could materially adversely affect such party's ability to perform its obligations under this Agreement or the consummation of the transactions contemplated by this Agreement.

4.7 <u>Brokers</u>. Parent has not employed or entered into any Contract with any investment banker, broker, finder, consultant or intermediary in connection with the transactions contemplated by this Agreement, pursuant to which the Company Holders could be liable for the fee or commission of such investment banker, broker, finder, consultant or intermediary, or for any similar fee or commission in connection with the Merger, this Agreement or the other transactions contemplated hereby.

4.8 <u>Funds</u>. As of the Effective Time, Parent will have cash and cash equivalents available in an amount sufficient to make the payments required to be made by it pursuant to Section 2.12 of this Agreement.

ARTICLE V COVENANTS

During the period commencing on the Agreement Date and continuing until the earlier of the Effective Time and the termination of this Agreement:

5.1 Financial Statements. The Company shall deliver to Parent:

(a) as soon as available, but no later than thirty (30) days after the last day of each month, an unaudited consolidated balance sheet and income statement covering the Company's consolidated operations for such month prepared in accordance with GAAP, except that such balance sheet and income statement need not contain footnotes and remain subject to changes resulting from year-end audit adjustments;

(b) as soon as available, but no later than two hundred forty (240) days after the last day of the Company's fiscal year, audited consolidated financial statements prepared under GAAP, consistently applied, together with an unqualified opinion on the financial statements from Ernst & Young LLP or other comparable independent certified public accounting firm or such other firm reasonably acceptable to Parent; provided, that the Company shall use commercially reasonable efforts to deliver the initial draft provided to such accounting firm of such financial statements to Parent as soon as it is available but no later than one hundred eighty (180) days after the last day of the Company's fiscal year; and

(c) within forty-five (45) days after the last day of each fiscal quarter, a consolidated balance sheet, income statement and statement of cash flows covering the Company's consolidated operations for such quarter prepared in accordance with GAAP, except that such balance sheet, income statement and statement of cash flows need not contain footnotes and remain subject to changes resulting from year-end audit adjustments, which shall be auditor reviewed if the Company elects to obtain such auditor review.

5.2 <u>Board Observer Rights</u>. The Company shall invite a representative designated by Parent to attend all meetings of the Company Board and all meetings of all committees of the Company Board in a nonvoting observer capacity and, in this respect, shall give such representative copies of all notices, minutes, consents, and other materials that it provides to its directors at the same time and in the same manner as provided to such directors; <u>provided</u>, <u>however</u>, that such representative shall agree to hold in confidence and trust and to act in a fiduciary manner with respect to all information so provided; and provided further, that the Company reserves the right to withhold any information and to exclude such representative from any meeting or portion thereof if, the Company Board reasonably determines that access to such information or attendance at such meeting could adversely affect the attorney-client privilege between the Company and its counsel, result in disclosure of trade secrets, present a possible conflict of interest or otherwise be contrary to the best interests of the Company.

5.3 [<u>Reserved</u>].

5.4 [<u>Reserved</u>].

5.5 Distribution of Loan Amount. The Company shall not distribute any portion of the proceeds of the Loan Amount to its securityholders.

5.6 <u>Issuance of Additional Indebtedness</u>. Except as permitted under the Loan Documents, the Company shall not incur any additional Indebtedness without the prior written consent of Parent.

5.7 Issuance of Additional Shares.

(a) If the Company issues any additional shares of capital stock that were not authorized under the Company's Certificate of Incorporation as in effect on the Agreement Date (or any securities that are exercisable or convertible therefor), then each of the Company, Parent, the Acquisition Sub and the Representative hereby agree that it will execute any amendments to this Agreement that are necessary in order to provide for the cancelation of such securities in exchange for a right to payment in accordance with this Agreement, and the Company agrees that it will as a condition precedent to any such issuance seek approval of and obtain such amendment by stockholders of the Company representing the affirmative vote of the capital stock of the Company required to approve such amendment.

(b) The Company will not grant, issue or sell any (i) shares of capital stock of the Company or any Subsidiary or (ii) options, warrants or convertible securities to purchase capital stock of the Company or any Subsidiary, unless the purchaser or recipient of such shares of capital stock, options, warrants or convertible securities executes and delivers to the Company a written consent acknowledging the existence of this Agreement and approving and consenting to the terms of this Agreement and the transactions contemplated hereby, including the Merger and the effect thereof on any such share of capital stock, option, warrant or convertible security, and the Company complies with its obligations set forth in Section 5.7(a).

5.8 Approvals and Consents; Further Assurances.

(a) The Company Board has unanimously declared this Agreement and the Merger and other transactions contemplated by this Agreement advisable and adopted a resolution recommending adoption and approval of this Agreement and the Merger by the Voting Holders. The Company shall set the Agreement Date as the record date for Voting Holders to approve and adopt this Agreement, the Merger and the transactions contemplated hereby. Promptly after execution of this Agreement, the Company shall distribute to its stockholders an action by written consent to approve and adopt this Agreement and the Merger in accordance with the Organizational Documents and pursuant to Sections 228 and 251(c) of the DGCL, and shall use commercially reasonable efforts to deliver to Parent duly executed written consents of stockholders approving and adopting this Agreement and the Merger by (i) 10:00 a.m. (Pacific Time) on February 3, 2014 for the Corporate Stockholder Approval and (ii) 11:59 p.m. (Pacific Time) on February 7, 2014 for the California Stockholder Approval. Promptly following the Agreement Date, the Company shall send (the date of such sending, the "Delivery Date"), to the Voting Holders, other than those who shall have executed and delivered a written consent prior to the Delivery Date, pursuant to Sections 228 and 251(c) of the DGCL, a notice and disclosure statement (the "Disclosure Statement"), which shall comply with applicable Laws, shall be substantially in the form and substance

previously agreed to by the Company and Parent, and shall include (A) a summary of this Agreement, the transactions contemplated hereby, including, but not limited to, the Merger, the Escrow Agreement and the Related Agreements, (B) the recommendation of the Company Board to adopt and approve this Agreement, the Escrow Agreement approve the Merger and the other transactions contemplated by this Agreement, (C) a statement that the Company Board has unanimously determined that the terms of the Merger and this Agreement are fair to and in the best interests of the Company and the Company's stockholders, and (D) a statement that appraisal rights are available for the Company Shares pursuant to Section 262 of the DGCL together with a copy of such section. From and after the Delivery Date, the Company shall use its commercially reasonable efforts to deliver to Parent duly executed written consents of Voting Holders, other than those who shall have executed and delivered a written consent prior to the Delivery Date, approving and adopting this Agreement and the Merger. The Disclosure Statement will not include any untrue statement of a material fact or omit to state any material fact necessary in order to make the statements contained herein or therein, in light of the circumstances under which made, not misleading. Promptly following the date upon which the Company obtains Stockholder Approval, the Company shall send written notice to each holder of Company Options and Company Warrants of the execution of this Agreement, a summary of the material terms of this Agreement, including Parent's right to exercise its option to acquire the Company, and any other information required by the terms of such Company Option or Company Warrant, except to the extent any such requirements have been waived in writing by such holder. Promptly following the Agreement Date, the Company shall send an additional written notice to each holder of a Company Option and Company Warrant of the anticipated closing date of the Merger, an estimate of the per share Merger Consideration that such holder would be entitled to receive if it exercises its Company Options or Company Warrants, and any other information required by the terms of such Company Option or Company Warrant, except to the extent any such requirements have been waived in writing by such holder, together with a copy of an Option Cancellation Agreement and/or Letter of Transmittal to be completed, executed and delivered by such holder.

(b) <u>Stockholder Meeting</u>. In addition to the solicitation of the Corporation Stockholder Approval required under Section 5.8(a), if, after the initial approval by the stockholders, an additional action by the stockholders of the Company is required under Delaware Law or California Law for the approval, re-approval or ratification of the Merger and this Agreement or any amendment to this Agreement at any time prior to the Effective Time, the Company shall (i) cause a meeting of the stockholders of the Company (a "<u>Stockholder Meeting</u>") to be called for purposes of approving, reapproving or ratifying the Merger and this Agreement or any amendment to this Section 5.8(b)(i), the Company shall cause such Stockholder Meeting for the same purposes. In the event that a Stockholder Meeting is required pursuant to this Section 5.8(b)(i), the Company shall cause such Stockholder Meeting. In the event of a solicitation for written consent in lieu of such Stockholder Meeting. In the event of a solicitation for written consent in lieu of such Stockholders of the Company in connection with a Stockholder Meeting. Any such information statement or proxy statement (also a Disclosure Statement) shall comply with applicable Laws and shall include (A) a summary of this Agreement and any amendments to this

Agreement, the transactions contemplated hereby, including, but not limited to, the Merger, the Escrow Agreement and all other Related Agreements, (B) the recommendation of the Company Board to adopt and approve this Agreement, as amended by any amendments to this Agreement, the Escrow Agreement and reapprove the Merger, (C) a statement that the Company Board has unanimously determined that the terms of the Merger and this Agreement, as amended by any amendments to this Agreement, are fair to and in the best interests of the Company and the Company's stockholders, and (D) a statement that appraisal rights are available for the Company Shares pursuant to Section 262 of the DGCL together with a copy of such section. The Company agrees not to distribute the Disclosure Statement until Parent has had a reasonable opportunity to review and comment on the Disclosure Statement. The Disclosure Statement will not include any untrue statement of a material fact or omit to state any material fact necessary in order to make the statements contained herein or therein, in light of the circumstances under which made, not misleading. Promptly following the date upon which the Company obtains such additional stockholder approval, the Company shall send written notice to each holder of Company Options and Company Warrants of such approval, a summary of the material terms of this Agreement, as amended by any amendments to this Agreement, and any other information required by the terms of such Company Option or Company Warrant, except to the extent any such requirements have been waived in writing by such holder.

(c) Subject to the terms and conditions of this Agreement, each party shall cooperate with the others and use its reasonable best efforts to take, or cause to be taken, all actions and to do, or cause to be done, all things necessary, proper or advisable to satisfy the conditions to Closing set forth in this Agreement and to consummate the transactions contemplated by this Agreement.

5.9 <u>Access to Information</u>. The Company shall, upon the reasonable request of Parent and during normal working hours, make members of its management team and Company Board available to the officers, employees, accountants, counsel and other representatives of Parent (the "<u>Parent</u> <u>Representatives</u>") for the purpose of providing Parent with information that Parent may reasonably request relating to the business and operations of the Company. All information obtained by Parent and the Parent Representatives pursuant to this Section 5.9 shall constitute "Confidential Information" pursuant to the terms of the Confidential Disclosure Agreement dated January 11, 2011, by and between Parent and the Company (the "<u>Confidentiality Agreement</u>"), the term of which agreement is hereby extended to the earlier of the Closing Date and the date of termination of this Agreement. No information or knowledge obtained in any investigation pursuant to this Section 5.9 shall affect or be deemed to modify or qualify any representation or warranty of the Company or the conditions to the obligations of the parties to consummate the Merger and the other transactions contemplated by this Agreement.

5.10 <u>Public Announcements</u>. Parent and the Company shall consult with and obtain the approval of (which approval shall not be unreasonably withheld) the other party before issuing any press release or other public announcement with respect to the Merger or this Agreement, and no party hereto shall issue or cause to be issued any such press release prior to such consultation and approval, except to the extent required by applicable Law, in which case the party proposing to issue such press release or make such public announcement shall use commercially reasonable efforts to consult in good faith with the other party before issuing any

such press release or making any such public announcement to attempt to agree upon mutually satisfactory text. Upon receipt of the Corporate Stockholder Approval and/or the Closing, Parent and the Company shall issue a mutually agreed upon press release announcing the transactions contemplated hereby. Notwithstanding the foregoing, nothing in this Section 5.10 shall be deemed to prohibit any party from making any disclosure that its counsel deems necessary in order to fulfill such party's disclosure obligations imposed by any Regulation of any national securities exchange or automated quotation system, so long as the disclosing party consults with the other party prior to such disclosure except that in no event shall the Company or any of the Company Representatives make any public statement or announcement regarding, or otherwise disclose, the existence, terms or subject matter of this Agreement prior to receipt of Stockholder Approval without the Parent's prior written consent.

5.11 Acquisition Proposals. The Company shall not, and will direct each officer, director, stockholder, employee, representative and agent of the Company (each a "<u>Company Representative</u>") not to, directly or indirectly, encourage, solicit, participate in or initiate discussions or negotiations with or provide any information to any Person or group (other than Parent or an Affiliate) concerning any offers or proposals for any merger of the Company, sale or license of all or substantially all of the assets of, or tender offer for or purchase of all or substantially all of the shares of capital stock of the Company or similar transactions involving the acquisition of the Company (an "<u>Acquisition Proposal</u>"). The Company and the Company Representatives shall immediately cease and cause to be terminated all discussions or negotiations with any Person or group conducted prior to the Agreement Date with respect to any Acquisition Proposal. The Company shall promptly notify Parent if it shall, on or after the Agreement Date, have received any request for information or access in connection with a possible Acquisition Proposal involving any Person or group (other than Parent or an Affiliate) or engaged in any discussions or negotiations with respect to any Acquisition Proposal. At Parent's request, the Company shall request the return or destruction of any confidential information shared in connection with any terminated discussions or negotiations with respect to any Acquisition Proposal.

5.12 <u>Employee Nondisclosure, Confidentiality and Invention Assignment</u>. The Company shall require all new employees to execute and deliver to the Company the Company's standard form nondisclosure, confidentiality and assignment of inventions agreement.

5.13 <u>Additional Loan Amounts</u>. On or before February 20, 2014, Parent shall loan up to an additional \$3,000,000 to the Company pursuant to a further amendment to the Loan Documents to purchase additional Meso Supplies. In addition, in the event that the Closing does not take place by March 10, 2014, or such other date as mutually agreed by Parent and the Company, and provided that the Company is not then in default of any provision of this Agreement, Parent shall loan to the Company such amount as is reasonably necessary to satisfy the Company's current operating expenses between such date and the Closing pursuant to a further amendment to the Loan Documents.

ARTICLE VI ADDITIONAL PRE-CLOSING COVENANTS

6.1 <u>Conduct of Business by the Company Pending Closing</u>. During the period (the "<u>Pre-Closing Period</u>") commencing on the Agreement Date and continuing until the earlier of (i) the Effective Time, and (ii) the termination of this Agreement, the Company and each of its Subsidiaries shall conduct its business in the ordinary course consistent with past practice and shall use commercially reasonable efforts to preserve intact the present business organization and maintain the material tangible assets of the Company and its Subsidiaries in their current physical condition (except for ordinary wear and tear), to maintain the material intangible assets of the Company (including all Company Intellectual Property), to keep available the services of the Company's present employees, to maintain in effect the Material Contracts (subject to the expiration of any Material Contract pursuant to its terms) and to preserve the present relationships of the Company and its Subsidiaries, customers, dealers, licensees and other Persons with which the Company and its Subsidiaries has business relations. In addition and without limiting the generality of the foregoing, except (i) as specifically contemplated or permitted by this Agreement, (ii) as required by applicable Laws, (iii) as set forth in <u>Schedule 6.1</u> (as it may be updated by agreement of the Company and Parent), or (iv) unless Parent expressly consents in writing in advance (such consent not to be unreasonably withheld, conditioned or delayed, and which consent may be given by e-mail or other electronic transmission), the Company and each of its Subsidiaries agrees that it will:

(a) not do any of the following: (i) sell, assign, transfer, deliver, license or otherwise dispose of any Company Intellectual Property; (ii) create or permit to exist any encumbrance on any Company Intellectual Property; (iii) respond or fail to respond to any communication where such response or failure to respond could adversely affect any Company Intellectual Property (except where Parent fails to provide its prior written consent in accordance with clause (iv) above after being provided with reasonable advance written notice thereof by the Company); (iv) amend or terminate any License Agreement or fail to perform and fulfill its obligations under any License Agreements in accordance with the terms thereof; (v) consent to any assignment of any License Agreements by the licensors thereof; or (vi) agree to do any of the foregoing or take any other action that does or would materially reduce or otherwise materially adversely affect the Company Intellectual Property or cause any representation or warranty set forth in Section 3.18 to become untrue as of the date of such action, including (A) granting or permitting any license of any Company Intellectual Property or (B) failing to diligently process, prosecute and maintain in full force and effect any Company Intellectual Property;

(b) not threaten in writing, initiate, or take any steps towards the commencement of any Action against any third party other than to enforce the Company's rights to Company Intellectual Property;

(c) not sell, license, mortgage or otherwise encumber or subject to any Lien (other than a Permitted Lien), or otherwise dispose of any, or any interest in any, properties or assets, which are material, individually or in the aggregate, to the Company and its Subsidiaries, taken as a whole;

(d) not adopt a plan of complete or partial liquidation, dissolution, merger, consolidation, restructuring, recapitalization, stock split or other reorganization of the Company (other than the Merger), declare any dividend or make any payment or distribution to any of the Company's stockholders, option holders, warrant holders or note holders, change the Company's certified public accountants, the Company's fiscal year ending date or any method of accounting or accounting practice by the Company, except for any such change required by reason of a change in GAAP or applicable Law, or amend or waive any provision in the Organizational Documents;

(e) not enter into any non-compete agreements or exclusivity agreements other than non-compete agreements applicable to employees or contractors (who are individuals) of the Company and its Subsidiaries;

(f) not enter into, amend in any material respect or terminate any Material Contract;

(g) not make, except in the ordinary course of business consistent with past practice, any new expenditure or intentionally incur any new Liability resulting in an obligation of the Company or its Subsidiaries, on a consolidated basis, in excess of \$250,000 for each such expenditure or Liability in any calendar month or \$750,000 for all such expenditures and Liabilities in the aggregate in any calendar month;

(h) not acquire or agree to acquire by merging or consolidating with, or by purchasing any equity interest in or a material portion of the assets of, or by any other manner, any business or any corporation, partnership, association, or other business organization or division thereof, enter into any joint venture, partnership, collaboration, limited liability company, or operating agreement with any Person or enter into any new line of business;

(i) not increase the amount of any bonus, bonus opportunity, salary or other compensation to any director, officer, member of senior management or employee, except for any such compensation and benefit increases (1) required pursuant to employment and other compensation agreements disclosed in the Company Disclosure Schedule, and (2) awarded as part of the Company's customary annual performance reviews in accordance with the Company's past practices, or grant any increase in benefits under any Employee Plan, or amend any existing Employee Plan, or adopt any new Employee Plan except to the extent required by Law or to comply with the Code);

(j) not make or change any election in respect of Taxes or change any Tax accounting period, adopt or request permission of any Tax Authority to change any accounting method in respect of Taxes, enter into any closing agreement in respect of Taxes, settle any claim or assessment in respect of Taxes, surrender or allow to expire any right to claim a refund of Taxes, consent to any extension or waiver of the limitation period applicable to any claim or assessment in respect of Taxes, or make any application for, negotiate or conclude any Tax ruling or arrangement with a Tax Authority, or take (or permit any Subsidiary to take) any such actions with respect to any Subsidiary of the Company, in each case whether or not in connection with the Merger;

(k) not engage in any action that would reasonably be expected to, or fail to take any action the failure of which to take would reasonably be expected to, directly or indirectly, result in a Company Material Adverse Effect or that would reasonably be expected to adversely affect the ability of the Company to consummate the Merger on the terms and conditions of this Agreement, require the consent of any third party in advance of or as a result of the transactions contemplated hereby or require the Company to re-solicit Stockholder Approval for this Agreement and the transactions contemplated hereby, including the Merger, except as otherwise permitted by Section 5.8(a); and

(l) not authorize any of, or commit or agree to take any of, the foregoing actions.

6.2 Antitrust Notification.

(a) Following written notice from Parent to the Company, but no later than five (5) Business Days following the Agreement Date, each of Parent and the Company shall (i) file with the United Stated Federal Trade Commission (the "<u>FTC</u>") and the United States Department of Justice (the "<u>DOJ</u>") the premerger notification and report form, if any, required or permitted as a result of the Merger and the other transactions contemplated hereby, and shall include any supplemental information requested in connection therewith pursuant to the Hart-Scott-Rodino Antitrust Improvements Act of 1976 (the "<u>HSR Act</u>") and (ii) make such other filings as are requested by Parent or are necessary or advisable in other jurisdictions in order to comply with all applicable Laws relating to competition, merger control or antitrust and shall promptly provide any supplemental information requested by any applicable Governmental Authority relating thereto (collectively, the "<u>Antitrust Filings</u>"). Any such Antitrust Filing and any supplemental information relating thereto shall be in substantial compliance with the requirements of the HSR Act or such other applicable Laws. To the extent permitted by applicable Law, the parties shall work together and shall furnish to one another such necessary information and reasonable assistance as the other may request in connection with its preparation of any filing or submission which is necessary under the HSR Act or such other applicable Laws.

(b) The parties shall use commercially reasonable efforts to obtain any clearance required under the HSR Act or such other applicable Laws for the Merger and the other transactions contemplated hereby, including (i) to secure the expiration or termination of any applicable waiting period under the HSR Act, and, in connection therewith, the parties agree to affirmatively request early termination as part of the filing under the HSR Act; (ii) to resolve any objections asserted with respect to the Merger contemplated hereby raised by any Governmental Authority; and (iii) to prevent the entry of, and to have vacated, lifted, reversed or overturned, any decree, judgment, injunction or other order that would prevent, prohibit, restrict or delay the Closing. The parties shall keep one another apprised of the status of any communications with, and any inquiries or requests for additional information from, the FTC, the DOJ or any other applicable Governmental Authority and shall comply promptly with any such inquiry or request. The Company and Parent shall each cooperate reasonable with the other in connection with resolving any inquiry or investigation by any Governmental Authority relating to the Antitrust Filings. The Company and Parent shall each give the other reasonable advance notice of, and the opportunity to participate in (directly or through its representatives) any inquiry or investigation by, or any material meeting or conference (whether by telecommunications or in person) with, any Governmental Authority relating to the Antitrust

Filings if, in the reasonable judgment of the party that is subject to the inquiry, investigation, meeting or conference, such participation by the other party is prudent and (based upon the advice of legal counsel) legally permissible.

(c) Notwithstanding anything to the contrary in this Agreement, nothing shall require or be construed to require Parent or any of its Affiliates, in order to obtain any clearance required or successful termination or expiration of any review of any Governmental Authority regarding the Merger or the other transactions contemplated by this Agreement, to (i) sell or hold separate, or agree to sell or hold separate, before or after the Effective Time, any assets, businesses or any interests in any assets or businesses, of Parent or any of its Affiliates or of the Surviving Corporation or any of its subsidiaries (or to consent to any sale, or agreement to sell, by Parent, by the Surviving Corporation or by any of their respective Affiliates of any assets or businesses, or any interests in any assets or businesses), or any change in or restriction on the operation by Parent of any assets or businesses (including any assets or businesses of the Surviving Corporation or any of its subsidiaries), (ii) enter into any agreement or be bound by any obligation that Parent may deem in its sole discretion to have an adverse effect on the benefits to Parent of the Merger, or (iii) initiate or participate in any legal proceeding with respect to any such matters.

(d) In the event that Parent is required, in order to obtain the consent or successful termination or expiration of any review under any Law regarding the Merger or the other transactions contemplated by this Agreement, to take any of the actions set forth in Section 6.2(c) or if such consent, successful termination or expiration has not been obtained within ninety (90) days following the date of Parent filing its Antitrust Filing under any applicable Antitrust Law, Parent shall have the right to abandon its efforts to obtain approval under such Antitrust Law of the Merger and the other transactions contemplated by this Section 6.2. If Parent so elects to abandon its efforts to seek such approval, it shall promptly give notice of such abandonment to the Company.

6.3 Filings and Consents; Further Assurances.

(a) During the Pre-Closing Period, each of Parent, Acquisition Sub and the Company shall, as promptly as practicable, use commercially reasonable efforts to obtain all other necessary Consents from Governmental Authorities and make all other necessary registrations and filings under applicable Law required in connection with the authorization, execution and delivery of this Agreement and the consummation of the transactions contemplated hereby, including the Merger. Parent, Acquisition Sub and the Company shall act in good faith and reasonably cooperate with the other in connection therewith and in connection with resolving any investigation or other inquiry with respect thereto. To the extent not prohibited by Law, each party to this Agreement shall use commercially reasonable efforts to furnish to each other all information required for any application, notification or other filing to be made pursuant to any Law and/or Company Permit in connection with the transactions contemplated by this Agreement. Each of Parent, Acquisition Sub and the Company, on the one hand, nor Parent or any of its Subsidiaries, on the other hand, shall independently participate in any meeting, or engage in any substantive conversation, with

any Governmental Authority in respect of any such Consent, investigation or other inquiry without giving Parent or the Company, as the case may be, prior notice of the meeting and discussing with Parent or the Company, as the case may be, the advisability of Parent's or the Company's representatives, as the case may be, participating in such meeting or conversation.

(b) The Company shall use commercially reasonable efforts to obtain all Consents from third parties that are set forth in Section 3.6 of the Company Disclosure Schedule ("<u>Company Third Party Consents</u>"). Parent and Acquisition Sub shall use their commercially reasonable efforts to obtain all Consents from third parties that are required for the consummation of the Merger.

(c) Section 280G Stockholder Approval. Prior to the Closing Date or at such other time as agreed by the Company and Parent, the Company shall submit to a vote of the stockholders of the Company, in a manner reasonably satisfactory to Parent, the right of any "disqualified individual" (as defined in section 280G(c) of the Code) to receive any and all payments, benefits or other compensation (collectively, the "<u>280G Payments</u>") contingent on the consummation of the transactions contemplated by this Agreement (within the meaning of section 280G(b)(2)(A)(i) of the Code) to the extent necessary so that no payment, benefit or other compensation received by such "disqualified individual" would be a "parachute payment" under section 280G(b) of the Code, in a manner that satisfies the stockholder approval requirements under section 280G(b)(5)(B) of the Code and regulations promulgated thereunder (the "<u>280G Approval</u>"). Consistent with section 280G(b)(5) and the regulations promulgated thereunder, such vote shall establish the right of the "disqualified individual" to the payment, benefit or other compensation being submitted to such vote. In addition, before the vote is submitted to the stockholders of the Company, the Company shall provide adequate disclosure to all of the stockholders entitled to vote, as determined under Section 280G, of all material facts concerning all payments, benefits or other compensation that, but for such vote, could be deemed "parachute payments" to a "disqualified individual" under section 280G of the Code in a manner that satisfies section 280G(b)(5)(B)(i) of the Code and regulations promulgated thereunder. Parent and its counsel shall have the right and a reasonable opportunity to review and comment on all documents to be delivered to the stockholders entitled to vote in connection with such 280G Approval. The Company shall secure waivers of any indemnification provisions in any agreements with disqualified individuals to the extent such provision indemnifies any such disqualified indiv

6.4 Directors' and Officers' Liability Insurance; Assumption of Indemnification Agreements.

(a) Pursuant to the provisions of its Certificate of Incorporation and by-laws, Parent shall, and Parent shall cause the Surviving Corporation to, indemnify and hold harmless the present and former officers, directors, employees or agents (including as a fiduciary with respect to an employee benefit plan) of the Company, any of its Subsidiaries or any of their respective predecessors (the "<u>Indemnified D&Os</u>") in respect of acts or omissions occurring while the Indemnified D&Os are officers, directors, employees or agents to the same extent as is provided under the Organizational Documents as of

the Agreement Date. Neither Parent nor the Surviving Corporation will amend, repeal or modify such provisions in any manner that would adversely affect the rights thereunder of such persons; provided, that, such indemnification shall be subject to any limitation imposed from time to time under applicable Law.

(b) Prior to the Effective Time, Parent shall purchase an extended reporting period endorsement under the Company's existing directors' and officers' liability insurance coverage and employment practices liability insurance coverage in a form reasonably acceptable to the Company that shall provide the members of the Company Board and the Company's officers with coverage for six (6) years following the Effective Time of not less than the existing coverage and have other terms not materially less favorable to the insured persons than the Company's directors' liability insurance coverage maintained by the Company at that time; provided, however, without Parent's agreement, Parent shall not be required to pay an aggregate cost for such endorsement greater than two times (the "<u>D&O Cap</u>") the annual premium at the time of purchase; and provided further, however, that if equivalent coverage cannot be obtained, or can be obtained only by paying an annual premium in excess of the D&O Cap.

(c) Parent shall assume, undertake, and agree to pay, perform, fulfill and discharge, from and after the Effective Time, the obligations of every nature of the Company under any indemnification agreements set forth in Section 3.17 of the Company Disclosure Schedule in accordance with the terms and conditions thereof.

(d) The provisions of this Section 6.4 shall survive consummation of the Merger and are expressly intended to benefit each of the Indemnified D&Os (or any of their heirs or representatives) who are entitled to indemnification by the Company as of the Agreement Date.

6.5 <u>Insurance</u>. The Company shall use commercially reasonably efforts to keep all insurance policies currently in effect, or comparable replacements therefor, in full force and effect through the Effective Time and to cause such insurance policies to be in full force and effect immediately following the Effective Time.

6.6 Employee Benefits.

(a) Parent and the Surviving Corporation shall ensure that each employee of the Company and its Subsidiaries who continues employment with Parent, the Surviving Corporation, or any of their respective Subsidiaries, effective as of the Closing Date, shall receive (i) salary and commission compensation for not less than twelve (12) months following the Closing Date that is at least as favorable to such employee as the salary compensation provided to such employee by the Company and its Subsidiaries prior to the Closing Date, and (ii) bonus compensation and employee benefits (including equity-based compensation) equivalent to the bonus compensation and employee benefits provided to similarly situated employees of Parent, subject to applicable Law. All employees of the Company and its Subsidiaries who continue employment with Parent or the Surviving Corporation, or any of their respective Subsidiaries, as applicable, at the Closing are hereinafter referred to as the "Continuing Employees".

(b) Parent shall use commercially reasonable efforts to, or shall cause the Surviving Corporation to use commercially reasonable efforts to, (i) provide that for purposes of eligibility to participate and vesting in benefits and benefit accruals in the case of paid time-off, the Continuing Employees will be credited with their years of service with the Company and its Subsidiaries and any predecessors thereof, but only to the extent such service was taken into account for comparable purposes under the applicable Employee Plans prior to the Closing Date, (ii) provide that the eligibility of the Continuing Employees to participate in any welfare benefit plan or program of the Company, Parent or their respective Subsidiaries, as applicable, will not, subject to the terms of the applicable plans and applicable Law, be subject to any eligibility waiting periods, evidence of insurability requirements or pre-existing condition limitations (to the extent such limitations did not apply to a pre-existing condition under the applicable Employee Plans prior to the Closing Date) and (iii) provide that each Continuing Employee will be given credit for all amounts paid by such Continuing Employee under any similar benefit plan for the plan year that includes the Closing Date for purposes of applying deductibles, co-payments and out-of-pocket maximums as though such amounts had been paid in accordance with the terms and conditions of the applicable plan maintained by Parent, Surviving Corporation, or any of their respective subsidiaries for the plan year in which the Closing Date occurs.

(c) Parent shall provide COBRA continuation coverage to all individuals who are M&A qualified beneficiaries (within the meaning assigned to such term under Q&A-4 of Treasury regulation Section 54.4980B-9) with respect to the transactions contemplated by this Agreement for the duration of the period to which such individuals are entitled to such coverage.

(d) Nothing contained in this Section 6.6 shall create any rights of any nature or kind whatsoever in any Continuing Employee or any other officer, contractor or employee or former officer, contractor or employee (including any beneficiary or dependent thereof) of the Company and its Subsidiaries in respect of continued employment for any specified period. This Section 6.6 shall inure exclusively to the benefit of and be binding upon the parties hereto and their respective successors, assigns, executors and legal representatives. Nothing in this Section 6.6, express or implied: (i) is intended to confer on any Person (including any Continuing Employee) other than the parties hereto or their respective successors and assigns any rights, remedies, obligations or liabilities under or by reason of this Agreement; (ii) shall require the Surviving Corporation or Parent or either of its Subsidiaries to employ any individual for any specific period of time; or (iii) shall constitute an amendment to any employee benefit plan or program.

6.7 Tax Matters.

(a) <u>Conduct of Business</u>. During the Pre-Closing Period, neither the Company nor any of its Subsidiaries shall make or change any material election in respect of Taxes, adopt or change any accounting method in respect of Taxes, enter into any closing agreement, settle any claim or assessment in respect of Taxes, file any amended Tax Return, surrender any refund claim, or consent to any extension or waiver of the limitation period applicable to any claim or assessment in respect of Taxes, except with the prior written consent of Parent, which shall not be unreasonably withheld.

(b) <u>Certain Taxes</u>. All transfer, documentary, sales, use, stamp, registration and other such Taxes and fees (including any penalties and interest) incurred in connection with the transactions contemplated by this Agreement shall be paid by the Company Holders when due, and the Company Holders will, at their own expense, file all necessary Tax Returns and other documentation with respect to all such transfer, documentary, sales, use, stamp, registration and other Taxes and fees, and, if required by applicable Law, Parent will join in the execution of any such Tax Returns and other documentation. At Parent's discretion, the amount paid to any Person pursuant to this Agreement will be reduced by the amount of Taxes payable by such Person pursuant to this Section 6.7(b). Any amount so withheld will be promptly remitted to the appropriate Tax Authority.

(c) <u>Tax Returns</u>. During the Pre-Closing Period, the Company shall, and shall cause each of its Subsidiaries to, timely file (or cause to be timely filed) all of its Tax Returns as they become due (taking all valid extension requests into account), all such Tax Returns to be true, correct and complete in all material respects, and the Company shall, and shall cause each of its Subsidiaries to, timely pay and discharge as they become due and payable all Taxes (other than Taxes contested in good faith by the Company in appropriate proceedings), assessments and other governmental charges or levies imposed upon it or its income or any of its property as well as all claims of any kind (including claims for labor, materials and supplies) that, if unpaid, may by Law become a lien or charge upon its properties. During the Pre-Closing Period, the Company will remain in compliance with all terms and conditions of any Tax exemption, Tax holiday, or other Tax credit or Tax reduction agreement, approval, or order of any Tax Authority, and will not take any action that would have any adverse effect on the validity or effectiveness of any such Tax exemption, Tax holiday or other Tax incentive under such an agreement.

(d) <u>Cooperation on Tax Matters</u>. The Company (and after the Closing, the Representative) and Parent shall cooperate fully, as and to the extent reasonably requested by the other, in connection with the preparation and filing of any Tax Return, any audit, litigation or other proceeding with respect to Taxes. Such cooperation shall include the retention and, upon such a party's request, the provision of records and information that are reasonably relevant to any such audit, litigation or other proceeding and within such party's possession and reasonable access to employees on a mutually convenient basis to provide additional information and explanation of any material provided hereunder. Parent and the Company (to the extent applicable) agree to (i) retain all books and records with respect to Tax matters pertinent to the Company and its Subsidiaries relating to any Pre-Closing Tax Period, and to abide by all record retention agreements entered into with any Governmental Authority and (ii) use commercially reasonable efforts to give the other party reasonable written notice prior to destroying or discarding any such books and records and, if the other party so requests, Parent and the Representative, as the case may be, shall allow the other party to take possession of such books and records.

(e) <u>Tax Certificates</u>. Parent and the Company (and after the Closing, the Representative) agree, upon request, to use commercially reasonable efforts to obtain any certificate or other document from any Governmental Authority or customer of the Company or any of its Subsidiaries or any other Person as may be necessary to mitigate, reduce or eliminate any Tax that could be imposed (including but not limited to with respect to the transactions contemplated by this Agreement).

(f) <u>Tax Contests</u>. Notwithstanding any of the foregoing, following the Closing, the provisions of Section 10.4 with respect to Third Party Claims shall govern with respect to any Tax audit or other Tax contest relating to the Company or any of its Subsidiaries if such Tax audit or Tax contest could give rise to an obligation on the part of the Company Holders to indemnify hereunder.

(g) <u>Allocation</u>. For purposes of this Section 6.7, in the case of any Taxes that are imposed on a periodic basis and are payable for a Straddle Period, the portion of such Tax related to the portion of such Tax period ending on and including the Closing Date shall (i) in the case of any Taxes other than gross receipts, sales or use Taxes and Taxes based upon or related to income, be deemed to be the amount of such Tax for the entire Tax period multiplied by a fraction, the numerator of which shall be the number of days in the Tax period ending on and including the Closing Date and the denominator of which shall be the number of days in the entire Tax period, and (ii) in the case of any Tax based upon or related to income and any gross receipts, sales or use Tax, be deemed equal to the amount that would be payable if the relevant Tax period ended on and included the Closing Date. All determinations necessary to give effect to the allocation set forth in the foregoing clause (ii) shall be made in a manner consistent with the prior practice of the Company and its Subsidiaries.

(h) <u>Tax Refunds</u>. If the Company is entitled to a refund or credit of income Taxes for any period ending prior to the Closing Date that is attributable to the carryback of losses, credits or similar items of the Company from any period beginning after the Closing Date and if the refund or credit is paid to Parent prior to the Escrow Termination Date, Parent shall forward the amount of such refund to the Escrow Agent for distribution to the Company Holders promptly after receipt, together with interest or other amount received in connection therewith.

6.8 <u>Allocation Certificate</u>. At least five (5) Business Days prior to the Closing Date, the Company shall deliver to Parent a certificate (the "<u>Allocation</u> <u>Certificate</u>") of the Company signed by the Representative (on behalf of the Company Holders) and by the Chief Financial Officer of the Company certifying as to the accuracy and completeness, in each case as of the Closing, of:

(a) the Company's calculation of all of the amounts used to calculate (and a calculation of each of) the Closing Payment, Closing Payment Reduction Amount, Closing Option/Warrant Amount, Fully Diluted Common Share Number, Fully-Diluted Preferred Shares, Per Share Common Closing Payment, Per Share Common Escrow Payment, Per Share Common Reimbursement Payment, Per Share Common Working Capital Payment, Per Share Preferred Closing Payment, Per Share Preferred Escrow Payment, Per Share Preferred Reimbursement Payment, Per Share Preferred Working Capital Payment, and any withholding obligations that apply to any payments;

(b) the identity and mailing address of each Company Holder, indicating (i) the number and type of Company Shares held by each such holder; (ii) (A) the number of Company Shares subject to Company Options held by such holder and (B) the exercise prices and extent vested thereof, (iii) (A) the number of any type of Company Shares subject to Company Warrants held by such holder and (B) the exercise prices and extent vested thereof, and (iv) (A) the number of any type of Company Shares subject to any other securities of the Company held by such holder and (B) the exercise prices and extent vested thereof; and

(c) the amount of (i) the Closing Payment to be paid to each Company Holder; (ii) each Company Holder's portion of any Working Capital Surplus, expressed as a percentage, (iii) each Company Holder's portion of the Escrow Consideration, expressed as a dollar amount and as a percentage; (iv) the portion of the Merger Consideration payable to each Company Holder without reduction for such Company Holder's portion of the Escrow Consideration; and (iv) any required withholdings (if any) from each such amount with respect to each Company Holder. Any portion of the Closing Payment and Working Capital Surplus to be paid to holders of Company Options shall be shown separately so as to allow the parties to determine the amounts to be paid by Parent and the amount to be paid by the Paying Agent.

6.9 Company Options and Company Warrants. The Company shall use commercially reasonable efforts to ensure that all Company Warrants and Company Options shall have been (i) exercised in full in accordance with their terms and the exercise price with respect thereto paid to the Company, (ii) terminated or expired no later than the Effective Time in accordance with their terms and/or (iii) converted solely into the right to receive the payments with respect thereto set forth in this Agreement, such that, as of immediately following the Effective Time, there will be no Company Warrants or Company Options outstanding (whether or not vested).

6.10 Pay-Off Letters. The Company shall use commercially reasonable efforts to cause the lenders of the Company and each Subsidiary, other than with respect to the Loan Amount, to prepare and deliver to the Company and Parent the Pay-Off Letters in a form reasonably acceptable to Parent no later than three (3) Business Days prior to the Closing Date, which Pay-Off Letters shall be updated, as necessary, on the Closing Date to specify the aggregate amount of Indebtedness outstanding thereunder as of the Closing Date. For the avoidance of doubt, to the extent that any convertible notes will not be converted into equity securities of the Company prior to or at the Effective Time in accordance with the terms and conditions of such notes, all amounts outstanding thereunder will be included as Company Indebtedness in the Closing Payment Reduction Amount, and upon payment by Parent of such amounts outstanding to the holders of such notes, no portion of the notes will remain convertible into equity securities of the Company.

6.11 <u>Notice of Certain Events</u>. During the Pre-Closing Period, each party hereto shall promptly notify the other parties hereto of (a) any event, condition, fact, circumstance, occurrence, transaction or other item of which such party becomes aware that would constitute a violation or breach by such party of this Agreement (or a breach by such party of any representation or warranty made by it contained herein) or, if the same were to continue to exist as of the Closing Date, would constitute the non-satisfaction of any of the conditions to the other party's obligation to consummate the Merger set forth in Article VII hereof, and (b) any event, condition, fact, circumstance, occurrence, transaction or other item of which such party becomes

aware which would have been required to have been disclosed by such party pursuant to the terms of this Agreement had such event, condition, fact, circumstance, occurrence, transaction or other item existed as of the Agreement Date; <u>provided that</u>, no such update shall be deemed to supplement or amend the Company Disclosure Schedule for the purpose of determining the accuracy of any of the representations and warranties made by the Company in this Agreement, including for purposes of Article VII or Article X hereof; and <u>provided further</u>, <u>however</u>, that any breach of this Section 6.11, to the extent relating to a breach of representation and warranty, shall be treated as breach of such representation and warranty (and not a breach of a covenant) for purposes of Article X.

ARTICLE VII CONDITIONS PRECEDENT TO THE MERGER

7.1 <u>Conditions to Obligation of Each Party to Effect the Merger</u>. The respective obligations of each party to effect the Merger shall be subject to the satisfaction at or prior to the Effective Time of the following conditions:

(a) <u>Stockholder Approval</u>. This Agreement and the Merger and the transactions contemplated hereby shall have been approved and adopted by the requisite vote of the stockholders of the Company in accordance with the terms of the DGCL, the Organizational Documents and, if applicable, California Law.

(b) <u>Governmental Consents</u>. All Consents of, or declarations or filings with, any Governmental Authority required to be obtained or made in order to consummate the Merger, if any, shall have been obtained or made.

(c) <u>No Injunctions or Restraints; Illegality</u>. No temporary restraining order, preliminary or permanent injunction or other Judgment (whether temporary, preliminary or permanent) issued by any Court of competent jurisdiction or other legal restraint or prohibition shall be in effect which prevents the consummation of the Merger, nor shall any proceeding brought by or before any Governmental Authority seeking any of the foregoing be pending, and there shall not be any action taken, or any Law or Judgment enacted, entered, enforced or deemed applicable to the Merger, which makes the consummation of the Merger illegal.

(d) <u>Certificate of Merger</u>. Prior to the Effective Time and following the satisfaction or waiver of all of the other conditions set forth in this Article VII, the Certificate of Merger shall be accepted for filing with the Secretary of State of the State of Delaware.

7.2 <u>Additional Conditions to Obligations of Parent and Acquisition Sub</u>. The obligations of Parent and Acquisition Sub to effect the Merger shall be subject to the satisfaction at or prior to the Effective Time of the following additional conditions:

(a) <u>Representations and Warranties</u>. The representations and warranties of the Company set forth in this Agreement that are qualified as to materiality (including in the definition of Company Material Adverse Change) shall be true and correct, and all other representations and warranties of the Company set forth in this Agreement that are not so qualified shall be true and correct in all material respects, in each case as of the Agreement Date and the Closing Date (as qualified by the Company Disclosure Schedules) with the same effect as though made as of the Agreement Date and the Closing Date, except that the accuracy of representations and warranties that by their terms speak as of a specified date will be determined as of such date.

(b) <u>Agreements and Covenants</u>. The Company shall have performed or complied, in all material respects, with each obligation, agreement and covenant to be performed or complied with by it under this Agreement at or prior to the Effective Time.

(c) <u>Consents</u>. Each of the Company Third Party Consents set forth in Schedule 7.2(c) (as it may be updated by agreement of the Company and Parent) shall have been received in form and substance reasonably acceptable to the Parent.

(d) <u>Resignations</u>. Parent shall have received copies of the resignations, effective as of the Closing, of each officer of the Company and each of its Subsidiaries and each member of the Company Board and the board of directors or similar body of each of its Subsidiaries (other than any such resignations which Parent designates, by written notice to the Company, as unnecessary).

(e) <u>Officer's Certificate</u>. The Company shall have delivered to Parent a certificate of the President and Chief Executive Officer of the Company, dated as of the Closing Date, certifying that the conditions set forth in Sections 7.1(a) and (c), and 7.2(a), (b), (l) and (p) have been satisfied.

(f) Allocation Certificate. The Company shall have delivered to Parent the Allocation Certificate pursuant to Section 6.8.

(g) <u>Secretary's Certificate</u>. The Company shall have delivered to Parent a certificate of the Company executed by the Secretary of the Company, dated as of the Closing Date, certifying the following: (i) the resolutions of the Board of Directors of the Company authorizing the execution, delivery and performance this Agreement and the Related Agreements and the transactions contemplated hereby and thereby; (ii) the resolutions of the Voting Holders authorizing the execution, delivery and performance this Agreement and the Related Agreement and the Related Agreements and the transactions contemplated hereby and thereby; (iii) the Organizational Documents of the Company and each of its Subsidiaries, and (iv) an incumbency certificate with the name, title, incumbency and signatures of the officers authorized to execute the agreements and other documents required to be delivered by the Company or any of its Subsidiaries in connection with the Closing pursuant to this Agreement.

(h) <u>Good Standing Certificates</u>. The Company shall have delivered to Parent with respect to the Company, a certificate of good standing from the Secretary of State of the State of Delaware and any other jurisdictions in which it is qualified, and with respect to each of the Company's Subsidiaries, a certificate of good standing from its jurisdiction of incorporation or formation and any other jurisdictions in which it is qualified.

(i) [<u>Reserved</u>].

(j) <u>Closing Statement</u>. The Parent shall have received the Preliminary Closing Statement from the Company.

(k) <u>FIRPTA Statement</u>. The Company shall have delivered a properly executed statement, dated as of the Closing Date, in a form reasonably acceptable to Parent, conforming to the requirements of Treasury Regulations Sections 1.1445-2(c)(3) and 1.897-2(h).

(1) <u>No Material Adverse Change</u>. No event, occurrence, change, effect or condition of any character shall have occurred that, individually or in the aggregate, has resulted in or would reasonably be expected to result in a Company Material Adverse Change other than an Excluded Company Material Adverse Change.

(m) Escrow Agreement. The Representative shall have delivered to Parent a fully executed Escrow Agreement.

(n) Paying Agent Agreement. The Representative shall have delivered to Parent a fully executed Paying Agent Agreement.

(o) <u>Stockholder Approval at a Stockholder Meeting</u>. In the event the Company is required under Delaware Law or California Law to obtain the approval, re-approval or ratification of this Agreement or any amendment to this Agreement by the Company's stockholders at any time prior to the Effective Time, the Company shall have complied in all respects with Section 5.8(b) and such additional stockholder approval shall have been obtained.

(p) <u>No Dissenter's Rights</u>. The number of Dissenting Shares shall not exceed 5% of the aggregate number of Company Shares outstanding on the Closing Date on an as-converted basis.

(q) <u>Company Options and Company Warrants</u>. All Company Options and Company Warrants shall have been (i) exercised in full in accordance with their terms and the exercise price with respect thereto paid to the Company, (ii) terminated or expired no later than the Effective Time in accordance with their terms and/or (iii) converted solely into the right to receive the payments with respect thereto set forth in this Agreement, in each case to the reasonable satisfaction of the Parent, such that as of immediately following the Effective Time there are no Company Warrants or Company Options outstanding (whether or not vested).

(r) <u>Waivers</u>. The Company shall have secured waivers of any indemnification provisions in any agreements with disqualified individuals pursuant to Section 6.3(c).

7.3 <u>Additional Conditions to Obligations of the Company</u>. The obligations of the Company to effect the Merger shall be subject to the satisfaction at or prior to the Effective Time of the following additional conditions:

(a) <u>Representations and Warranties</u>. The representations and warranties of Parent and Acquisition Sub set forth in this Agreement that are qualified as to materiality shall be true and correct, and all other representations and warranties of Parent and Acquisition Sub set forth in this Agreement that are not so qualified shall be true and correct in all material respects, in each case as of the Agreement Date and the Closing Date (as qualified by the Parent Disclosure Schedule, if any) with the same effect as though made as of the Agreement Date and the Closing Date, except that the accuracy of representations and warranties that by their terms speak as of a specified date will be determined as of such date.

(b) <u>Agreements and Covenants</u>. Each of Parent and Acquisition Sub shall have performed or complied, in all material respects, with each obligation, agreement and covenant to be performed or complied with by it under this Agreement at or prior to the Effective Time.

(c) <u>Parent Officer's Certificate</u>. Parent shall have delivered to the Company a certificate of the President or Chief Executive Officer of Parent, dated as of the Closing Date, certifying that the conditions set forth in Sections 7.1(c), 7.3(a) and (b) have been satisfied.

(d) <u>Acquisition Sub Officer's Certificate</u>. Acquisition Sub shall have delivered to the Company a certificate of the President or Chief Executive Officer of Acquisition Sub, dated as of the Closing Date, certifying that the conditions set forth in Sections 7.3(a) and (b) have been satisfied.

ARTICLE VIII TERMINATION AND EXPENSES

8.1 <u>Termination</u>. This Agreement may be terminated and the Merger contemplated hereby may be abandoned at any time prior to the Effective Time, notwithstanding the Stockholder Approval:

(a) By mutual written consent duly authorized by the Company Board and the Parent;

(b) By either Parent or the Company if the Merger shall not have been consummated on or before the date (the "<u>Outside Date</u>") thirty (30) Business Days after the Agreement Date; <u>provided</u>, <u>however</u>, that if the Merger shall not have been consummated solely due to any Consent from any Governmental Authority necessary for the consummation of the Merger not having been received, then, unless Parent gives notice under Section 6.2(d) that it is abandoning its efforts to seek approval under applicable Law relating to any Antitrust Filings, such date shall be extended until such approvals are obtained, but in any event, not more than ninety (90) days after the date that would otherwise be the Outside Date; and <u>provided further</u>, that the right to terminate this Agreement under this Section 8.1(b) shall not be available to any party whose failure to fulfill any obligation under this Agreement has been the cause of, or resulted in, the failure of the Merger to have been consummated on or before such date;

(c) By either Parent or the Company, if (i) a Court or Governmental Authority shall have issued a Judgment or taken any other action, in each case, which has become final and non-appealable and which restrains, enjoins or otherwise prohibits or prevents the Merger; or (ii) there shall have been any Law enacted, promulgated or issued or deemed applicable to the Merger by any Governmental Authority that would make the consummation of the Merger illegal; or (iii) Parent gives notice under Section 6.2(d) that it is abandoning its efforts to seek approval under applicable Law relating to any Antitrust Filings;

(d) By Parent, if neither Parent nor Acquisition Sub is in material breach of any its obligations under this Agreement, and if the Company shall have breached in any material respect any of its representations or warranties or failed to perform in any material respect any of its covenants or other agreements contained in this Agreement, which breach or failure to perform would render unsatisfied any condition contained in Section 7.1 or 7.2, and (i) is incapable of being cured, or (ii) if capable of being cured is not cured within the earlier of ten (10) Business Days following notice of such breach and the Outside Date;

(e) By the Company, if it is not in material breach of any of its obligations under this Agreement, and if Parent or Acquisition Sub shall have breached in any material respect any of its representations or warranties or failed to perform in any material respect any of its covenants or other agreements contained in this Agreement, which breach or failure to perform would render unsatisfied any condition contained in Section 7.1 or 7.3, and (i) is incapable of being cured, or (ii) if capable of being cured is not cured within the earlier of ten (10) Business Days following notice of such breach and the Outside Date;

(f) By Parent if the Company shall breach (x) Section 5.8(b) (Stockholder Meeting) and such breach has not been cured within 48 hours of notice by Parent of such breach or (y) Section 5.12 (Acquisition Proposals);

(g) By Parent if there is a Company Material Adverse Change other than an Excluded Company Material Adverse Change; or

(h) By Parent if Corporate Stockholder Approval has not been obtained and delivered to Parent by the deadline set forth in Section 5.8 of this Agreement.

8.2 <u>Effect of Termination</u>. In the event of the termination of this Agreement pursuant to Section 8.1, this Agreement (other than as set forth in this Section 8.2 and the confidentiality provisions of Section 5.9, which shall survive such termination) will forthwith become void, and there will be no Liability on the part of Parent, Acquisition Sub, the Company or the Company Holders or any of their respective officers or directors to the other under this Agreement and all rights and obligations hereunder of any party hereto will cease, except that (i) no party shall be relieved of any Liability arising from any willful breach by such party of any provision of this Agreement and (ii) the rights of either party to compel specific performance by the other party of its respective obligations under this Agreement shall not be impaired. Notwithstanding the termination of this Agreement, the Loan Documents will remain in full force and effect in accordance with their terms.

8.3 <u>Expenses</u>. Notwithstanding anything to the contrary herein, all filing fees for Antitrust Filings under Antitrust Laws shall be paid by Parent or the Surviving Corporation. In the event the Merger is not consummated, all fees and expenses incurred in connection with the Merger and the other transactions contemplated by this Agreement, including all legal, accounting, tax and financial advisory, consulting, investment banking and all other fees and expenses of third Persons shall be the obligation of the party incurring such fees and expenses.

ARTICLE IX REPRESENTATIVE

9.1 Authorization of the Representative; Inability to Perform.

(a) Each Company Holder, by its approval of this Agreement and/or its execution and delivery of a Letter of Transmittal or an Option Cancellation Agreement, makes, constitutes and appoints the Representative such Person's true and lawful attorney-in-fact for and in such Person's name, place, and stead and for its use and benefit, to prepare, execute, certify, acknowledge, swear to, file, deliver, or record any and all agreements, instruments or other documents, and to take any and all actions, that are within the scope and authority of the Representative provided for in this Section 9.1. The grant of authority provided for in this Section 9.1 is coupled with an interest and is being granted, in part, as an inducement to the parties hereto to enter into this Agreement and shall be irrevocable and survive the death, incompetency, bankruptcy or liquidation of any Company Holder and shall be binding on any successor thereto and shall survive the payment of all or any portion of the Escrow Fund.

(b) By virtue of the adoption of this Agreement and the approval of the transactions contemplated hereby, including the Merger, by the Company Holders, and/or by virtue of the execution and delivery of a Letter of Transmittal or an Option Cancellation Agreement by the Company Holders, each Company Holder shall be deemed to have agreed to appoint the Representative as its agent and attorney-in-fact for and on behalf of the Company Holders in connection with, and to facilitate the consummation of the transactions contemplated by, this Agreement and any other Related Agreements, and in connection with the activities to be performed on behalf of the Company Holders under this Agreement, for the purposes and with the powers and authority hereinafter set forth in this Article IX, which shall include the full power and authority:

(i) to take such actions and to execute and deliver such amendments, modifications, waivers and consents in connection with this Agreement and the Related Agreements and the consummation of the transactions contemplated hereby and thereby as the Representative, in his reasonable discretion, may deem necessary or desirable to give effect to the intentions of this Agreement and the Related Agreements;

(ii) as the agent of the Company Holders, to enforce and protect the rights and interests of the Company Holders arising out of or under or in any manner relating to this Agreement and each other Related Agreement and, in connection therewith, to: (A) resolve all questions, disputes, conflicts and controversies concerning (1) any adjustments to the Merger Consideration following the Effective Time pursuant to Section 2.17 (Working Capital Adjustment), (2) the payment (or non-payment) of any Escrow Consideration and (3) indemnification claims pursuant to Article X; (B) employ such agents, consultants and professionals, to delegate authority to his agents, to take such actions and to execute such documents on behalf of the Company Holders in connection with Article II and Article X as the Representative, in his reasonable discretion, deems to be in the best interest of the Company Holders; (C) assert or institute any Claim, Action or investigation; (D) investigate, defend, contest or litigate any Action initiated by Parent or the Surviving Corporation, or any other Person, against the Representative, and receive process on behalf of any or all Company Holders in any such Claim, Action or investigation and compromise or settle on such terms as the Representative shall determine to be appropriate, give receipts, releases and discharges on behalf of all of the Company Holders with respect to any such Claim, Action or investigation; (E) file

any proofs, debts, claims and petitions as the Representative may deem advisable or necessary; (F) settle or compromise any Claims asserted under Article II or Article X; (G) assume, on behalf of all of the Company Holders, the defense of any Claim that is the basis of any claim asserted under Article II or Article X; and (H) file and prosecute appeals from any decision, Judgment or award rendered in any of the foregoing Claims, Actions or investigations, it being understood that the Representative shall not have any obligation to take any such Actions, and shall not have liability for any failure to take such any action;

(iii) to enforce payment of any other amounts payable to the Company Holders, in each case on behalf of the Company Holders, in the name of the Representative;

(iv) to waive or refrain from enforcing any right of the Company Holders or any of them arising out of or under or in any manner relating to this Agreement or any other Related Agreement; and

(v) to make, execute, acknowledge and deliver all such other agreements, guarantees, orders, receipts, endorsements, notices, requests, instructions, certificates, stock powers, letters and other writings, and, in general, to do any and all things and to take any and all action that the Representative, in his sole and absolute direction, may consider necessary or proper or convenient in connection with or to carry out the activities described in paragraphs (i) through (iv) above and the transactions contemplated by this Agreement and the other Related Agreements.

(c) The Company Holders who immediately prior to the Effective Time held at least a majority of the outstanding Company Shares (on a fullydiluted basis) held by all Company Holders (the "<u>Majority Effective Time Holders</u>") may, in their sole discretion, upon 30 days' prior written notice to Parent, remove the Representative from such position. In the event the Representative becomes unable to perform his responsibilities hereunder or resigns or is removed from such position, the Majority Effective Time Holders shall select another representative to fill the vacancy of the Representative, and such substituted representative shall be deemed to be a Representative for all purposes of this Agreement and the documents delivered pursuant hereto.

(d) Parent and the Surviving Corporation shall be entitled to rely exclusively upon the communications of the Representative relating to the foregoing as the communications of the Company Holders. None of Parent or the Surviving Corporation shall be held liable or accountable in any manner for any act or omission of the Representative in such capacity.

9.2 <u>Representative Reimbursement Set-Aside</u>. The Representative Reimbursement Set-Aside shall be held by the Representative in a segregated interest bearing account for the benefit of the Company Holders for reimbursements payable to the Representative under this Section 9.2. Any portion of the Representative Reimbursement Set-Aside originally deposited with the Representative at the Closing that has not been utilized by the Representative pursuant to the terms of this Agreement on or prior to the Escrow Termination Date or, if later, the date on which all indemnification claims of Indemnified Persons outstanding at the Escrow Termination Date have been discharged in full, shall be distributed by the Representative in accordance with Sections 2.10, 2.13 and 2.14.

9.3 Compensation; Exculpation.

(a) Upon approval of a majority in interest of the Company Holders who hold Preferred Stock (calculated on an as-converted basis) prior to the Effective Time, or upon approval of a majority in interest of the Company Holders who are beneficiaries in the Escrow Fund after the Effective Time, the Representative may be paid a fee, commission or other compensation by the Company Holders for the performance of service hereunder on such terms and conditions as may be agreed by the Representative and such Company Holders.

(b) In dealing with this Agreement and any instruments, agreements or documents relating thereto, and in exercising or failing to exercise all or any of the powers conferred upon the Representative hereunder or thereunder (i) the Representative shall not assume any, and shall incur no, responsibility whatsoever while acting in good faith to any Company Holder by reason of any error in judgment or other act or omission performed or omitted hereunder or in connection with this Agreement or any other Related Agreement, unless by the Representative's gross negligence or willful and intentional misconduct, and (ii) the Representative shall be entitled to rely on the advice of counsel, public accountants or other independent experts experienced in the matter at issue, and any error in judgment or other act or omission of the Representative pursuant to such advice shall in no event subject the Representative to liability to any Company Holder unless by the Representative's gross negligence or willful and intentional misconduct. Except as set forth in the previous sentence, notwithstanding anything to the contrary contained in this Agreement, the Representative, in his role as Representative, shall have no liability whatsoever to Parent or the Surviving Corporation or any other Person.

(c) All of the immunities and powers granted to the Representative under this Agreement shall survive until the termination of this Agreement.

(d) The adoption of this Agreement and the Stockholder Approval of the transactions contemplated hereby, including the Merger, shall also be deemed to constitute approval of all arrangements relating to the transactions contemplated hereby and to the provisions hereof binding upon the holders of Voting Holders, including this Section 9.3.

ARTICLE X INDEMNIFICATION

10.1 General Indemnification.

(a) Subject to the limitations set forth in this Article X, each Company Holder shall, from and after the Closing Date, severally and not jointly (based on the proportion of the Merger Consideration that has been received by such Company Holder under this Agreement bears to the total Merger Consideration that has been received by all Company Holders under this Agreement), indemnify, defend and hold harmless from and after the Closing Date, Parent, Acquisition Sub and the Surviving Corporation and each of their respective present and former officers, directors, employees, agents, representatives, assigns, successors and Affiliates (each, an "Indemnified Party" and collectively the "Indemnified Parties"), against any and all Liabilities, losses, claims, damages (including solely with

respect to Third Party Claims and not with respect to any other claims, punitive, special, consequential, "lost profits", diminution in value, exemplary or similar damages claimed by such third Person), causes of action, lawsuits, administrative proceedings (including informal proceedings), investigations, audits, demands, assessments, adjustments, Judgments, settlement payments, deficiencies, Taxes, penalties, fines and costs and expenses (including reasonable attorneys' fees and disbursements of every kind, nature and description) (collectively, "Damages"), related to or arising, directly or indirectly, out of or caused by any of the following:

(i) the failure of any representation or warranty of the Company set forth herein to be true and correct in all respects as of the Agreement Date and as of the Closing Date (in each case as qualified by the Company Disclosure Schedules) or the failure of any representation or warranty of the Company set forth in any certificate, document or other instrument delivered by or on behalf of the Company pursuant to Section 7.2 to be true and correct in all respects, disregarding for purposes of this Section 10.1(a)(i) all knowledge, materiality and similar qualifications contained therein for purposes of calculating Damages but not for purposes of determining whether an inaccuracy or breach has occurred;

(ii) any failure or refusal by the Company (prior to the Effective Time) or the Representative (after the Effective Time) to fully satisfy, comply with or perform any covenant or agreement contained in this Agreement or in any certificate, document or other instrument delivered pursuant to Section 7.2 of this Agreement required to be performed by the Company (prior to the Effective Time) or the Representative (after the Effective Time);

(iii) any Dissenting Share Payments;

(iv) the failure of any item set forth in the Allocation Certificate to be accurate, true and correct in all respects as of the Closing (or if updated by the Representative following Closing as required by Section 2.13(b) or Section 2.14(b), as of the time of such update);

(v) any Taxes of the Company attributable to any Pre-Closing Tax Period (to the extent such Taxes exceed the aggregate amount accrued in the calculation of Working Capital for Taxes attributable to any Pre-Closing Tax Period);

(vi) regardless of any disclosure on the Company Disclosure Schedules or any action taken with respect to a shareholder vote in accordance with Section 6.3(c) of this Agreement, the treatment as an "excess parachute payment" (within the meaning of Code Section 280G(b)) of any payment made by the Company on or prior to the Closing Date or otherwise required to be paid pursuant to a written or oral contract or agreement or otherwise by Parent, the Company, the Surviving Corporation or their respective Affiliates before, on or after the Closing Date, including, without limitation, any portion of the Merger Consideration;

(vii) the failure of the Company to have obtained, prior to the Effective Time, the exclusive right to develop, manufacture, perform, sell and commercialize any Commercial Product being developed, planned, manufactured, sold, performed or commercialized by the Company or any of its Subsidiaries as of the Effective Time, subject to any limitations contained in any license agreement listed in the Company Disclosure Schedules, free from any claim, demand, liability or obligation arising from or in connection with the assertion of infringement or misappropriation of Intellectual Property rights by any third party against the Indemnified Parties; or

(viii) any Change of Control Payments, including any payments made following the Closing, that were not included in the Closing Payment Reduction Amount.

(b) Subject to Section 10.2, the Escrow Funds shall be available to reimburse the Indemnified Parties for any Damages for which they are entitled to be indemnified pursuant to this Section 10.1. Notwithstanding anything in this Agreement to the contrary, only Parent shall be entitled to directly enforce any indemnification rights of any Indemnified Parties that they may be entitled to hereunder.

10.2 Limitations of Indemnification Obligations.

(a) <u>Deductible</u>. There shall be no liability for indemnification under Section 10.1(a)(i) unless the aggregate amount of Damages thereunder exceeds the Deductible, at which time the Indemnified Parties will be entitled to indemnification for only the amount of such Damages that exceed the Deductible; <u>provided</u>, <u>however</u>, that the Deductible shall not apply to:

(i) any Claims for actual fraud with intent to deceive or intentional misrepresentation;

(ii) any Claims arising out of the misrepresentation, breach or inaccuracy of any representation or warranty made by the Company in Section 3.1 (Organization, Good Standing, Qualification and Power), Section 3.2 (Subsidiaries), Section 3.4 (Capitalization), Section 3.5 (Authorization; Binding Obligation) or Section 3.24 (Brokers) (collectively, the "<u>Fundamental Representations</u>"); or

(iii) any Claims arising out of the misrepresentation, breach or inaccuracy of any representation or warranty made by the Company in Section 3.15 (Employment and Benefit Matters); Section 3.21 (Environmental Matters) or Section 3.22 (Tax Liabilities) (collectively, the "<u>Special</u> <u>Representations</u>" and, together with the Fundamental Representations, the "<u>Specified Representations</u>").

(b) Sole and Exclusive Remedy; Recourse Against Escrow Funds and in Excess of Escrow Funds. The rights of Indemnified Parties to assert Claims against the Escrow Fund held in escrow pursuant to the Escrow Agreement shall, from and after the Closing, be the sole and exclusive remedies for money damages of Indemnified Parties for any Claims pursuant to Article X or otherwise under this Agreement and no Company Holder shall have any liability in excess of its portion of the Escrow Consideration, except as follows:

(i) for causes of action by the Indemnified Parties seeking equitable relief pursuant to Section 11.7 hereof;

(ii) in the case of Claims against a Person for such Person's own actual fraud with intent to deceive or intentional misrepresentation;

(iii) in the case of Claims arising out of the misrepresentation, breach or inaccuracy of any Special Representations, the maximum aggregate Damages payable by each Company Holder shall be limited to twenty-five percent (25%) of the Merger Consideration received by such Company Holder under this Agreement; or

(iv) in the case of Claims arising out of the misrepresentation, breach or inaccuracy of any Fundamental Representations or any representation, warranty or covenant contained in the Letter of Transmittal and/or Option Cancellation Agreement delivered by such Company Holder pursuant to this Agreement, the maximum aggregate Damages payable by each Company Holder shall be limited to the Merger Consideration received by such Company Holder under this Agreement.

10.3 Survival and Expiration of Representations and Warranties.

(a) The representations and warranties of the Company in this Agreement or in any other agreement, exhibit, schedule, certificate, instrument or other writing delivered by or on behalf of the Company pursuant to this Agreement shall survive the Closing until the Escrow Termination Date; provided, <u>however</u>, that each of the Specified Representations shall survive the Closing until sixty (60) days following the expiration of the applicable statute of limitations (including any extension thereof) with respect thereto. The date upon which any representation or warranty shall terminate is referred to herein as the "<u>Survival Date</u>." Any Claim for indemnification in accordance with this Article X must be made prior to the end of the applicable Survival Date. Upon expiration of such Survival Date, Indemnified Parties shall have no right to indemnification unless an Indemnified Party delivers to the Representative an Indemnification Notice prior to the applicable Survival Date. It is the express intent of the parties that, if the applicable survival period for an item as contemplated by this Section 10.3(a) is shorter than the statute of limitations that would otherwise have been applicable to such item, then, by contract, the applicable statute of limitations with respect to such item shall be reduced to the shortened survival period contemplated hereby. The parties further acknowledge that the time periods set forth in this Section 10.3(a) for the assertion of claims under this Agreement are the result of arms-length negotiation among the parties and that they intend for the time periods to be enforced as agreed by the parties.

(b) All covenants of the Company in this Agreement, including the obligations set forth in Article X, shall survive the Closing, continue in effect and expire in accordance with their respective terms.

(c) Notwithstanding anything to the contrary herein, any Claim alleging any misrepresentation, breach or inaccuracy of any representation or warranty made prior to the Survival Date with respect to the applicable representation or warranty shall survive the expiration of such representation or warranty until final resolution of such Claim and payment of any Damages associated therewith, but only for the purpose of the resolution of such Claim.

10.4 Indemnification Procedures. All claims for indemnification under this Article X ("Claims") shall be asserted and resolved as follows:

(a) <u>Claims Other than Third Party Claims</u>. In the event any Indemnified Party should have a Claim under this Article X that does not involve a Third Party Claim being asserted against or sought to be collected from such Indemnified Party, the Indemnified Party shall deliver written notice of such Claim with reasonable promptness and stating the nature, basis and amount of the Claim in reasonable detail to the Representative. The failure by any Indemnified Party to so notify the Representative shall not relieve any liability under this Article X, except to the extent that the ability to defend such Claim or demand shall have been materially prejudiced as a result of such failure. If the Representative disputes the liability with respect to such Claim, the Representative and the Indemnified Party shall proceed in good faith to negotiate a resolution of such dispute and, if not resolved through negotiations, such dispute shall be resolved by litigation in an appropriate court of competent jurisdiction.

(b) <u>Third Party Claims</u>. Promptly after receipt by any Indemnified Party of notice of the commencement of any action by a third party in respect of which, if successful, the Indemnified Party would be entitled to indemnification under Article X (a "<u>Third Party Claim</u>"), the Indemnified Party shall notify the Representative in writing (an "Indemnification Notice"), but any failure to so notify the Representative shall not relieve the Indemnifying Party from any liability that it may have to the Indemnified Party other than to the extent the Indemnifying Party is materially prejudiced thereby. Such notification shall include a description in reasonable detail (to the extent known by the Indemnified Party) of the facts constituting the basis for such Third Party Claim and the amount of the Damages claimed.

(c) If the Representative objects to the indemnification of an Indemnified Party in respect of any claim or claims specified in any Indemnification Notice, the Representative shall deliver a written notice to such effect to the Indemnified Party within thirty (30) days after receipt by the Representative of such Indemnification Notice. Thereafter, the Representative and the Indemnified Party shall attempt in good faith to agree upon the rights of the respective parties for a period of not less than sixty (60) days after receipt by the Indemnified Party of such written objection with respect to each of such claims to which the Representative has objected. If the Indemnified Party and the Representative agree with respect to any of such claims, the Indemnified Party and the Representative shall promptly prepare and sign a memorandum setting forth such agreement and, if applicable, an instruction to the Escrow Agent. Should the Indemnified Party shall be entitled to pursue its available remedies for resolving its claim for indemnification. Notwithstanding the foregoing, in the event that the Indemnified Party seeks recovery against the Escrow Fund, the time periods set forth herein shall run concurrently with and be without duplication of the time periods set forth in the Escrow Agreement.

(d) <u>Indemnified Party Defense</u>; <u>Settlement</u>. The Indemnified Party shall have the right in its sole discretion to conduct the defense of a Third-Party Claim; <u>provided</u>, <u>however</u>, that any settlement of any such Third-Party Claim shall not be effected without the prior written consent of the Representative, which consent shall not be unreasonably withheld, conditioned or

delayed but if such consent is unreasonably withheld, conditioned or delayed then such consent shall not be required. If any such action or claim is so settled or if there be a final judgment for the plaintiff in any such action, the Indemnified Party shall be entitled to indemnification for the amount of any Damage relating thereto. The Indemnified Party will keep the Representative reasonably apprised of the progress and status of, and consult with the Representative regarding, the defense of any Third Party Claim and take into account good faith reasonable suggestions of the either the Representative or its counsel with respect to the defense of the Third Party Claim.

(e) <u>Representative Defense; Settlement</u>. In the event the Indemnified Party elects not to defend the Third Party Claim, the Representative may defend such claim at the Company Holders' sole cost and expense. In such event, the Representative shall have no right to settle, adjust or compromise any Third-Party Claim without the express written consent of the Indemnified Party, which consent shall not be unreasonably withheld, conditioned or delayed.

(f) <u>Advisement</u>. The party controlling the defense of the Third Party Claim shall keep the other party advised of the status of such Third Party Claim and the defense thereof and shall consider recommendations made by the other party with respect thereto.

(g) <u>Agreed Claims</u>. Claims for Damages specified in any Indemnification Notice to which the Representative did not object in writing within thirty (30) days of receipt of such Indemnification Notice, claims for Damages covered by a memorandum of agreement of the nature described in Section 10.5(c) and claims for Damages the validity and amount of which have been the subject of resolution by arbitration or of a final non-appealable judicial determination are hereinafter referred to, collectively, as "<u>Agreed Claims</u>." The Indemnified Party shall be entitled to payment for any Agreed Claim within ten (10) Business Days of the determination of the amount of any such Agreed Claims.

10.5 Insurance. The amount of Damages recoverable by an Indemnified Party under this Article X with respect to a Claim shall be reduced by the amount of any payment actually received by such Indemnified Party (or an Affiliate thereof), with respect to the Damages to which such Claim relates, from an insurance carrier. If an Indemnified Party (or an Affiliate) receives any insurance payment in connection with any Claim for which it has already received an indemnification payment from the Indemnifying Party, it shall pay to the Indemnifying Party, within thirty (30) days of receiving such insurance payment, an amount equal to the excess of: (i) the amount previously received by the Indemnified Party under this Article IX with respect to such Claim plus the amount of the insurance payments received; minus (ii) the sum of any increased insurance premiums payable by the Indemnified Party as a result of such insurance claim plus the amount of Damages with respect to such Claim which the Indemnified Party has become entitled to receive under this Article IX.

10.6 <u>No Subrogation</u>. Following the Closing, no Company Holder shall have any right of indemnification, contribution or subrogation against Parent, the Company, the Surviving Corporation, or any of their Subsidiaries, with respect to any indemnification claim made against the Company or any Company Holder under Article X.

10.7 <u>Merger Consideration Adjustment</u>. The Company, Parent, and the Representative agree to treat each indemnification payment pursuant to this Article X as an adjustment to the Merger Consideration paid to the Company Holders under this Agreement for all Tax purposes and shall take no position contrary thereto unless required to do so by applicable Tax Law pursuant to a determination as defined in Section 1313(a) of the Code.

ARTICLE XI MISCELLANEOUS

11.1 Entire Agreement. This Agreement, together with its schedules and exhibits, the Prior Merger Agreement, together with its schedules and exhibits (unless and until superseded pursuant to the preamble of this Agreement), the Letter Amendment (unless and until superseded pursuant to its terms), the Confidentiality Agreement (as amended by Section 5.9 of this Agreement), the Loan Documents, and all other ancillary agreements, documents and instruments to be delivered in connection herewith, contain the entire understanding of the parties with respect to the subject matter hereof and supersede all prior agreements, either oral or written. Each party to this Agreement acknowledges that no representations, inducements, promises or agreements, orally or otherwise, have been made by any other party, or by anyone acting on behalf of any party, that are not embodied herein, and that no other agreement, statement, or promise with respect to the subject matter of this Agreement that is not contained in this Agreement shall be valid or binding other than those contained in the documents set forth in the first sentence of this Section 11.1.

11.2 Amendment and Waiver.

(a) No failure or delay on the part of any party in exercising any right, power or remedy hereunder shall operate as a waiver thereof, nor shall any single or partial exercise of any such right, power or remedy preclude any other or further exercise thereof or the exercise of any other right, power or remedy. The remedies provided for herein are, subject to Section 10.5, cumulative and are not exclusive of any remedies that may be available to any party at Law, in equity or otherwise.

(b) Except as otherwise specifically set forth in this Agreement, this Agreement may be amended by the parties at any time, whether before or after the Stockholder Approval has been obtained; <u>provided</u>, <u>however</u>, that, after the Stockholder Approval has been obtained, there shall be made no amendment that by Law requires further approval by stockholders of either party, without the further approval of such stockholders. This Agreement may not be amended except by an instrument in writing signed on behalf of each of the parties.

(c) Except as otherwise specifically set forth in this Agreement, any amendment, supplement or modification of or to any provision of this Agreement and any waiver of any provision of this Agreement shall be effective (i) only if it is made or given in writing and signed by Parent and (A) prior to the Closing, the Company and (B) after the Closing, the Representative or, in the case of a waiver, by the party granting the waiver and (ii) only in the specific instance and for the specific purpose for which made or given.

11.3 <u>No Third-Party Beneficiaries</u>. Except as otherwise provided in this Agreement including, without limitation, the present and former directors and officers of the Company who may benefit from the provisions of Section 6.4 and the Indemnified Parties who may benefit

from the provisions of Article X of this Agreement among others, this Agreement is for the sole benefit of the parties and their permitted successors and assigns and nothing herein expressed or implied shall give or be construed to give to any Person, other than the Parties and such successors and assigns, any legal or equitable rights hereunder. It is expressly acknowledged that Section 6.6 shall not confer upon any Employee any right, benefit or remedy of any nature whatsoever under or by reason of this Agreement, including in respect of the matters set forth in Section 6.6 or elsewhere in this Agreement.

11.4 <u>Assignment</u>. No party hereto shall assign or otherwise transfer this Agreement or any of its rights hereunder, or delegate any of its obligations hereunder, without the prior written consent of the other parties hereto; <u>provided</u>, <u>however</u>, Parent and Acquisition Sub shall have the right, without obtaining the consent of the Company or the Representative, to assign or otherwise transfer this Agreement and its rights and obligations hereunder to any Person who acquires all or substantially all the outstanding voting stock or all or substantially all of the assets and business of Parent. Subject to the foregoing, this Agreement and the rights and obligations set forth herein shall inure to the benefit of, and be binding upon the parties hereto, and each of their respective successors, heirs and permitted assigns.

11.5 <u>Effect of Waivers</u>. No waiver by any party, whether express or implied, of its rights under any provision of this Agreement shall constitute a waiver of the party's rights under such provisions at any other time or a waiver of the party's rights under any other provision of this Agreement. No failure by any party to take any action against any breach of this Agreement or default by another party shall constitute a waiver of the former party's right to enforce any provision of this Agreement or to take action against such breach or default or any subsequent breach or default by the other party. To be effective any waiver must be in writing and signed by the waiving party.

11.6 <u>Governing Law; Venue</u>. This Agreement shall be governed by the Laws of the State of Delaware, without giving effect to any choice of Law or conflict of Law provision or rule that would cause application of the Laws of any jurisdiction other than the State of Delaware. Each of the parties to this Agreement irrevocably submits to the non-exclusive jurisdiction of the Courts of the State of Delaware, for the purpose of any Action arising out of or relating to this Agreement. Each of the parties to this Agreement consents to service of process by delivery pursuant to Section 11.10 hereof and agrees that a final Judgment in any Action shall be conclusive and may be enforced in other jurisdictions by suit on the Judgment or in any other manner provided by Law. Each of the parties hereto waives ay right to trial by jury with respect to any action related to or arising out of this Agreement or any transaction contemplated hereby.

11.7 Equitable Relief. The rights and remedies of the parties hereto shall be cumulative. The transactions contemplated by this Agreement are unique transactions and any failure on the part of any party to complete the transactions contemplated by this Agreement on the terms of this Agreement will not be fully compensable in damages and the breach or threatened breach of the provisions of this Agreement would cause the other parties hereto irreparable harm. Accordingly, in addition to and not in limitation of any other remedies available to the parties hereto for a breach or threatened breach of this Agreement, the parties shall be entitled to seek specific performance of this Agreement and seek an injunction restraining any such party from such breach or threatened breach, specific performance and/or other equitable relief to prevent the breach of such obligations.

11.8 Interpretation. The schedules and exhibits attached hereto are an integral part of this Agreement. All schedules and exhibits attached to this Agreement are incorporated herein by this reference and all references herein to this "Agreement" shall mean this Agreement together with all such schedules and exhibits. When a reference is made in this Agreement to Sections, subsections, schedules or exhibits, such reference shall be to a Section, subsection, schedule or exhibit to this Agreement unless otherwise indicated. The words "include," "includes" and "including" when used herein shall be deemed in each case to be followed by the words "without limitation." The word "herein" and similar references mean, except where a specific Section or Article reference is expressly indicated, the entire Agreement rather than any specific Section or Article. The table of contents and the headings contained in this Agreement are for reference purposes only and shall not affect in any way the meaning or interpretation of this Agreement. As used herein, all pronouns shall include the masculine, feminine, neuter, singular and plural thereof whenever the context and facts require such construction. Headings have been inserted on the sections of the Company Disclosure Schedule for convenience of reference only and shall to no extent have the effect of amending or changing the express description of the sections of the Company Disclosure Schedule as set forth in this Agreement.

11.9 <u>Severability</u>. If any term or other provision of this Agreement is invalid, illegal or incapable of being enforced by any rule of Law, or public policy, all other conditions and provisions of this Agreement shall nevertheless remain in full force and effect.

11.10 <u>Notices</u>. All notices or other communications which are required or permitted hereunder shall be in writing and sufficient if delivered personally or sent by nationally-recognized overnight courier providing evidence of delivery or by registered or certified mail, postage prepaid, return receipt requested, at the following addresses or sent by facsimile, with confirmation received, to the facsimile number specified below:

If to Parent or Acquisition Sub:

Myriad Genetics, Inc. 320 Wakara Way Salt Lake City, Utah 84108 Attention: Chief Executive Officer General Counsel Facsimile: 801-584-3640 Mintz Levin Cohn Ferris Glovski

With copies to:

Mintz, Levin, Cohn, Ferris, Glovsky and Popeo, P.C. One Financial Center Boston, MA 02111 Attention: Jonathan R. Kravetz, Esq. and Scott A. Samuels, Esq. Facsimile: 617-542-2241

Crescendo Bioscience, Inc. 341 Oyster Point Blvd. South San Francisco, CA 94080 Attention: Chief Executive Officer Facsimile: 650-871-7155
Cooley LLP 3175 Hanover Street Palo Alto, CA 94304-1130 Attention: Jennifer F. DiNucci, Esq. Facsimile: 650-849-7400
MDV IX, L.P. 3000 Sand Hill Road, 3-290 Menlo Park, CA 94025 Attention: General Partner Facsimile: 650-854-7365
Cooley LLP 3175 Hanover Street Palo Alto, CA 94304-1130 Attention: Jennifer F. DiNucci, Esq. Facsimile: 650-849-7400

or to such other address as the party to whom notice is to be given may have furnished to the other party in writing in accordance herewith. All such notices or communications shall be deemed to be received (a) in the case of personal delivery, on the date of such delivery, (b) in the case of nationally-recognized overnight courier, on the date of delivery as indicated by the evidence of delivery furnished by the courier when sent, (c) in the case of mailing, on the third Business Day following the date on which the piece of mail containing such communication was posted, and (d) in the case of facsimile, on the date of delivery (if confirmed by machine generated receipt).

11.11 <u>Representation by Counsel</u>. Each party hereto acknowledges that it has been advised by legal and any other counsel retained by such party in its sole discretion. Each party acknowledges that such party has had a full opportunity to review this Agreement and all related exhibits, schedules and ancillary agreements and to negotiate any and all such documents in its sole discretion, without any undue influence by any other party hereto or any third party.

11.12 <u>Construction</u>. The parties have participated jointly in the negotiations and drafting of this Agreement and in the event of any ambiguity or question of intent or interpretation, no presumption or burden of proof shall arise favoring or disfavoring any party by virtue of the authorship of any of the provisions of this Agreement.

11.13 <u>Calculation of Merger Consideration</u>. Notwithstanding anything contained in this Agreement to the contrary, except as otherwise expressly provided in this Agreement, the parties hereto covenant and agree that no amount shall be (or is intended to be) included, in whole or in part (either as an increase or a reduction), more than once in the calculation of (including any component of) the Merger Consideration or any component thereof or calculation relating

thereto, or any other calculated amount pursuant to this Agreement if the effect of such additional inclusion (either as an increase or a reduction) would be to cause such amount to be over- or under-counted for purposes of the transactions contemplated by this Agreement.

11.14 <u>Materiality</u>. The specification of any dollar amount in the representations and warranties or otherwise in this Agreement or in the Company Disclosure Schedules is not intended and shall not be deemed to be an admission or acknowledgment of the materiality of such amounts or items, nor shall the same be used in any dispute or controversy between the parties to determine whether any obligation, item or matter (whether or not described herein or included in any schedule) is or is not material for purposes of this Agreement.

11.15 <u>Disclosure Schedules</u>. The inclusion of any information in the Company Disclosure Schedules shall not be deemed an admission or acknowledgment, in and of itself and solely by virtue of the inclusion of such information in the Company Disclosure Schedules, that such information is required to be listed in the Company Disclosure Schedules or that such items are material to the Company, Parent or Acquisition Sub, as the case may be. The headings, if any, of the individual sections of each of the Company Disclosure Schedules are inserted for convenience only and shall not be deemed to constitute a part thereof or a part of this Agreement. The Company Disclosure Schedules are arranged in sections corresponding to those contained in Article III merely for convenience, and the disclosure of an item in one section of the Company Disclosure Schedules as an exception to a particular representation or warranty shall be deemed adequately disclosed as an exception to certain other representations or warranties to the extent that it is reasonably apparent from a reading of such disclosure item that it would also apply to such other representations or warranties, notwithstanding the presence or absence of an appropriate section of the Company Disclosure schedules or warranties or an appropriate cross reference thereto.

11.16 <u>Headings</u>. The section headings contained in this Agreement are inserted for convenience only and shall not affect in any way the meaning or interpretation of this Agreement.

11.17 <u>Counterparts</u>. This Agreement may be executed in two or more counterparts, any one of which need not contain the signatures of all parties, but all of which counterparts when taken together will constitute one and the same agreement. Facsimile signatures shall constitute original signatures for all purposes of this Agreement.

11.18 <u>Certain Waiver</u>. EXCEPT AS SET FORTH IN THE DEFINITION OF "DAMAGES", EACH PARTY, TO THE FULLEST EXTENT PERMITTED BY LAW, IRREVOCABLY WAIVES ANY RIGHTS THAT IT MAY HAVE TO PUNITIVE, SPECIAL, EXEMPLARY, CONSEQUENTIAL, LOST PROFITS, DIMINUTION IN VALUE OR SIMILAR DAMAGES BASED UPON, OR ARISING OUT OF, THIS AGREEMENT OR ANY COURSE OF CONDUCT, COURSE OF DEALING, STATEMENTS OR ACTIONS OF ANY OF THEM RELATING THERETO.

11.19 <u>Conflict Waiver</u>. Each of the parties acknowledges and agrees, on its own behalf and on behalf of its directors, members, partners, officers, employees, and Subsidiaries that the Company is the client of Cooley LLP (the "<u>Firm</u>"), and not any of its individual Company Holders. After the Closing, it is possible that the Firm will represent the Company Holders, the Representative and their respective Affiliates (individually and collectively, the "<u>Seller Group</u>") in connection with (a) the transactions contemplated herein, in the Escrow Agreement, or (b) the Escrow Fund and any claims made thereunder pursuant to this Agreement or the Escrow Agreement. Parent and the Company hereby agree that the Firm (or any successor) may represent the Seller Group in the future in connection with (i) matters that may arise under this Agreement or the Escrow Agreement or (ii) the administration of the Escrow Fund and any claims that may be made thereunder pursuant to this Agreement or the Escrow Agreement. The Firm (or any successor) may serve as counsel to all or a portion of the Seller Group or any director, member, partner, officer, employee, representative, or Affiliate of the Seller Group in connection with any litigation, claim or obligation arising out of or relating to this Agreement or the Escrow Agreement or the transactions contemplated by this Agreement or the Escrow Agreement. Each of the parties consents thereto, and waives any conflict of interest arising therefrom, and each such party shall cause any Subsidiary thereof to consent to waive any conflict of interest arising from such representation. Each of the parties acknowledges that such consent and waiver is voluntary, that it has been carefully considered, and that the parties have consulted with counsel or have been advised they should do so in this connection.

[Remainder of Page Intentionally Left Blank. Signature Page Follows.]

IN WITNESS WHEREOF, the parties hereto have executed this Amended and Restated Agreement and Plan of Merger by their duly authorized representatives as of the Agreement Date.

PARENT:

MYRIAD GENETICS, INC.

By: /s/ Peter D. Meldrum

Peter D. Meldrum President and CEO

ACQUISITION SUB:

MYRIAD CRESCENDO, INC.

By: /s/ Mark C. Capone

Mark C. Capone President

COMPANY:

CRESCENDO BIOSCIENCE, INC.

By: /s/ William A. Hagstrom

William A. Hagstrom President and CEO

<u>REPRESENTATIVE</u>:

MDV IX, L.P.

By: /s/ Josh Green Josh Green

General Partner

[SIGNATURE PAGE TO AMENDED AND RESTATED AGREEMENT AND PLAN OF MERGER]

Exhibit 99.1



Media Contact:

News Release

Investor Contact:

Scott Gleason (801) 584-1143 <u>sgleason@myriad.com</u>

Ron Rogers (801) 584-3065 <u>rrogers@myriad.com</u>

Myriad Genetics to Acquire Crescendo Bioscience®

Diversifies Myriad's Product Portfolio into High Growth Autoimmune Market

SALT LAKE CITY, Utah, February 4, 2014 – Myriad Genetics, Inc. (NASDAQ: MYGN) announced today that it has entered into a definitive agreement to acquire Crescendo Bioscience, Inc., a global leader in autoimmune diagnostics, for \$270 million in cash which will be reduced by \$25 million for the repayment of a loan made to Crescendo and customary adjustments in accordance with the acquisition agreement.

The transaction is expected to close before the end of Myriad's fiscal year 2014, pending satisfactory completion of customary closing items and regulatory approval. A discussion of the acquisition will be provided during Myriad's second fiscal quarter earnings call later today at 4:30 pm EST.

"Crescendo Bioscience fits well into our diagnostic portfolio that is focused on saving lives and improving the quality of life of patients across major diseases," said Peter D. Meldrum, president and CEO of Myriad Genetics. "Crescendo has pioneered protein-based diagnostics for monitoring disease activity in patients with rheumatoid arthritis and this acquisition diversifies our business into a new high growth, multibillion dollar market opportunity. We are pleased to welcome the 130 employees of Crescendo to the Myriad team and look forward to the impact of this innovative and dedicated group on our organization."

"Crescendo has built a strong and growing position in the rheumatoid arthritis diagnostic market, and I believe Myriad will enable us to move to the next level in terms of scale and growth," said William Hagstrom, president and CEO of Crescendo Bioscience. "We envision multiple opportunities over the next several years where, as a combined company, we can expand our presence into international markets and provide new innovative products that help improve the lives of patients suffering from autoimmune diseases."

Crescendo was founded in July 2002 and is a leading molecular diagnostic laboratory in the area of inflammatory and autoimmune diseases. Their core product, Vectra[®] DA, is a quantitative, protein-based test to routinely assess rheumatoid arthritis disease activity and provide rheumatologists with expanded

clinical insights to more effectively treat their patients. In the United States, more than 1.5 million individuals suffer from rheumatoid arthritis (RA) resulting in over \$11 billion in healthcare costs. Crescendo is also building a comprehensive understanding of the biology of other autoimmune diseases and has a pipeline of products under development for a wide range of diseases and conditions managed by rheumatologists. Crescendo additionally has created novel software products for physicians, including VectraViewTM which allows a comprehensive overview of all RA patients' level of disease activity and trends and MyRA[®] which is a patient tracking and communications tool.

In 2013, Crescendo was ranked the 2nd fastest growing healthcare company in North America on Deloitte's 2013 Technology Fast 500 list. During the quarter ended December 31, 2013, Crescendo tested 27,000 RA patient samples, an increase of 23 percent over the prior quarter ended September 30, 2013. Crescendo Bioscience, Inc. will retain its name and operate as a wholly owned subsidiary of Myriad Genetics, Inc. based in South San Francisco.

Benefits of the Transaction

- Facilitates Myriad's Entry into the High Growth Autoimmune Market: Myriad sees a major growth opportunity for molecular diagnostics to aid patients in the autoimmune market. Beyond rheumatoid arthritis there are several other areas in rheumatology as well as the broader autoimmune market where molecular diagnostics could play a major role in patient risk assessment, diagnosis, prognostic outlook and therapy selection. This complements Myriad's current disease focus which includes products in oncology, women's health, urology, and dermatology.
- Diversifies Myriad's Product Revenues: A core strategic initiative for Myriad is to continue to diversify its revenue stream into new attractive
 markets. Vectra DA represents a \$3.0 billion global market opportunity, and Myriad believes Crescendo will be a major growth driver for the
 Company looking forward.
- Enhances Myriad's Strength in Protein-Based Diagnostics: Myriad, a world leader in molecular diagnostics, believes that it will further strengthen its leadership position in protein diagnostics with the acquisition of Crescendo. The three major sources of novel biomarkers are DNA, RNA, and proteins and Myriad believes that Crescendo will contribute significantly to Myriad's industry leading diagnostic product pipeline and complement the protein expertise at Myriad RBM.
- Creates Opportunities to Leverage Myriad's Commercial Infrastructure: As one of the largest molecular diagnostic laboratories in the United States, Myriad expects to leverage its operational expertise in customer service, sales and marketing, managed care, and product development to further expand Crescendo's domestic business. Additionally, Myriad's expects that its international presence will open up new market opportunities for Crescendo's products outside of the United States.

• Offers Long-Term Financial Benefits: Myriad expects Crescendo to generate significant operating profits in the future and to be accretive to earnings beginning in fiscal year 2016. In April 2013, Vectra DA received national coverage for reimbursement by the Centers for Medicare and Medicaid Services (CMS) and is in the process of obtaining private insurance coverage. The typical RA patient may be monitored with a Vectra DA test at least twice per year.

Financing and Approvals

Myriad intends to fund the transaction entirely through cash on hand and anticipates having sufficient cash following the transaction to continue its current \$300 million share repurchase program. The transaction is expected to be completed before the end of fiscal year 2014 and is subject to the satisfaction of customary closing conditions and regulatory approvals.

Advisors

JP Morgan Securities LLC is acting as financial advisor to Myriad Genetics and Mintz Levin Cohn Ferris Glovsky and Popeo PC is serving as legal counsel.

Conference Call

The Company will discuss the proposed acquisition during its second fiscal quarter earnings call on Tuesday, February 4, 2014 at 4:30 pm EST. Participating on the call will be Peter Meldrum, president and CEO; William Hagstrom, president of Crescendo Bioscience; Mark Capone, president of Myriad Genetic Laboratories; and James Evans, CFO. The dial-in number for domestic callers is (800) 891-8357. International callers may dial (212) 231-2921. All callers will be asked to enter the reservation number 21703449. An archived replay of the call will be available for seven days by dialing (800) 633-8284 and entering the above reservation number. The conference call also will be available through a live webcast at <u>www.myriad.com</u>.

About Crescendo Biosciences

Crescendo Bioscience, Inc., is a molecular diagnostics company dedicated to developing and commercializing quantitative blood tests for rheumatoid arthritis (RA) and other autoimmune diseases, located in South San Francisco, CA. Crescendo Bioscience develops quantitative, objective, blood tests to provide rheumatologists with deeper clinical insight to help enable more effective management of patients with autoimmune and inflammatory diseases. For more information, please visit the company's website at <u>http://www.CrescendoBio.com</u>.

About Myriad Genetics

Myriad Genetics is a leading molecular diagnostic company dedicated to making a difference in patients' lives through the discovery and commercialization of transformative tests to assess a person's risk of developing disease, guide treatment decisions and assess risk of disease progression and recurrence. Myriad's molecular diagnostic tests are based on an understanding of the role genes play in human disease and were developed with a commitment to improving an individual's decision making process for monitoring and treating disease. Myriad is focused on strategic directives to introduce new products, including companion diagnostics, as well as expanding internationally. For more information on how Myriad is making a difference, please visit the Company's website: www.myriad.com. Myriad logo, BART, BRACAnalysis, Colaris, Colaris AP, Melaris, Myriad myPath, Myriad myPlan, Myriad myRisk, TheraGuide, Prezeon, Panexia, and Prolaris are trademarks or registered trademarks of Myriad Genetics, Inc. in the United States and foreign countries. MYGN-F, MYGN-G

Safe Harbor Statement

This press release contains "forward-looking statements" within the meaning of the Private Securities Litigation Reform Act of 1995, including statements relating to the timing of the closing of the proposed acquisition; the fit of Crescendo into the Company's diagnostic portfolio and diversification of the Company's business as a result of this acquisition; the expected addition of Crescendo employees to the Myriad team; Crescendo's position in the rheumatoid arthritis diagnostic market and its ability to move to the next level in terms of scale and growth with the Company's assistance; Crescendo's vision of multiple opportunities over the next several years; the anticipated benefits of the acquisition of Crescendo set forth under the caption "Benefits of the Transaction"; the Company's intent to fund the transaction entirely through cash and marketable securities on hand and the Company's expectation of having sufficient cash following the transaction to continue its current \$300 million share repurchase program;; and the Company's strategic directives under the caption "About Myriad Genetics". These "forward-looking statements" are management's present expectations of future events and are subject to a number of risks and uncertainties that could cause actual results to differ materially and adversely from those described in the forward-looking statements. These risks include, but are not limited to: the risk that sales and profit margins of our existing molecular diagnostic tests and companion diagnostic services may decline or will not continue to increase at historical rates; risks related to changes in the governmental or private insurers reimbursement levels for our tests; risks related to increased

competition and the development of new competing tests and services; the risk that we may be unable to develop or achieve commercial success for additional molecular diagnostic tests and companion diagnostic services in a timely manner, or at all; the risk that we may not successfully develop new markets for our molecular diagnostic tests and companion diagnostic services, including our ability to successfully generate revenue outside the United States; the risk that licenses to the technology underlying our molecular diagnostic tests and companion diagnostic services tests and any future tests are terminated or cannot be maintained on satisfactory terms; risks related to delays or other problems with operating our laboratory testing facilities; risks related to public concern over our genetic testing in general or our tests in particular; risks related to regulatory requirements or enforcement in the United States and foreign countries and changes in the structure of the healthcare system or healthcare payment systems; risks related to our ability to obtain new corporate collaborations or licenses and acquire new technologies or businesses on satisfactory terms, if at all; risks related to our ability to successfully integrate and derive benefits from any technologies or businesses that we license or acquire; the risk that we or our licensors may be unable to protect or that third parties will infringe the proprietary technologies underlying our tests; the risk of patent-infringement claims or challenges to the validity of our patents or other intellectual property; risks related to changes in intellectual property laws covering our molecular diagnostic tests and companion diagnostic services and patents or enforcement in the United States and foreign countries; risks of new, changing and competitive technologies and regulations in the United States and internationally; and other factors discussed under the heading "Risk Factors" contained in Item 1A of our most recent Annual Report on Form 10-K filed with the Securities and Exchange Commission, as well as any updates to those risk factors filed from time to time in our Ouarterly Reports on Form 10-Q or Current Reports on Form 8-K. All information in this press release is as of the date of the release, and Myriad undertakes no duty to update this information unless required by law.



News Release

Media Contact:Ron RogersInvestor Contact:Scott Gleason(801) 584-3065(801) 584-1143rrogers@myriad.comsgleason@myriad.com

Myriad Genetics Reports Financial Results for Second Quarter of Fiscal Year 2014

Revenues Increase 37 Percent Year-Over-Year; Fiscal Year 2014 Guidance Raised

Salt Lake City, February 4, 2013 – Myriad Genetics, Inc. (NASDAQ: MYGN) today announced financial results for its fiscal second quarter and six months ended December 31, 2013. Revenue for the second quarter was \$204.1 million, a 37 percent increase over the same period in the prior year. Second quarter earnings per diluted share (EPS) were \$0.66, an increase of 57 percent compared to the second quarter of 2013.

"I am extremely proud of the Myriad team as this quarter represents the tenth consecutive quarter where our revenue growth has exceeded 20 percent," said Peter D. Meldrum, president and CEO of Myriad. "Looking ahead, we plan to continue to execute upon our strategic plan of transitioning our hereditary cancer market, expanding our business internationally, and diversifying our revenue base with new product launches."

Fiscal Second Quarter 2014 Results

- Molecular diagnostic testing revenue in the second quarter rose to \$196.2 million, an increase of 39 percent compared to the second quarter of 2013. We believe that our increased sales, marketing, and education efforts resulted in wider acceptance of our molecular diagnostic tests by the medical community and increased patient testing volumes. Revenue from the Oncology segment was \$101.6 million, an increase of 12 percent compared to the second quarter of 2013. Women's Health revenue totaled \$94.6 million, an increase of 90 percent from the same period in the prior year.
 - Revenue from the BRAC*Analysis*® test was \$141.2 million, a 28 percent increase over the prior year period, and represented 69 percent of total revenue in the second quarter.

- Revenue from the BART[™] test was \$24.7 million, a 57 percent increase over the same period a year ago, and represented 12 percent of total revenue.
- Revenue from the COLARIS[®] and COLARIS *AP*[®] tests was \$15.6 million, an increase of 29 percent compared to the second quarter of 2013, and represented 8 percent of total revenue.
- Revenue from the Myriad myRiskTM Hereditary Cancer test, which launched on September 3, 2013, was \$11.5 million in the second quarter, representing 6 percent of total revenue.
- Revenue from Myriad's other molecular diagnostic tests was \$3.2 million, an increase of 25 percent compared to the previous year, and represented approximately 1 percent of total revenue.
- Companion diagnostic service revenue in the fiscal second quarter was \$7.9 million, a 7 percent decrease over the same period in 2013. Companion diagnostic revenue represented 4 percent of total revenue.
- Operating income in the second fiscal quarter improved to \$83.0 million, an increase of 49 percent compared to the second quarter of 2013. Operating margins in the fiscal second quarter were 40.7 percent compared to 37.3 percent in the prior year period. The improvement in operating margins was driven primarily by operational leverage in SG&A and R&D expenses.
- Net income for the fiscal second quarter was \$50.4 million, an increase of 44 percent compared to the same period in 2013.
- During the quarter, the Company repurchased 3.2 million shares or \$77.8 million of common stock under its stock repurchase program. Fiscal second quarter diluted weighted average shares outstanding were 76.8 million compared to 84.2 million in the same period last year.
- The Company ended the quarter with \$488.8 million in cash, cash equivalents and marketable investment securities compared to \$468.3 million at December 31, 2012, representing a 4 percent increase year-over-year.

Year-to-Date Performance

- Total revenue for the first half of fiscal 2014 was \$406.5 million, an increase of 44 percent over the \$282.6 million reported for the first half of fiscal 2013.
- Operating income for the first half of fiscal 2014 was \$165.9 million, an increase of 59 percent over the \$104.2 million reported for the same period of the prior year.

- Net income for the first half of fiscal 2014 was \$105.8 million, an increase of 62 percent over the \$65.2 million in the first half of the prior year.
- In the first half of fiscal 2014, diluted earnings per share increased 72 percent to \$1.33 compared to \$0.78 for the same period in the prior year.

Business Highlights

- Myriad launched the Myriad myPlan[™] Lung Cancer test in October 2013 to assist physicians in determining a newly diagnosed patient's risk of dying from lung cancer within five years. We believe that this information can assist a physician decide whether an early stage patient should receive surgery alone or surgery plus adjuvant chemotherapy or radiation therapy.
- In November 2013, Myriad launched the Myriad myPath[™] Melanoma test to leading dermatopathologists throughout the United States. The test was validated in a 464 patient clinical study where it demonstrated over 90 percent accuracy in differentiating malignant melanoma from benign skin lesions. Myriad myPath Melanoma is the only test on the market that can accurately diagnose all four major melanoma subtypes.
- Myriad presented data at the San Antonio Breast Cancer Symposium demonstrating that the Myriad myRisk Hereditary Cancer test detected 52
 percent more patients with a high risk for hereditary breast and ovarian cancer than was found by testing for the BRCA1 and BRCA2 genes
 alone. Additionally, Myriad presented data demonstrating that the Myriad myRisk test detected 60 percent more patients with a high
 predisposition risk of hereditary colon and endometrial cancer than was found by testing the major colon cancer genes alone. The Myriad myRisk
 Hereditary Cancer test has unparalleled accuracy of 99.98 percent.
- Myriad's Investigational Device Exemption (IDE) for BRACAnalysis testing was accepted by the U.S. Food and Drug Administration (FDA) for use in AstraZeneca's Phase 3 clinical trials for olaparib and for use in Biomarin's Phase 3 clinical trial for BMN-673.
- Myriad announced a new research collaboration with Janssen to use Myriad's *BRACAnalysis* test in Janssen's Phase 3 clinical trial with Yondelis (tabectedin) in the treatment of advanced-relapsed epithelial ovarian, primary peritoneal, or fallopian tube cancers.
- Myriad signed a research collaboration with Biomarin in the fiscal second quarter to use Myriad's HRD test to identify tumor types that may be sensitive to BioMarin's investigational new drug, BMN-673. This represents Myriad's second publicly announced research collaboration with the Company's HRD test.

- Myriad presented data at the San Antonio Breast Cancer Symposium showing that its HRD test was highly predictive of cisplatin response in
 patients with triple-negative breast cancer. This is the second study validating the HRD test in successfully predicting platinum response in triplenegative breast cancer patients.
- Myriad presented data at the Annual Meeting of the Society for Urologic Oncology (SUO) from four clinical studies with Prolaris testing
 demonstrating its ability to accurately predict the aggressiveness of a man's prostate cancer. Prolaris testing has now been studied in 11 clinical
 trials involving over 5,000 patients.
- Myriad presented interim data from its PROCEDE 500 study at the ASCO GU Cancer Symposium showing that physicians changed their intended therapy and selected a different treatment based on the Prolaris test score in 65 percent of their cases. In 40 percent of patients, physicians reduced the therapeutic burden on the patients and opted for active surveillance in lieu of surgery or radiation therapy. In 25 percent of patients, physicians opted for more aggressive therapies.

Increased Fiscal Year 2014 Outlook

Based on the growth in its core markets and the acquisition of Crescendo Bioscience announced today, Myriad is raising its revenue expectations for fiscal year ending June 30, 2014. The Company is now forecasting revenues of \$740 to \$750 million, compared to previous guidance of \$700 to \$715 million. This new guidance represents 21 to 22 percent revenue growth when compared to the prior fiscal year and incorporates the Company's projection for approximately \$10 million in revenue from the Crescendo acquisition, the completion of which is subject to regulatory clearance and satisfaction of customary closing conditions.

Myriad also is increasing its diluted earnings per share guidance for fiscal year 2014. The Company is projecting diluted earnings per share of \$2.09 to \$2.12 compared to previous guidance of \$1.92 to \$1.97. This new guidance represents 18 to 20 percent EPS growth. While this guidance does not contemplate additional share repurchase activity, Myriad is committed to its share repurchase program and expects that it will have sufficient cash reserves to complete the acquisition of Crescendo Bioscience and continue repurchasing shares under its current \$300 million repurchase authorization.

Conference Call and Webcast

A conference call will be held on Tuesday, February 5, 2014, at 4:30 p.m. Eastern Time to discuss Myriad's financial results for the fiscal second quarter of 2014. The dial-in number for domestic callers is (800) 891-8357. International callers may dial (212) 231-2921. All callers will be asked to enter the reservation number 21703449. An archived replay of the call will be available for seven days by dialing (800) 633-8284 and entering the above reservation number. The conference call also will be available through a live webcast at <u>www.myriad.com</u>.

About Myriad Genetics

Myriad Genetics is a leading molecular diagnostic company dedicated to making a difference in patients' lives through the discovery and commercialization of transformative tests to assess a person's risk of developing disease, guide treatment decisions and assess risk of disease progression and recurrence. Myriad's molecular diagnostic tests are based on an understanding of the role genes play in human disease and were developed with a commitment to improving an individual's decision making process for monitoring and treating disease. Myriad is focused on strategic directives to introduce new products, including companion diagnostics, as well as expanding internationally. For more information on how Myriad is making a difference, please visit the Company's website: www.myriad.com.

Myriad, the Myriad logo, BART, BRACAnalysis, Colaris, Colaris AP, Melaris, Myriad myPath, Myriad myPlan, Myriad myRisk, TheraGuide, Prezeon, Panexia, and Prolaris are trademarks or registered trademarks of Myriad Genetics, Inc. in the United States and foreign countries. MYGN-F, MYGN-G

Safe Harbor Statement

This press release contains "forward-looking statements" within the meaning of the Private Securities Litigation Reform Act of 1995, including statements relating to the Company's plans to execute its strategic plan of transitioning its hereditary cancer market, expanding its business internationally, and diversifying its revenue base with new product launches; the announcement and expected closing of the acquisition of Crescendo Bioscience; the Company's financial guidance under the caption "Increased Fiscal Year 2014 Outlook," including the additional revenues expected upon the completion of the Crescendo acquisition; the Company's commitment to its share repurchase program and continuation of repurchasing shares under the Company's current \$300 million authorization; the availability of cash reserves to complete the acquisition of Crescendo Bioscience; and the Company's strategic directives under the caption "About Myriad Genetics". These "forward-looking statements" are management's present expectations of future events and are subject to a number of risks and uncertainties that could cause actual results to differ materially and adversely from those described in the forward-looking statements. These risks include, but are not limited to: the risk that sales and profit margins of our existing molecular diagnostic tests and companion diagnostic services may decline or will not continue to increase at historical rates; risks

related to changes in the governmental or private insurers reimbursement levels for our tests; risks related to increased competition and the development of new competing tests and services; the risk that we may be unable to develop or achieve commercial success for additional molecular diagnostic tests and companion diagnostic services in a timely manner, or at all; the risk that we may not successfully develop new markets for our molecular diagnostic tests and companion diagnostic services, including our ability to successfully generate revenue outside the United States; the risk that licenses to the technology underlying our molecular diagnostic tests and companion diagnostic services tests and any future tests are terminated or cannot be maintained on satisfactory terms; risks related to delays or other problems with operating our laboratory testing facilities; risks related to public concern over our genetic testing in general or our tests in particular; risks related to regulatory requirements or enforcement in the United States and foreign countries and changes in the structure of the healthcare system or healthcare payment systems; risks related to our ability to obtain new corporate collaborations or licenses and acquire new technologies or businesses on satisfactory terms, if at all; risks related to our ability to successfully integrate and derive benefits from any technologies or businesses that we license or acquire; the risk that we or our licensors may be unable to protect or that third parties will infringe the proprietary technologies underlying our tests; the risk of patent-infringement claims or challenges to the validity of our patents or other intellectual property; risks related to changes in intellectual property laws covering our molecular diagnostic tests and companion diagnostic services and patents or enforcement in the United States and foreign countries; risks of new, changing and competitive technologies and regulations in the United States and internationally; and other factors discussed under the heading "Risk Factors" contained in Item 1A of our most recent Annual Report on Form 10-K filed with the Securities and Exchange Commission, as well as any updates to those risk factors filed from time to time in our Quarterly Reports on Form 10-Q or Current Reports on Form 8-K. All information in this press release is as of the date of the release, and Myriad undertakes no duty to update this information unless required by law.

MYRIAD GENETICS, INC. AND SUBSIDIARIES CONDENSED CONSOLIDATED INCOME STATEMENTS (Unaudited)

		Three Months Ended			Six Months Ended			
(in thousands, except per share amounts)	Dec. 31	<u>, , , , , , , , , , , , , , , , , , , </u>	2. 31, 2012		<u>31, 2013</u> 89,144	_	. 31, 2012	
Molecular diagnostic testing			140,651			Ъ	267,919	
Companion diagnostic services		7,902	8,489		17,383		14,658	
Total revenue	20	4,060	149,140	4	06,527		282,577	
Costs and expenses:								
Cost of molecular diagnostic testing		2,755	15,566		44,194		29,498	
Cost of companion diagnostic services		3,376	4,318		7,418		7,713	
Research and development expense	1	7,090	14,107		33,893		25,507	
Selling, general, and administrative expense	7	7,840	59,563	1	55,119		115,691	
Total costs and expenses	12	1,061	93,554	2	40,624		178,409	
Operating income	8	2,999	55,586	1	65,903		104,168	
Other income (expense):								
Interest income		1,330	1,385		2,691		2,753	
Other		(185)	14		(623)		(114)	
Total other income		1,145	1,399		2,068		2,639	
Income before income taxes	8	4,144	56,985	1	67,971		106,807	
Income tax provision (benefit)	3	3,784	21,949		62,146		41,635	
Net income	<mark>\$</mark> 5	0,360 \$	35,036	1	05,825		65,172	
Earnings per share:								
Basic	\$	0.67 \$	0.43	\$	1.37	\$	0.80	
Diluted	\$	0.66 \$	0.42	\$	1.33	\$	0.78	
Weighted average shares outstanding								
Basic	7	5,070	81,692		77,323		81,632	
Diluted	7	6,825	84,240		79,312		84,091	

Condensed Consolidated Balance Sheets (Unaudited)

	Dec. 31, 2013	Jun. 30, 2013
(In thousands)		
Cash, cash equivalents, and marketable investment securities	\$ 488,782	\$ 531,064
Trade receivables, net	84,137	94,333
Other receivables	3,463	3,373
Prepaid expenses	4,237	5,963
Equipment and leasehold improvements, net	31,313	27,602
Note receivable	23,000	21,667
Other assets	13,000	13,000
Intangibles, net	12,842	13,330
Goodwill	56,850	56,850
Deferred tax assets	39,057	36,639
Total assets	\$ 756,681	\$ 803,821
Accounts payable and accrued liabilities	\$ 63,440	\$ 62,466
Deferred revenue	3,952	2,043
Uncertain tax benefits	13,318	10,718
Stockholders' equity	675,971	728,594
Total liabilities and stockholders' equity	\$ 756,681	\$ 803,821